



DISSERTATION

Statistical Analysis of Reliability and Variability Effects in CMOS Technologies Based on Single-Defect Spectroscopy

zur Erlangung des akademischen Grades **Doktor der technischen Wissenschaften**

EINGEREICHT AND DER **Technische Universität Wien** Fakultät für Elektrotechnik und Informationstechnik

VON

Dipl.-Ing. Konstantinos Tselios Matr. Nr. 12037282

UNTER BETREUUNG VON Associate Prof. Dipl.-Ing. Dr.techn. Michael Waltl

Wien, 19. Februar 2024



Acknowledgements

My doctoral journey has been an enriching experience, and I would like to acknowledge the individuals who have been instrumental in making it possible. Firstly, my heartfelt gratitude goes to PROFESSOR MICHAEL WALTL for his unwavering support, invaluable guidance, and supervision throughout my Ph.D. studies. I am grateful for his expertise, mentorship, and timely feedback, which have been instrumental in shaping my research.

I would like to express my appreciation to PROFESSOR TIBOR GRASSER, whose guidance have been crucial to the completion of my research. Thank you for your valuable insights, suggestions, and advice, which have helped me navigate challenging research questions.

I want to thank my colleagues ALEXANDROS PROVIAS, YOANLYS HERNANDEZ, JAKOB MICHL and DOMINIC WALDHÖR for valuable discussions and especially THERE-SIA KNOBLOCH, CHRISTIAN SCHLEICH and BERNARD STAMPFER who further contributed to my thesis by proofreading it.

I extend my gratitude to the non-scientific staff at the Institute for Microelectronics, particularly EWALD HASLINGER, MANFRED KATTERBAUER, DIANA POP, PETRA KAMPTNER-JONAS, and MARKUS SCHLOFFER. They have provided invaluable assistance in several other aspects of my academic journey, and their efforts in ensuring a conducive working environment at the institute are highly appreciated.

I would like to express my sincere gratitude to my family and friends, who have been my constant support system throughout not only my academic journey but also in every aspect of my life. I am particularly grateful to my parents SOTIRIOS and EFROSINI, whose unwavering love, encouragement, and sacrifices have made it possible for me to pursue my dreams. Additionally, I extend my appreciation to my brother CHRISTOS and girlfriend SOFIA for their unending support, understanding, and motivation. Thank you for being there for me every step of the way.



Abstract

The metal-oxide-semiconductor field effect transistor (MOSFET) is the integral building block of modern electronic devices. Its influence has triggered substantial cultural, societal, and economic shifts over the past few decades. Recognizing the sustained relevance of MOSFET technology, there is a considerable emphasis on amplifying their performance through ongoing research and development. Nevertheless, it is crucial to acknowledge that MOSFETs contain defects at the atomic scale due to their internal structure, fabrication procedures, and operating conditions. These defects represent obstacles to the reliability and functionality of MOSFETs as they trap electric charges and degrade the performance of these transistors. The impact of these defects is increasingly prominent as the trend toward transistor miniaturization continues. This situation raises issues such as bias-temperature instabilities (BTI) and random telegraph noise (RTN).

This work aims to contribute to understanding defects' impact on the performance of MOSFETs through experimental characterization and statistical modeling. The main focus is to analyze the impact of single defects on the device performance. By leveraging the extended measure-stress-measure (eMSM) on nanoscale devices, it is possible to extract the impact of single defects on the threshold voltage shift ΔV_{th} , one of the most core parameters of a transistor. Single-defect spectroscopy, used in conjunction with the defect-centric model (DCM) to evaluate the data, enables the accurate extraction of defect characteristics. This is essential for understanding the statistically distributed nature of active defects, particularly in terms of their number and impact on ΔV_{th} . This approach is especially crucial for identifying high-impact defects, known as "killer" defects, which can lead to immediate device and circuit failures in nanoscale nodes. From the results of the statistical analysis, the impact of device geometry and body bias on average threshold shift that can be induced by a single defect can be obtained, enhancing the understanding of time-dependent variations across different technologies.

Additionally, reliability simulations are employed to calculate the cumulative response of many defects on large devices and enable a comparison of theoretical trap parameters with the measurement data. For the experiments, three distinct technologies are investigated, enabling conclusions to be drawn about their distribution and physical properties. Combining experimental measurements, statistical modeling, and compact-physics simulations can lead to understanding defects in MOSFETs and device and circuit performance and reliability.



vi

Contents

Co	Contents v		
Li	st of]	Figures	xi
Li	st of .	Abbreviations x	xiii
1	Intr	oduction	1
	1.1	History of MOSFETS	2
	1.2	Reliability Issues	4
	1.3	Variability and Yield	5
	1.4	Single Defect Analysis	6
	1.5	Defect Centric Perspective	7
	1.6	Scope of this Work	8
	1.7	Outline	9
2	Def	ects in Field Effect Transistors	11
	2.1	Interface Defects	11
	2.2	Oxide defects	13
		2.2.1 Oxide Defects in SiO_2	14
		2.2.2 Oxide Defects in SiON	15
		2.2.3 Oxide Defects in HfO_2	16
	2.3	Semiconductor substrate defects	17
3	Phy	sical Modeling of Charge Transfer Reactions Involving Defects	19
	3.1	Modeling Defect Interactions with Markov Chains	19
	3.2	SRH-like models for the transition rate	21
		3.2.1 Recombination through defects - SRH model	21
		3.2.2 Kirton Uren model	23
	3.3	Non-Radiative Multi-phonon Model	24
		3.3.1 The 2-State Non-Radiative Multi-Phonon Model	25
		3.3.2 The 4-State Non-Radiative Multi-Phonon Model	26
4	Reli	ability Simulations	31
	4.1	Compact Physics Framework (Comphy)	31
	4.2	Electrostatics	31

		4.2.1	Effective Density of States	32
		4.2.2	Carrier Concentrations at Thermal Equilibrium	32
		4.2.3	Fermi Level at Thermal Equilibrium	32
		4.2.4	Electrostatics of a 1D MOS Structure	33
	4.3	Occup	ation Probabilities	35
	4.4	Distrik	oution of Defect Parameters	36
		4.4.1	Effective Single Defect Decomposition	37
	4.5	Issues	Related to Charge Sheet Approximation	38
5	Sing	gle Defe	ect Statistics	41
	5.1	Introd	uction	41
	5.2	Statist	ical Distributions of Defects	41
	5.3	Total 2	$V_{\rm th}$ Distribution	43
		5.3.1	Derived Parameters	44
6	Mea	sureme	ent Methods	47
	6.1	Time-z	zero Characterization	47
		6.1.1	Transfer Characteristics	47
		6.1.2	Capacitance - Voltage (CV) Measurements for Characterizing	
			MOSFET Devices	49
	6.2	Time-I	Dependent Characterization	51
		6.2.1	Measure-Stress-Measure (MSM) Methods	51
		6.2.2	TDDS	52
		6.2.3	RTN Measurements	53
	6.3	Measu	rement Setup	55
		6.3.1	Custom-Made Low Noise Measurement Unit for MOSFET Relia-	
			bility Measurements	56
7	Edg	e Detec	tion Algorithms	59
	7.1	Canny	^v Edge Detector	59
	7.2	Otsu's	Algorithm	60
8	Mea	sureme	ents and Results	63
	8.1	Impac	t of Defects on SiO_2 Transistors $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	63
		8.1.1	Devices and Measurements	64
		8.1.2	Measurement Analysis	66
		8.1.3	Negative BTI in pMOS Devices	67
		8.1.4	Positive BTI in nMOS devices	72
		8.1.5	Body Bias Dependence	75
		8.1.6	Conclusions	78
	8.2	Single	-defect analysis of BTI employing SiON Transistors	80
		8.2.1	Devices and Measurements	80
		8.2.2	Positive BTI in nMOS Transistors	80
		8.2.3	Negative BTI in nMOS Transistors	82

		8.2.4	Negative BTI of pMOS Transistors	83
		8.2.5	Comparing NBTI and PBTI degradation	84
		8.2.6	Simulation Results	85
		8.2.7	Conclusions	88
	8.3	Statist	ical Characterization of RTN in high- κ devices using Smart Arrays	90
		8.3.1	Devices and Measurements	90
	8.4	Extrac	tion of Step Heights	90
	8.5	Distri	butions of Step Heights	93
	8.6	Param	neter Extraction	94
9	Sun	nmary a	and Outlook	97
	9.1	Summ	nary	97
	9.2	Outlo	ok	98
Re	References 101			101
Li	st of :	Publica	itions	117



List of Figures

. 8
. 13
. 14
15

2.4	The structural arrangement of a-HfO ₂ (with Hf represented by silver and O represented by red) exhibits a greater level of disorder compared to a-SiO ₂ (with Si represented by yellow and O represented by red). This disparity in disorder serves as a potential cause for intrinsic charge trapping, which could provide an explanation for the PBTI behavior observed in high- κ devices. Figure taken from [49].	16
3.1	Example of Markov chain with three discrete states. The arrows indicate the transition rates between the states.	20
3.2	The band diagram schematically represents the capture and emission mechanisms for acceptor-like traps (depicted in blue) and donor-like traps (shown in red). As electron traps transition between neutral and negatively charged states, hole traps fluctuate between neutral and posi- tively charged states. The diagram portrays the initial states of the defects before the occurrence of the specified transition. Figure adapted from [50]	22
3.3	The configuration coordinate diagram depicts the Kirton-Uren model, where a defect interacts with the reservoir through electron exchange. The position of the parabola is determined by the energy of the carrier. The thermal activation involved in the capture process is represented by the capture barrier, denoted as ΔE_B . In the diagram, the empty defect with an electron in the conduction band is depicted in black, while the captured electron at the defect site is shown in grey. The dashed line represents the energy zero of the system, corresponding to the empty defect with an electron at the Fermi level. This diagram is based on [80, 50].	24
3.4	Potential energy surfaces (PESs) with a parabolic shape for the charged (depicted in grey) and neutral state (depicted in black) as employed in the NMP defect model, accompanied by the model's parameters. This representation is recreated based on [50].	25
3.5	The four-state NMP model characterizes the capture and emission times of individual defects in response to changing biases and temperatures. Broadly, the model considers two stable states, 1 (neutral) and 2 (charged), as well as two meta-stable states, 1' and 2'. It is crucial to consider these states when trying to accurately represent the emission times and other related characteristics that are dependent on bias [76].	27
3.6	The four-state NMP model characterizes the capture and emission times of individual defects in response to changing biases and temperatures. Broadly, the model takes into account two stable states, 1 (neutral) and 2 (charged), as well as two meta-stable states, 1' and 2'. These states are essential for accurately depicting the bias-dependent emission times and other associated attributes [76].	28

4.1	Comphy employs an effective 2-state NMP model to describe the charge transition of a pre-existing oxide defect. This transition from state <i>i</i> to state <i>j</i> occurs by overcoming the energy barrier ϵ_{ij} . The model's harmonic potential energy surfaces are characterized by the trap level E_{T} , the relaxation energy E_{R} , and the curvature ratio <i>R</i> . For the computation of the tunneling factor, the spatial coordinate x_{T} is employed. Figure taken from [KTJ4]	36
4.2	A defect can be characterized by the following parameters: relaxation energy (E_R), trap level (E_T), spatial position in the oxide (x_T), curvature ratio (R), and configuration coordinate offset (ΔQ). It is important to note that R and ΔQ are associated with the energy levels, while x_T is assumed to have a uniform distribution. Additionally, the energy levels follow a normal distribution within the defect bands. Recreated from [51]	36
4.3	The use of Gaussian trap bands in the 28 nm technology, as discussed in [66], indicates that only a fraction of the distribution contributes to active defects. The greater part of the defect parameters extracted from the trap band results in charge transition times exceeding 10 ⁷ seconds (as seen on the left), rendering them experimentally unattainable. These prolonged charge transition periods correlate with relaxation energies ER that are significantly larger than those derived from DFT calculations (as seen on the right). Therefore, they can be considered as artifacts of the Gaussian distributions. The figure is a reproduction from [KTJ3].	38
4.4	The influence of the regularization parameter γ is significant. When γ is small, the solution necessitates an excessively high defect density, which is not physically realistic. On the other hand, excessively large values of γ result in rapidly increasing approximation errors. The ideal value for γ resides between these two unfavorable regions. Originally published in [KTJ3]	39
5.1	The left part of the figure illustrates typical measure-stress-measure (MSM) sequences obtained from 40 devices of equal dimensions ($W \times L = 135 \text{ nm} \times 350 \text{ nm}$). Using the Canny algorithm, integrated into the data analysis framework [113, 116] used in this work, discrete steps representing charge transition events are extracted from the recovery traces, leading to the creation of a corresponding distribution. The middle panel demonstrates that the CCDF adheres to an exponential distribution while, for comparison, a uniform distribution obtained from the charge sheet approximation (CSA) is also included. Finally, the right panel exhibits a quantile-quantile (Q-Q) plot, comparing the distribution of the number of steps per device to a Poisson distribution. This plot validates that the number of steps per device can be accurately described by a Poisson	

42

5.2	Experimental data from the largest device in the measurement set show that discrete steps are not observable in larger devices. Therefore, the statistical averages of the defects can be determined using DCM on the ΔV_{th} traces	45
6.1	$I_D(V_G)$ measurements for a set of devices, with the channel width $W = 135$ nm and the length $L = 350$ nm. Using the constant current method, the threshold voltage V_{th} is extracted at $I_D = I_{ref} \frac{W}{L}$. The reference current I_{ref} is assumed to be 100 nA in this case. The V_{th} values range between 0.8 and 1 V for the identical devices of this technology.	48
6.2	The example results from a 100 kHz C-V measurement on nMOS devices display distinct accumulation and inversion branches at the peak values of negative and positive biases, respectively. The capacitance observed during the depletion phase is significantly lower. Additionally, the inversion branch tends to exhibit a steeper slope compared to the accumulation branch. $V_{\rm th}$ can also be estimated by the minimum value	
6.3	of the plot	50
	$I_{\rm D}(V_{\rm G})$	52
6.4	The presented recovery traces highlight the threshold voltage drift follow- ing PBTI stress in scaled SiON nMOS transistors. These devices exhibit recovery in discrete steps. Within these traces, electron emissions are identifiable as negative steps (in blue), and hole emissions as positive steps (in red) affecting the <i>V</i> ₁ behavior. Originally published in [KTI4]	53
6.5	The eMSM-based measurement strategy is utilized to evaluate the change in the threshold voltage. Once the initial $I_D(V_G)$ is measured, stress and recovery voltages are applied in alternating fashion, with increasing stress and recovery durations (t_s and t_r). During the relaxation phase, the source current is recorded and subsequently converted into an equiva- lent ΔV_{th} using the initially measured $I_D(V_G)$ characteristics. Originally	
6.6	published in [KTJ5] The outcomes commonly observed when implementing the eMSM sequence on SiON devices with dimensions $W \times L = 10 \times 10 \ \mu m^2$ exhibit a continuous behavior in terms of the measured drain current (I _D). This arises from the simultaneous emission of previously trapped charges by a multitude of defects across the large device area. The individual influence of a single defect on the device's performance is too minute to	54
	be discerned through measurements. Originally published in [KTJ3]	55

6.7	Four representative RTN signals exhibit varying mean charge capture and emission times, as well as step heights. From these RTN signals, the power spectral density (depicted in the same color as the correspond- ing RTN signal) can be derived, adhering to a Lorentzian distribution. The corner frequency of the Lorentzian corresponds to the mean cap- ture/emission time τ of the RTN signal. The superposition of uniformly distributed Lorentzian signals results in a 1/ <i>f</i> behavior, which is widely recognized in large area devices.	56
6.8	A constant gate voltage measurement setup schematic for methods fo- cused on channel conductivity is displayed. The drain and gate voltages of the device are controlled by DACs. The gate voltage is directly applied to the DUT, while the drain current is connected to the positive terminal of an OPAMP operating as a TIA. The negative input terminal of the OPAMP is linked to the DUT's drain connection. The TIA's voltage out- put, corresponding to V_{out} , is captured by an analog-to-digital converter (ADC). Recreated from [50].	57
6.9	A diagram of a constant source current measurement configuration for channel conductivity-focused methods is presented. The drain bias of the DUT is directly controlled, while the gate bias is managed by the OPAMP. The OPAMP circuit adjusts the MOSFET gate bias such that the source current matches the reference current determined by the DAC bias. Recreated from [50].	57
7.1	The Canny method for identifying edges is showcased using a represen- tative measurement trace. The input signal, ΔV_{th} , is convolved with the first derivative of a Gaussian pulse with a specified variance, <i>G</i> . This results in a signal <i>R</i> , which features peaks corresponding to the initial signal steps. All peaks exceeding a predetermined threshold are identi- fied and utilized to indicate the locations of the steps in the initial trace. The peak heights can be acquired from either the original signal or the peak heights. Recreated from [126]	61
7.2	For an RTN signal (left), a normalized histogram (right) with <i>L</i> bins is computed. For each bin, the inter-class variance σ_b^2 can be determined, and the bin with the maximum σ_b^2 provides the best threshold value. The locations where the signal transitions from above to below the threshold value can be extracted, and the corresponding step heights and time constants are obtained.	62

8.1	Schematic of the carrier densities in the channel of a scaled device and the resulting current percolation path. (a) The randomly distributed dopants lead to locally decreased carrier densities. The neutral defect does not influence the current path. (b) When the defect is charged, the restrictions on the percolation path increase leading to a degreesed drain current	
	and a higher threshold voltage [50]. Originally published in [KTJ7]	64
8.2	Recovery traces obtained from scaled pMOS and nMOS devices showing the drift of the threshold voltage after NBTI/PBTI stress over time, mea- sured on a scaled SiO ₂ technology. As can be seen, the recovery proceeds in positive discrete steps for pMOS and negative for nMOS. Increased variability is observed between nominally identical devices. Originally published in [KT]7].	65
8.3	Recovery traces obtained from large pMOS and nMOS devices showing the drift of the threshold voltage after NBTI/PBTI stress. The traces recover similarly in a continuous manner as the increased number of de- fects with reduced impact on ΔV_{th} lead to a smaller variability. Originally published in [KTI7]	65
8.4	The CCDFs of the extracted discrete steps are shown. The pMOS data seem to follow a unimodal exponential function, while nMOS data can be described by a bimodal exponential function. The maximum limit of	05
8.5	the CSA is shown as vertical dashed lines. Originally published in [KTJ7]. The CCDFs of step heights of hole traps extracted after NBTI stress on pMOS devices. The dashed lines depict the unimodal exponential distribution used to describe the results. A recovery bias close to V_{th} was applied during the recovery phase. Originally published in [KTJ7].	66 68
8.6	(Top) The η values for the unimodal fit used in Figure 8.5 are plotted against the area. The devices with the same <i>W</i> scale nicely, but the devices with different <i>W</i> deviate from this trend, showing higher step heights. (Bottom) The η values for the unimodal fit are plotted against $W \times \sqrt{L}$. The fitted line seems to accurately capture all the data obtained from different geometries. Originally published in [KTJ7].	69
8.7	The η values are plotted against $W \times \sqrt{L}$ trend. The fitted line nicely captures both experimental and DCM values. For some geometries, the markers used for the experimental data and data from the DCM model overlap as their values are too close to see a difference on a logarithmic scale. Originally published in [KTI7]	69
8.8	The average number of defects per device is plotted against both dimen- sional trends. The marker and color of the symbols follow the same	70
8.9	Convention as previous figures. Originally published in [K1J7] The $I_D(V_G)$ characteristics of the smallest devices of the set. The grey dashed lines indicate the measurement conditions where the recovery	70
	traces were recorded. Originally published in [KTJ7]	70

8.10	The η values for the on-state measurements are plotted against $W \times \sqrt{L}$. The fitted line accurately captures all the data obtained from different	
	geometries and the DCM values for the two large geometries. Originally published in [KTJ7].	71
8.11	Traces of the smallest device of the set measured close to Vth (red lines) and at high operating conditions. At high operating conditions, the recovery is no longer observed in a discrete manner due to the increased	71
8.12	The average number of active defects for the two measurement cases plotted for all the geometries. Higher values are observed when devices	/1
	are measured at the on-state. Originally published in [KTJ7].	72
8.13	The CCDFs of step heights of electron traps extracted after PBTI stress on nMOS devices. The bimodal exponential distributions (solid lines) describe the results more accurately than the unimodal exponential dis- tributions (dashed lines). Originally published in [KTC1]	73
8.14	(Top) The average step heights induced by single defects (η) for the data used in Figure 8.13 are plotted against the area. The red line shows the scaling of the values following a unimodal exponential distribution, and the gray dashed line is used as a comparison to the CSA. The RMSE values were normalized to compare the two different η ranges. (Bottom) The average step heights induced by a single defect (η) for the two distri- butions used in Figure 8.13 are plotted against $W \times \sqrt{L}$. A smaller error is observed in the latter case for the unimodal distribution. Originally published in [KTI7]	74
8.15	The average step heights induced by single defects (η) is used together with calculated values from DCM for the large devices that are plotted against $W \times \sqrt{L}$. The step detection values are taken from the second branch of bimodal distribution as they are closer to DCM values than the values extracted from unimodal distribution. Originally published in [KTJ7].	74
8.16	Probability distribution of the initial V_{th0} values across a sample set of nanoscaled devices under varying V_B conditions, conforming to normal distributions. This change is largely attributed to modifications occurring in the depletion layer thickness X_{dep} as a response to alterations in BB	75
8.17	Settings	75
8.18	nounced slope	75
	their pMOS counterparts.	76

8.	19 Exponential distributions of individual charge ΔV_{th} step heights for a consistent device sample set under varied V_{B} conditions. RBB shows a heightened variance (elevated η), whereas the FBB exhibits a diminished variance (lower η).	76
8.	20 The average step heights induced by a single defects (η) against the $V_{\rm B}$ for pMOS devices. Increased $V_{\rm B}$ leads to higher η values. The smaller geometry exhibits higher η values at $V_{\rm B} = 0$ and a more pronounced slope.	77
8.2	21 TDDS on two different BB settings on a single device. The studied step height is heavily influenced. The larger RBB leads to an increase in the step height, an observation consistent with what has been observed in the CCDFs of multiple devices	77
8.	22 The CCDFs of step heights of hole traps extracted after NBTI stress on pMOS devices. The dashed lines depict the unimodal exponential function used to explain the results.	78
8.2	23 The average step heights induced by a single defect (η) for the data used in Figure 8 are plotted against the $V_{\rm B}$. Increased - $V_{\rm B}$ leads to higher η values. The slope is less steep for the nMOS case	78
8.2	24 TDDS for two different BB conditions on a selected device. The step heights at larger RBB are increased, an observation consistent with what has been observed in the CCDFs. BB impact is smaller in nMOS devices than in pMOS devices.	79
8.2	25 The figure presents the band diagram of nMOS devices under PBTI con- ditions. It illustrates the presence of electron traps (depicted in blue) and hole traps (depicted in red), along with their respective AERs for charge trapping. The blue region represents the AER for charge transitions be- tween the defects and the channel. In contrast, the red region represents the AER for charge transitions between the defects and the gate. These regions define the energetic boundaries within which the defects can change their charge state during the experiments, thereby contributing to the drift of the measurement signal. It should be noted that the height of the AERs varies with the applied gate bias. On the right side, the distribution of step heights, measured from SiON nMOS transistors after undergoing PBTI stress, is shown for three sets of devices with different geometries. The majority of observed traps are electron traps (depicted on the left), although a certain number of hole traps (depicted on the right) can also be observed. The electron traps show a bimodal expo- nential distribution, while the holes traps show a unimodal exponential behavior. Originally published in [KTC2]	81

8.26	(Left) The band diagram of nMOS devices for the NBTI conditions. In contrast to the PBTI case, more hole traps can contribute to total ΔV_{th} now as the channel's AER covers the band-gap's lower half. The AERs for charge transitions between the defects and the channel and for charge transitions between the defects and the gate are marked in blue and red color, respectively. (Right) Distribution of step heights of defects extracted after NBTI stress. Bimodal exponential behavior can be observed for both types of traps. Originally published in [KTC2].	83
8.27	Distribution of step heights of hole traps extracted after NBTI stress on pMOS devices. The dashed lines depict the unimodal exponential function used to explain the results. originally published in [KTJ4]	84
8.28	Extracted impact of a single defect on the $\Delta V_{\text{th}}(\eta)$ and the number of traps per device (N _T) considering unimodal exponential distribution. Clearly visible is the underestimation of the CSA of the impact of the defects, on the device behavior. The maximum impact of CSA is considered by considering the traps directly at the interface. Originally published in [KTC2]	85
8.29	The extracted values for the two modes of the bimodal exponential distribution are shown. As can be seen from Figure 8.25, the distribution follows more a bimodal exponential behavior than a unimodal one. Again, the CSA significantly underestimates the impact of the defects, which leads to a too pessimistic estimation for defect density from ΔV_{th} . Originally published in [KTC2].	86
8.30	The extracted values for the unimodal exponential distribution of the hole traps of pMOS devices. The η values seem to be proportional to the gate area. The max limit of CSA approximation is around three times smaller than values extracted from the CCDFs. Originally published in [KTC2].	86
8.31	Extracted defect bands from eMSM measurements on the SiON devices. nMOS bands are magnified to be accurately visible in the figure. Recre- ated from [KTJ3].	87
8.32	Samples are obtained from the extracted defect distributions within the band diagram. The search regions used in the ESiD algorithm are represented by shaded areas. The highest concentration of both electron (blue) and hole (red) traps is found within the IL. Originally published [KTJ3].	87

8.33	In the top figure MSM measurements (dots) on large-area nMOS devices are shown, as presented in [KTJ3], at $T = 100$ °C under three different stress conditions. The lines are simulated with the BTI simulator Comphy averaged for 100 sets of sampled step heights using the CCDFs extracted in section 8.2.2. The regions marked in grey are shown in more detail in the two bottom figures. Every light line presents a BTI simulation with a sampled set of step heights, the solid line is the average of all performed simulations. Originally published in [KTJ4]	88
8.34	In the top figure MSM measurements (dots) on large-area pMOS devices are shown, as presented in [KTJ3], at $T = 100$ °C under three different stress conditions. The lines are simulated with the BTI simulator Comphy averaged for 100 sets of sampled step heights using the CCDFs extracted in Section 8.2.4. The regions marked in grey are shown in more detail in the two bottom figures. Every light line presents a BTI simulation with a sampled set of step heights, the solid line is the average of all performed simulations.	89
8.35	Initial $I_D(V_G)$ measurements of the biggest geometry in the bias range to capture the RTN signals. Originally published in [51, 143]	91
8.36	After identifying the distinctive levels of an arbitrary RTN signal (shown at the top), various parameters can be derived, such as step magnitudes (shown at bottom left) and the times for charge capture and emission (shown at bottom right and middle). However, in this particular signal, the distributions of t_e and t_c fail to meet the criteria set by the χ^2 and Kolmogorov-Smirnov tests. This exclusion ensures the avoidance of erroneous detections, thus making this particular trace unsuitable for statistical analysis.	92
8.37	After identifying the distinct steps present in an arbitrary RTN signal (depicted at the top), numerous parameters can be derived, such as step magnitudes (shown at bottom left) and the times for charge capture and emission (shown at bottom right and middle). In this particular trace, the distributions of t_e and t_c successfully pass both the χ^2 and Kolmogorov-Smirnov tests, ensuring accurate detection. As a result, this trace is suitable for utilization in the statistical analysis.	92
8.38	The CCDFs of step heights of hole traps extracted after RTN measure- ments on pMOS devices at $T=4$ K. The dashed lines depict the unimodal exponential distribution that has been used to describe the results. At this temperature, the exponential distribution seems unable to describe the results, also the mean values do not scale with the area but present a random behavior	02
		20

8.39	The CCDFs of step heights of hole traps extracted after RTN measurements on pMOS devices at $T = 7 K$. The dashed lines depict the unimodal	
	exponential distribution that has been used to describe the results. Similar	
	to the $T = 7 K$ case, the exponential distribution seems unable to describe	
	the results and also the mean values seem to not scale with the area but	
	present a random behavior, too.	93
8.40	The CCDFs of step heights of hole traps extracted after RTN measure-	
	ments on pMOS devices at T= 225 K. The dashed lines depict the uni-	
	modal exponential distribution that has been used to describe the results.	
	At this temperature, the exponential distribution captures nicely the	
	results. Also, a scaling with the area starts to exist.	94
8.41	The CCDFs of step heights of hole traps extracted after RTN measure-	
	ments on pMOS devices at T= 300 K. The dashed lines depict the uni-	
	modal exponential distribution that has been used to describe the results.	
	At this temperature similar to the T= 225 K case, the exponential distribu-	
	tion captures nicely, and the area scaling is clear.	94
8.42	The number of devices exhibiting RTN versus temperature for all the dif-	
	ferent areas. Even though the results are mixed, more devices exhibiting	
	RTN can be seen in higher temperatures	95
8.43	The extracted η values for the T= 225 K case presented in Figure 8.40 are	
	plotted against the area. The results follow the area trend except for a	
	small outlier at L=170 nm. The experimental values are higher than the	
	maximum limit of CSA, similar to the other studies presented in the thesis.	95
8.44	The extracted η values for the T= 225 K case presented in Figure 8.41 are	
	plotted against the area. The results follow the area trend with a smaller	
	RMSE than in the $T = 225$ K case. The experimental values are higher than	
	the maximum limit of CSA, although their deviation is smaller than the	
	T = 225 K case.	96
8.45	The extracted η values for the T= 225 K case presented in Figure 8.40 are	
	plotted against the $W \times \sqrt{L}$ trend. The results follow the trend with the	
	same RMSE as observed in the area trend.	96
8.46	The extracted η values for the T= 300 K case presented in Figure 8.41 are	
	plotted against the $W \times \sqrt{L}$ trend. The results follow the trend with the	0.1
	same RMSE as observed in the area trend	96

List of Abbreviations

ADC analog-to-digital coverter

aRTN anomalous RTN

BJT bipolar junction transistor

BTI bias temperature instability

CCDF complementary cumulative distribution function

CDF cumulative distribution function

CMOS complementary MOS

CV capacitance-voltage

DAC digital-to-analog coverter

DCIV direct-current current-voltage

DFT density functional theory

DIBL drain-induced barrier lowering

DLTS deep-level transient spectroscopy

DUT device under test

eMSM extended MSM

ESR electron spin resonance

FET field-effect transistor

GIDL gate-induced drain leakage

HC hot carrier

HMM hidden Markov model

IV current-voltage

MIS metal-insulator-semiconductor

MOS metal-oxide-semiconductor

MOSCAP MOS capacitor

MOSFET MOS field-effect-transistor

MSM measure stress measure

NMP non-radiative multi-phonon

NMR nuclear magnetic resonance

OPAMP operational amplifier

OTF on-the-fly

- **PDF** probability density function
- PES potential energy surface

PSD power spectral density

RAM random access memory

RTN random telegraph noise

RTS random telegraph signal

SILC stress induced leakage current

SIMS secondary ion mass spectroscopy

SNR signal-to-noise ratio

SPICE simulation program with integrated circuit emphasis

SRH Shockley-Read-Hall

TCAD technology computer-aided design

TDDS time-dependent defect spectroscopy

TDRC thermal dielectric relaxation current

WKB Wentzel-Kramers-Brillouin

XPS X-ray photoelectron spectroscopy

Chapter 1 Introduction

In the modern world, electronic devices have become an integral part of daily life. This has been mainly enabled as the electronic industry has seen rapid progress in recent years with the development of smaller, faster, and more reliable devices. At the heart of most of the electronic devices lies the metal-oxide-semiconductor (MOS) transistor as one of the main building blocks, which at a first glance seems like a simple stack of three material layers [1]. MOS structures function by exploiting the field effect: applying voltage to their metal gate terminal modulates the conductivity between the source and drain terminals. This creates a channel at the semiconductor-insulator interface, allowing for the switching of electronic signals, making them essential to reproduce the binary logic necessary for logic gates of computer architectures. Due to the usage of the field effect these structures are commonly called metal-oxide-semiconductor fieldeffect transistors (MOSFETs). Beyond the realm of digital electronics, transistors are also extensively used in analog applications, functioning as amplifiers due to their high input impedance, low output impedance, and minimal power consumption. Their significance is particularly noteworthy in amplifying weak signals from sensors, microphones, and other low-level input sources, a function critical to the communication applications in the current era [2].

However, the continuous improvements and evolution of MOS devices are limited by parameters such as the speed and power consumption amongst other reliability issues. These constraints primarily stem from defects in the structure, either located in the oxide or at the interface between the oxide and the semiconductor. These defects can alter the electrical properties of the MOSFET by capturing and releasing charges in response to applied biases or temperature changes. This phenomenon can cause fluctuations in the threshold voltage, subthreshold slope, and transconductance of the device, leading to performance degradation and reduced reliability [3, 4]. In the worstcase scenario, single defects can cause catastrophic device failure. The impact of single defects is particularly significant in advanced nanoscale MOSFETs where the margin for variability in the device's electrical characteristics decreases. Therefore, any defect causing even slight deviations in these characteristics can have a substantial impact on the device's operation. Improving the reliability of MOS devices is a critical challenge that needs to be addressed to ensure the smooth operation of electronic systems. This thesis aims to contribute to a better understanding of MOS device degradation phenomena and their mechanisms related to single defects, focusing on their impact on threshold voltage (V_{th}) of devices and circuits. The research presented in this thesis highlights the challenges associated with describing the impact of defects statistically and provides an overview of modeling efforts to capture the essence of the physical mechanism behind charge capture. By investigating these phenomena, the goal is to provide valuable insights that researchers and practitioners can leverage to improve the reliability and lifetime of MOS devices and, ultimately, the electronic systems that depend on them.

1.1 History of MOSFETS

The invention and evolution of the MOSFET have played a critical role in shaping the modern electronics industry. Before its invention in the 1960s, the electronic industry relied on bipolar junction transistors (BJTs) for amplification and switching applications. However, BJTs had several limitations, including high power consumption, high input capacitance, and poor thermal stability.

Julius Edgar Lilienfeld, an Austrian physicist, was the first to conceptualize the idea of using the field effect to modulate conductivity in a semiconductor triode structure, which he filed for a patent in 1928 [5]. The idea of a MOS device, which would eventually lead to the development of the MOSFET, was first proposed in the late 1950s. Although Lilienfeld filed for a patent on the idea in 1925, it was not until the invention of the MOS capacitor in the late 1950s that the possibility of a MOS transistor became a reality [6]. This marked a significant milestone in the electronic industry, as it offered several advantages over BJTs, including low power consumption, low input capacitance, and high thermal stability. These advantages paved the way for the widespread use of MOSFETs in modern electronics.

The first devices were developed in the late 1960s and early 1970s, which were relatively simple compared to the advancements of today. In 1963 the researchers Sah and Wanlass from Fairchild suggested a new type of MOSFET logic combining both nMOS and pMOS, and filed a patent which was granted in 1967. This combination was called complementary MOS (CMOS) [7], an idea which is behind on most of the electronics of today. The main reliability issue during this period was the limited lifespan of the devices, as they were prone to breakdown over time due to high electric fields at the oxide-semiconductor interface.

During the 1970s and 1980s, advancements were made in the processing techniques used to manufacture MOSFETs, resulting in improved device reliability. The use of new materials, such as silicon dioxide and nitrided oxides, allowed for the creation of more robust and reliable devices. However, the major reliability issue during this period was the limited power handling capability of the devices, as the high temperatures generated by high current levels could cause permanent damage.

During the 1990s and early 2000s, there was a shift in focus towards making MOS-

FETs smaller and improving their power handling capabilities, following the principles of Moore's Law [8]. This law predicts that transistor density will approximately double every two years, leading to better performance and lower production costs. To ensure this continuous growth, semiconductor manufacturers began reducing the dimensions of transistors, such as their width, length, and insulating layer thickness. However, this aggressive downsizing resulted in several challenges. One of these challenges was the difficulty in scaling down supply voltages without affecting the on/off current ratio. Additionally, it became necessary to maintain a minimum thickness for the insulating layer to prevent excessive leakage currents.

These challenges prompted the exploration of alternative materials for gate stacks. High-k materials, like Hafnium oxide (HfO₂), combined with metal-gate contacts, emerged as a solution for cutting-edge devices [9, 10]. These materials enabled the reduction of the effective oxide thickness (EOT) while maintaining the necessary insulating properties. As a result, high-k gate stacks are now commonly utilized in highperformance CMOS applications. Apart from incorporating high-k materials, another strategy to enhance CMOS performance is the utilization of high mobility channels [11, 12]. For example, replacing the conventional silicon channel with a silicon-germanium (SiGe) channel allows for higher mobility, leading to larger on currents. Recent research has focused on gate-all-around (GAA) transistors, a cutting-edge design where the gate material surrounds the channel from all sides, offering enhanced control over the channel and significantly reducing leakage currents. This innovation represents a key evolution from FinFET architecture. Additionally, Forksheet devices, an advanced offshoot of the GAA design, have been explored for their potential to further reduce short-channel effects and increase drive current, showcasing a promising path for scaling beyond the limitations of traditional MOSFETs [13, 14]. Lastly, it should be noted that 2D materials are being explored for novel MOS devices. One advantage of using these materials is that the interface between the 2D material and the insulator does not have interface defects since their connection relies on van der Waals bonds, in contrast to the covalent bonds found in Si/SiO2 interfaces. However, the development of these devices is still in its early stages and requires substantial advancements in fabrication processes to establish this technology [KTJ1].

However, the continuous process of scaling faces inherent limitations imposed by atomic dimensions. In 2016, the International Technology Roadmap of Semiconductors (ITRS) [15], responsible for establishing development standards and outlining future advancements, published its final roadmap. This decision was prompted by the imminent boundaries of the classical scaling approach, which were expected to be reached in the 2020s. Consequently, the IEEE International Roadmap for Devices and Systems (IRDS) [16] was introduced in 2016 as a successor to the ITRS, focusing on a wider range of future developments. Within this context, every year's version of IRDS mentions the importance of further development in the oxides of the devices.

1.2 Reliability Issues

The reliable and stable operation of electronic devices can be compromised by various reliability issues that arise from harsh electrical stress, elevated temperature, radiation, and mechanical stress, negatively impacting the transfer characteristics of the device and affecting the reliability of interconnects [17, 18]. In the worst case, oxide breakdown can occur, leading to an increase in oxide leakage currents [19, 20]. Among the different types of device performance degradation mechanisms, this work focuses on the electrical issues caused by defects in the atomic structure of the device. Defects can change their charge state by charge capture and emission, thereby leading to changes in device performance. To comprehend the intricacies of charge trapping mechanisms, it becomes imperative to discern between different degradation mechanisms that prevail under distinct bias conditions, as depicted in Figure 1.1.



Figure 1.1. Illustrating the prevailing degradation mechanisms for various bias regimes, a diagrammatical depiction of the V_G , V_D space is presented. This schematic representation highlights the crucial factors influencing device performance and reliability under different operating conditions. Recreated from [21].

The Bias Temperature Instability (BTI) is a widely investigated mechanism responsible for performance degradation in electronic devices. This phenomenon is primarily characterized by variations in device parameters, such as threshold voltage (V_{th}) and channel mobility (μ), which result from charge trapping within the oxide layer when a bias is applied to the device's gate. Charge trapping is accelerated by elevated temperatures (T) and gate biases (V_G), but it can be partially reversed when the gate bias is reduced. Both positive BTI (PBTI) and negative BTI (NBTI) can occur in nMOS and pMOS devices. The general focus is on NBTI in pMOS devices and PBTI in nMOS devices, as these are associated with the typical bias conditions under operation. In silicon, the pMOS/NBTI combination receives greater attention since it often exhibits the most substantial degradation due to defects and the energetic alignment of the hole defect band [22, 23, 4]. An additional phenomenon stemming from the same defects that cause BTI is Random Telegraph Noise (RTN), which occurs when the energy level of defects is situated near the Fermi level of the channel at a given operational point. RTN represents a state of dynamic equilibrium for the channel/gate dielectrics system, while BTI represents the disturbed system reverting to this equilibrium state [24]. This effect is characterized by the random capture and release of charges by the defects, leading to noise in the channel current. The magnitude of the RTN effect escalates with device scaling and is linked to various challenges in integrated circuits, such as increased failure probabilities in SRAM and jitter in ring oscillators [25, 26, 27, KTJ1].

Stress-Induced Leakage Current (SILC) or trap-assisted tunneling is another mechanism that allows defects in the oxide to facilitate charge tunneling between the channel and the gate, resulting in elevated gate leakage currents. This rise in device power consumption can lead to thermal failure and reduced retention time for EEPROM and FLASH memory cells following numerous write and erase cycles [28, 29].

Finally, Hot Carrier Degradation (HCD) is caused by carriers with high kinetic energy that cause damage close to the interface, where they can break Si-H bonds or become trapped, leading to the creation of interface states. The highest energy carriers are typically observed close to the end of the channel, where most of the damage is typically observed. The interface states can charge and affect the channel conduction similarly to BTI defects [30, 31].

Even though this work predominantly emphasizes BTI and RTN, it's crucial not to overlook the substantial impact of the other two degradation mechanisms, SILC and HCD. These mechanisms are vitally essential and contribute extensively to ensuring a reliable, consistent functionality of both devices and circuits, forming an integral part of the broader picture of device reliability and stability.

1.3 Variability and Yield

Another critical issue with modern electronic devices is the variability in scaled nodes. MOSFETs are subject to various sources of variability, including time-zero device variability and time-dependent variability, commonly referred to as device reliability. During the manufacturing process, process variations inevitably occur, causing deviations from the ideal characteristics of seemingly identical devices. This variability can have a detrimental impact on the performance of single devices and circuits and also affects the yield of the devices.

Time-zero device variability, which is a result of limited control over the fabrication process, can be classified as local and random variations within a die, variations across a single wafer due to inhomogeneities during processing, and variations between different wafers in a lot due to changing processing conditions during manufacturing [32, 33]. Other factors contributing to variability include fluctuations in surface roughness and thickness of patterned structures, metal grain roughness, and random discrete dopants.

Variations in surface roughness and thickness of patterned structures contribute

to line edge roughness (LER), leading to stochastic fluctuations in device geometry [34]. In high-k transistors, metal grain roughness (MGR) must be considered, as these devices necessitate a metal gate contact on top of the high-k dielectrics. The metal layer's work-function depends on metal grain's orientation, introducing random variations in the threshold voltage [35]. Random discrete dopants (RDDs) play a crucial role in scaled transistors as the arbitrary placement of the limited number of dopant atoms in such devices results in threshold voltage fluctuations exceeding tens of millivolts [36, 37]. The influence of RDD fluctuations on the standard deviation of the threshold voltage ($\sigma_{V_{th}}$) can be approximately represented by an analytic function [38]:

$$\sigma_{V_{\rm th}}^2 \approx \sqrt{4q^3 \varepsilon_{\rm Si} \phi_B} \left(\frac{t_{\rm ox}}{\varepsilon_{\rm ox}}\right)^2 \frac{\sqrt{N_{\rm tot}}}{W \times L} \tag{1.1}$$

where ϕ_B denotes the surface potential and N_{tot} represents the doping concentration. Importantly, $\sigma_{V_{\text{th}}}$ depends on the oxide thickness (t_{ox}) and the total dopant concentration (N_{tot}). Other examples of process variations include annealing effects and lithographic limitations [38, 39].

Besides time-zero device variability, MOSFET devices may undergo time-dependent variability, which pertains to changes in device characteristics over time due to aging and environmental factors. BTI and HCD are two prevalent sources of time-dependent variability. Consequently, the overall variability of deeply-scaled devices arises from a combination of time-zero and time-dependent variability. Accurately describing the time-dependent and overall statistical distributions is thus essential for reliably predicting the performance of future deeply downscaled technologies.

To mitigate the impact of variability, various techniques have been developed, including using high-quality gate oxide materials, optimizing device structures, and implementing stress and recovery cycles to counteract BTI effects. Additionally, circuit-level design strategies such as redundancy and error-correction codes can also help improve MOSFET-based circuits' reliability in the presence of variability.

1.4 Single Defect Analysis

To describe time-dependent variability and the statistics of defects influencing nanoscale technologies, extracting the impact of individual defects in electrical measurements is crucial. Single defect spectroscopy [40, 41, 42] is an advanced characterization method focusing on the study of the behavior and properties of individual defects in semiconductor materials and devices. This potent technique offers researchers a unique opportunity to investigate single defect characteristics, such as charge state dynamics and energy levels. Examining these properties allows scientists to optimize device performance, improve reliability, and deepen the understanding of defect-related phenomena in semiconductors [43].

In recent years, single defect spectroscopy has become a vital tool for examining gate dielectrics in MOSFETs. Specifically, it has been applied to investigate high-k

dielectric materials, like hafnium oxide (HfO₂) [44, 45], which are extensively used in modern semiconductor devices due to their capacity to mitigate short-channel effects and maintain low leakage currents at state-of-the-art supply voltages.

Through single defect spectroscopy, researchers can reveal the following aspects of individual defects in semiconductor materials and devices:

- **Charge Trapping Dynamics:** The technique enables scientists to explore the charge trapping and de-trapping processes of defects in gate dielectrics. Comprehending these dynamics is essential for optimizing device performance and identifying potential reliability concerns.
- Energy Level Characterization: Single defect spectroscopy allows the determination of defect energy levels, which is crucial for understanding the impact of defects on device operation and devising strategies to alleviate their effects.
- **Spatial Localization:** Analyzing individual defects provides insights into their spatial distribution within the semiconductor material. This information is valuable for determining defect origins and identifying methods to reduce their occurrence during the fabrication process.
- **Temporal Behavior:** Single defect spectroscopy also elucidates the temporal behavior of defects, including their stability and lifetime. This knowledge is critical for evaluating the long-term reliability of semiconductor devices and developing strategies to prolong their operational lifetimes.

The utilization of single defect spectroscopy can contribute significantly to the comprehension of the behavior exhibited by individual defects in semiconductor materials and devices [46, 43]. This technique offers intricate insights into the properties and conduct of single defects, holding great promise for driving advancements in semiconductor device performance, reliability, and manufacturing processes.

1.5 Defect Centric Perspective

The results extracted from single defect analysis can create the framework to describe defect reliability based on their statistics. Given that only a limited number of defects have an impact on the time-dependent effects observed in deeply-scaled devices, it is crucial to understand the degradation mechanisms at the individual defect level to simulate time-dependent variability in circuits accurately [47, 48]. This concept forms the foundation of the defect-centric approach, as depicted in a hierarchical manner in Figure 1.2. The initial step is defining fundamental terms related to variability and proceeds to examine the characteristics of individual defects, explaining how these properties can be propagated to higher hierarchical levels. A commonly used statistical framework, known as Defect-Centric Model (DCM), is used to describe the distribution of devices over time through simple equations. The model is based on the statistics of the number of defects per device which follows a Poisson distribution with a mean value of $N_{\rm T}$ and the statistics of the effect of a single defect on the threshold voltage which is assumed to be exponentially distributed with an average value of η [24]. To accurately estimate the parameter distributions of devices and circuits at the end of their useful lifetime, it is important to combine both time-zero and time-dependent statistics. Lastly, it is crucial to propagate and simulate the combined variability at the circuit level.



Figure 1.2. The defect-centric picture allows for hierarchical traversal, whereby insights into the properties of defects at the atomic level can be propagated upwards to inform circuit design decisions. Originally published in [47].

1.6 Scope of this Work

The purpose of this thesis is to investigate the impact of defects within the oxide and at the semiconductor/oxide interface on the operating characteristics of MOS transistors. As the lateral dimensions of MOS transistors have been reduced, their switching speed and power consumption have improved. However, with the reduction of lateral dimensions, defects within the oxide and at the semiconductor/oxide interface have an enhanced impact on the device characteristics. This leads to increased variability in device characteristics such as threshold voltages, sub-threshold swing, and carrier mobility, even when comparing devices that are nominally identical.

In order to investigate the impact of defects in semiconductor devices, a series of measurements were conducted on numerous commercial pMOS and nMOS devices. Based on the data collected, statistical distributions of defect properties were created to explore the relationship between statistical quantities, like the link between the average threshold shift of a single emission event and the lateral device dimensions. To achieve a more accurate understanding of the effect of charge traps on device behavior and to gather valuable information for the entire technology, a combined approach involving single-defect extractions and the use of the DCM was employed. This approach enables

a more precise assessment of the impact of charge traps on device behavior, which is vital for the improvement of the layout and performance of circuits.

The focus of this thesis is to study the variability of MOS transistors by analyzing the contributions of defects on the device behavior. The findings from this research provide helpful insights into designing more reliable circuits [KTJ2], and can be used to gain a deeper understanding of the impact of defects on MOS transistors. The approach taken in this work is also applicable to the study of other technologies and can be used as a useful tool for improving the reliability of electronic devices.

1.7 Outline

The study presented here offers a comprehensive exploration of the impact of defects on MOSFETs, encompassing both experimental and theoretical aspects. The thesis is structured into several chapters discussing the characterization and modeling of defects.

The introductory chapter sets the stage by providing background information and context for the study. Subsequent chapters focus on specific aspects of defect characterization and modeling.

Chapter 2 provides an overview of the primary types of electrically active defects commonly found in MOSFETs. The chapter explains their origins and interactions with the devices, featuring relevant oxides specific to silicon technology such as the SiO_2 , SiON, and HfO₂.

Chapter 3 discusses various mathematical modeling approaches for defects and their effects on device behavior. It introduces key concepts, such as employing a Markov chain with capture and emission rates to represent defects. The chapter covers different modeling techniques and outlines the mathematical tools essential for defect modeling.

Chapter 4 discusses the simulation framework utilized in the study, based on electrostatics and the non-radiative multiphonon (NMP) model. The chapter addresses the impact of the charge sheet approximation in simulations and highlights the need to describe the average impact of defects.

Chapter 5 centers on the statistical characterization of defects, exploring the distributions used to explain their behavior. It introduces the defects-centric model, which provides a framework for describing the behavior of single defects based on the distributions of their amplitude and concentration.

Chapter 6 explores experimental techniques employed in defect characterization, discussing various methods such as capacitance-voltage (C-V) measurements, conductance techniques, and noise measurements. The chapter also outlines the equipment and setups utilized for experimental defect characterization.

Chapter 7 focuses on extracting defect parameters from measurements, bridging the gap between experimental defect characterization and defect modeling. The chapter examines different approaches for extracting information about defects from measurements, emphasizing the importance of accurate parameter extraction for effective defect

TU Bibliotheks Die approbierte gedruckte Originalversion dieser Dissertation ist an der TU Wien Bibliothek verfügbar. WIEN ^{vour knowledge hub} The approved original version of this doctoral thesis is available in print at TU Wien Bibliothek.

modeling.

Finally, Chapter 8 presents defect characterization studies conducted on three different technologies, incorporating combinations of the methods discussed in previous chapters. The chapter describes the experimental setups and provides a detailed analysis of the results. The data obtained from these experiments serves to validate the models and methods discussed earlier in the study.

Collectively, this work offers a comprehensive examination of defects in MOSFETs, encompassing both theoretical and experimental aspects. The study provides a deeper understanding of how defects impact MOSFET performance by exploring different defect types, their mathematical modeling, and experimental characterization methods.

The implications of this study are stimulating for the development of improved defect management strategies in MOSFETs. By identifying and comprehending the mechanisms underlying various defect types, researchers and engineers can work towards minimizing their impact and enhancing device reliability.

The inclusion of defect characterization studies conducted on multiple technologies demonstrates the practical relevance of this research. These experiments offer valuable insights into defect behavior in different materials and device structures, thereby informing the future development of more robust and reliable MOSFETs.
Chapter 2

Defects in Field Effect Transistors

The presence of electrically active defects in the oxide or at the interface between the oxide and semiconductor of an MOS transistor can cause reliability issues, such as BTI, HCD, and RTN, that negatively impact device performance and can lead to device and circuit failure. Even though manufacturing processes have improved, defect formation cannot be fully avoided during fabrication or operation, making it crucial to understand the physical processes behind these issues to model them and consider the possible alterations on the device performance for circuit designs.

Various theoretical and experimental approaches are used to identify and study specific defect candidates. These include Density Functional Theory (DFT) and Technology Computer-Aided Design (TCAD) simulations for theoretical analysis, electrical examination tools, and physical characterization methods such as Electron Paramagnetic Resonance (EPR) spectroscopy and X-ray Photoelectron Spectroscopy (XPS). Thereby, researchers study the trapping properties of these defects to understand their impact on device performance better.

As technology advances to each new node, emerging requirements necessitate a comprehensive understanding of the effects of defects to ensure proper design and manufacturing processes. Continuous research is crucial as new materials and device structures emerge, and novel defect types may arise, further expanding the limits of electronics technology. In this chapter, the most significant known defect candidates for interface and oxide defects to date will be discussed.*Parts of this chapter have been previously published in the work of Waldhoer et al.* [49] and the PhD theses of Stampfer [50] and Michl [51].

2.1 Interface Defects

The process of manufacturing devices often leads to defects in the interfacial layer between the semiconductor substrate and insulating layer due to the structural mismatch between the materials. The density and configuration of these defects are highly influenced by the semiconductor and insulator materials used, the processing parameters, and the presence of other species during processing that may passivate these defects. Limiting the formation during fabrication or passivation of these interface defects is a significant challenge in developing new material systems. For instance, the application of in-situ oxidation and hydrogen passivation has resulted in a low interface defect density of approximately $10^{10}/(\text{cm}^2\text{eV})$ in silicon devices [52]. This process optimization, combined with the high quality of the oxide, is a major factor in the continued success of the Si/SiO₂ material system in modern electronics.

The study of interface defects in silicon-based MOSFETs has a long-standing background. Characterization methods, such as electron spin resonance (ESR) also frequently referred to as electron paramagnetic resonance (EPR), have been extensively used to examine the nature of point defects in semiconductors and insulators. These techniques analyze the interaction between microwave absorption or carrier generation/recombination processes and a magnetic field, thereby enabling the investigation of the composition of these defects. Specifically, the interaction occurs with unpaired electrons found in unsaturated atomic bonds. In the case of a semiconductor-insulator system, these paramagnetic point defects can function as trapped charges, thereby altering the device's parameters. As a result, ESR investigations play a crucial role in the study of device degradation.

The inherent mismatch in crystal structure between silicon (Si) and the amorphous layer of thermally grown silicon dioxide (SiO₂) gives rise to unsaturated bonds at the interface of these two materials [53]. These unsatisfied bonds can be observed using ESR and are referred to as dangling bonds on a silicon atom at the Si/SiO₂ interface (referred to as *P*b center), distinguished by their anisotropy and Landé g-factor [54]. The orbitals associated with these bonds can accommodate up to two electrons and exhibit two trap levels within the substrate bandgap. A schematic of "dangling" bonds can be seen in Figure 2.1. These specific point defects have been linked to interface traps that can be electrically measured.

In the 1960s, Nishi et al. [55] conducted a pioneering study and discovered three distinct types of defect signatures on (111) interfaces. Later, similar P_b centers were identified on (110) and (100) surfaces [56, 57]. The original P_b center (\cdot Si \equiv Si3) is now known as the P_{b0} center on (110) surfaces, while chemically different P_{b1} centers, which are suspected to be oxidized P_b centers (\cdot Si \equiv Si₂O), were also found on (100) surfaces [58]. The centers which are known to occur in various chemical variations, include the P_{b0} center in the form of (\cdot Si \equiv Si₃) and the P_{b1} center. However, the atomistic structure of the latter still needs to be fully understood.

High densities of interface defects pose a significant challenge, as they can negatively impact the controllability of carrier density in the channel and increase noise in the channel current. The trapped charges at interface defects can perturb the electrostatic surface potential, reducing the channel's mobility due to Coulomb scattering and leading to long-term degradation of device parameters. This issue can be exacerbated by hot carrier stress, which can cause the depassivation of previously passivated interface defects, thereby increasing the number of electrically active interface defects.

In addition to ESR measurements, other techniques are used to study interface



Figure 2.1. At the interface of Si/SiO₂ with a (100) orientation, *P*b centers can manifest in two chemical forms: the P_{b0} center, characterized by the (\cdot Si \equiv Si₃) structure, and the P_{b1} center, whose atomic structure is not fully comprehended. The dangling bond without a pair of electrons is depicted by blue triangles in the figure, which has been reproduced from [58].

defects include capacitance-voltage (CV) measurements, charge pumping (CP) measurements, deep-level transient spectroscopy (DLTS), and direct-current current-voltage (DCIV) measurements. While these techniques provide information on the average properties of the defects, they do not allow the study of the impact of charge capture and emission at individual interface states on device behavior.

2.2 Oxide defects

In earlier research on MOSFET reliability, interface defects were the primary focus of the investigation, but as the understanding of device degradation advanced, defects in the bulk oxide were also considered as part of a more comprehensive analysis. The defects located within a device's insulating oxide layer can seriously impact its performance. These oxide defects, also known as border states, can form naturally during oxide growth or as a result of impurities or distorted chemical bonds within the oxide. Oxide defects can exchange charges with both the gate and channel, although the process is typically slower compared to interface defects. This is due to the spatial separation between the defect and the charge reservoirs, the large structural relaxation that occurs at the defect site of commonly studied oxide defect candidates. The energy level of oxide defects shifts with the applied gate bias, leading to a strong dependence of their charge capture and emission rates on the gate bias. A charged oxide defect can interact with carriers in the inversion layer, reducing the local carrier density and affecting the mobility in the subthreshold regime and the threshold voltage of the device. Despite the variations among these oxide traps, they have all been associated with BTI, as they can exchange charge carriers with the substrate. Current research is focused on the precise physical description of this process, including its dependence on the oxide field and temperature.

In small gate area devices, the impact of charge trapping at oxide defects can be particularly significant. This is because the smaller capacitance of these devices leads to each defect having a larger impact on the surface potential and ,thus, the threshold voltage shift. The magnitude of the perturbation of a charged defect on the channel



Figure 2.2. Ball-and-stick representations of two potential hole trap candidates in amorphous SiO2: (a) the hydrogen bridge and (b) the hydroxyl E' center. In these models, silicon atoms are yellow, oxygen atoms are red, and hydrogen atoms are gray. The blue clouds depict the spin density for neutral defects, signifying the location of the unpaired electron, while for positive defects, they illustrate the distribution of the captured hole. Both defects can exist in four configurations, with two being neutral (states 1 and 1') and two being charged (states 2 and 2'). Figures adapted from [59, 60]

current depends on its distance from the channel and its position relative to random dopants. In these devices, even a single charged oxide defect can significantly reduce in the source-drain current due to its effect on the percolation path in the channel.

2.2.1 Oxide Defects in SiO₂

Silicon dioxide (SiO₂) is a dielectric material that has been extensively studied in MOSFET devices. Recent research suggests that hydrogen-related defects, such as hydrogen bridges (HB) and hydroxyl-E' centers (H-E'), are responsible for degradation phenomena like BTI and RTN [61].

During device processing, hydrogen is commonly used to passivate dangling bonds at the interface between the semiconductor and SiO_2 . As a result, it is highly likely that defects involving hydrogen are formed [61]. Furthermore, hydrogen can easily diffuse in the SiO_2 layer, where it can form electrically active defects. Figure 2.2 shows the neutral (1) and positive (2) charge states of the two defect candidates containing hydrogen.

When the H atom binds to a Si atom, the resulting defect configuration that involves two Si atoms is called a hydrogen bridge (HB). The HB can capture a hole and form a stable positive configuration known as a puckered state, which can transit between unpuckered and puckered states through metastable states. Hydrogen atoms can also bind with the oxygen atom and form a hydroxyl-E' center. These defects can only occur in a-SiO₂ and require strained Si-O bonds to form. In contrast to oxygen vacancies, hydrogen-related defects have trap levels close to the Si band edges, as shown in Figure 2.3 and are, therefore, more important for understanding the mechanisms behind RTN and BTI.

Most defect studies have focused on charge trapping resulting from known defect candidates. However, there have been some experimental studies that require the



Figure 2.3. The various defect candidates studied in SiO₂, HfO₂ and SiON have trap levels distributed throughout the band gaps of these materials. Oxygen vacancies (OV) have trap levels distant from the silicon conduction band, thus having a limited impact on BTI and RTN. Conversely, the trap levels of hydrogen-related defects like the hydroxyl- E_k center or the hydrogen bridge, along with interface defects such as the P_b center or the KN center, are near the silicon band edges, making them significantly relevant for BTI. Moreover, intrinsic charge trapping due to polarons is a key factor for electron trapping in high- κ devices. Originally published in [51].

existence of trap bands that cannot be directly linked to any known defects [62]. This type of charge trapping can have a significant impact on wide bandgap devices, such as Silicon Carbide (SiC). It can also occur spontaneously in $a-SiO_2$ and is caused by structures with elongated Si-O bonds, which facilitate the localization of an additional electron at a certain Si atom. The trap band of these structures lies close to the conduction band edge of Si, as shown in Figure 2.3.

Unlike interface defects, oxide defects have a more complex atomic structure that enables them to exist in multiple configurations with the same net charge. Typically, the transition from one stable configuration to another via a metastable state occurs when one of the silicon atoms transitions through the triangle formed by three oxygen atoms, resulting in a so-called puckered configuration. Due to the amorphous nature of SiO₂, these defects exhibit a wide range of properties, such as their trap levels and transition barriers, which contribute to the observed variety in time constants mentioned earlier.

2.2.2 Oxide Defects in SiON

Plasma-nitrided oxides (PNO) are commonly used in silicon technology to reduce gate leakage current [63]. However, a high concentration of nitrogen atoms at the oxidesemiconductor interface can result in a large amount of K_N centers. The K_N centers are characterized by silicon dangling bonds, where silicon is bonded to three nitrogen atoms, potentially with oxygen as second-nearest neighbor atoms [64]. These K_N centers are situated within the amorphous silicon oxynitride layer and exhibit electrical properties as interface states and bulk dielectric defects. They serve as sites for recombination and tunneling. Notably, the negative bias temperature instability observed in these plasma-nitrided gate stacks does not involve the formation of P_{b0} , P_{b1} , or E' centers



Figure 2.4. The structural arrangement of a-HfO₂ (with Hf represented by silver and O represented by red) exhibits a greater level of disorder compared to a-SiO₂ (with Si represented by yellow and O represented by red). This disparity in disorder serves as a potential cause for intrinsic charge trapping, which could provide an explanation for the PBTI behavior observed in high- κ devices. Figure taken from [49].

[63]. Although the defect candidates of near-interface defects change, hydrogen-related defects can still play a major role in nitrided oxides, as H atoms are used for passivating $P_{\rm b}$ centers [65]. $K_{\rm N}$ centers have a much narrower distribution of their energetic trap levels than $P_{\rm b}$ centers, which are broadly distributed with at least two peaks, one directly above and one directly below the midgap. The energetic trap level distribution of $K_{\rm N}$ centers is very close to the Si midgap, as seen in Figure 2.3 [66].

2.2.3 Oxide Defects in HfO₂

The scaling of MOSFETs has led to the physical thickness of insulator to be a few nm to maintain channel control. However, as the thickness of the insulator decreases, leakage currents increase, leading to several problems, such as higher of-state power consumption, heat dissipation, and a decrease in insulator reliability. Therefore, alternative insulators with larger dielectric constants, such as HfO₂ ($\varepsilon_r = 25$), ZrO₂ ($\varepsilon_r = 25$), or Al₂O₃ ($\varepsilon_r = 9.1$), have become necessary to increase gate capacitance without introducing leakage effects, as SiO₂ ($\varepsilon_r = 3.9$) is no longer sufficient.

Compared to SiO_2 , HfO_2 exhibits a higher degree of disorder, as depicted in Figure 2.4, and intrinsic charge trapping is suspected to be a central reliability issue in HfO_2 -based oxides [67]. HfO_2 is a non-glass-forming oxide, resulting in varying coordination numbers depending on the oxide phase, and it can partially crystallize during annealing, introducing additional structural uncertainties [68]. These factors result in various defect candidates, making the modeling of HfO_2 insulators extremely challenging. It has been suggested that HfO_2 can form electron polarons [69], which can be stable at room

temperature and could be responsible for positive bias temperature instability (PBTI). Uncoordinated Hf ions, capable of trapping electrons, drive the formation of electron polarons. An electron trapped in this manner can localize at multiple Hf sites, forming a ring where the Hf atoms are drawn towards the center, while the O anions are pushed outward [70].

Certain defect types discovered in amorphous silicon dioxide (a-SiO₂) have been identified in amorphous hafnium dioxide (a-HfO₂). One such example is the occurrence of oxygen vacancies (OV) in HfO₂, which can exist in neutral, positive, and negative states [71]. In HfO₂, due to the presence of ionic bonding, oxygen vacancies exhibit behavior similar to F-centers in ionic crystals. This means that any additional charges are concentrated within the vacancy, resulting in strong localization of the charges [72]. It has been suggested that the negative states of OV might be responsible for the abundance of electron traps observed in high- κ devices [73]. Previous studies have reported trap levels ranging from 1.2 eV to 1.8 eV below the conduction band of HfO₂ [73]. However, recent investigations utilizing first-principle calculations propose that the energy level of oxygen vacancies is located just below the conduction band of HfO₂ [74] and is therefore oxygen vacancies cannot be directly used to explain the experimental results [49] based on the optical [74] and electrical measurements [66].

In addition to their potential function as charge traps, the presence of oxygen vacancies in hafnia-based RRAM devices is being studied as a potential cause of resistive switching. It has been observed that when electrons are injected into HfO₂, oxygen vacancies and interstitial 0^{-2} ions can form Frenkel-pairs. Additionally, preexisting oxygen vacancies can act as electron traps, which can facilitate the formation of these Frenkel-pairs nearby, resulting in a stable divacancy [75]. This process suggests that oxygen vacancies may aggregate, leading to a significant reduction of resistivity due to the formation of a conductive filament.

2.3 Semiconductor substrate defects

Defects of the semiconductor substrate can be introduced intentionally during device manufacturing through the introduction of dopants to control conductivity and other parameters or unintentionally due to contamination resulting in deep-level impurities. Bulk defects have a secondary role in inversion layer devices and are only significant in systems where proper operation requires a certain minority lifetime, such as CMOS optical sensors, RAM, or photodiodes. The presence and concentration of bulk defects can be quantified using DLTS and various physical or chemical methods. However, as this thesis focuses on the reliability issues related to oxide and interface defects, bulk defects will not be examined.

CHAPTER 2. DEFECTS IN FIELD EFFECT TRANSISTORS

Chapter 3

Physical Modeling of Charge Transfer Reactions Involving Defects

Detailed modeling is essential to comprehend the impact of defects on the function of devices. For this purpose, two different approaches are mainly utilized, empirical and physics-based. Empirical models rely on simplistic analytical formulas to describe measurement data, whereas physics-based models aim to explain the device behavior based on the superposition of the impact of single defects. To facilitate reliability studies of MOSFETs and simulate this process, it is crucial to mathematically describe the physical process and statistical distribution of properties of trapping and de-trapping. It is important to note that charge trapping is a dynamic process influenced by several factors, including the defect's applied biases, temperature, and intrinsic properties, such as its trap level and relaxation energy. Therefore, developing a comprehensive theoretical model that considers all aspects of charge trapping poses a significant challenge. In this chapter, the focus is on physical-based models that describe the charge-trapping kinetics of defects and their interaction with a device. The statistical models will be discussed later in this work. *Parts of this section are based on the work of Grasser et al.* [76, 23] and the PhD thesis of Stampfer [50].

3.1 Modeling Defect Interactions with Markov Chains

Charge trapping models have been developed to describe the dynamic charging and discharging processes in individual defects. In the context of this research, various defect models have been discussed, all of which utilize a continuous-time Markov chain [77] to mathematically represent the charged state of defects. However, the fundamental differences among these models lie in how they interpret the transition rates between different states and the number of possible states within the Markov chain.

The core principle of using Markov chains to model defects is based on the Markov property assumption. This property posits that defects have no memory of past states and their future behavior is determined exclusively by their current state. Consequently, defects exhibit a memoryless nature in this modeling framework.



Figure 3.1. Example of Markov chain with three discrete states. The arrows indicate the transition rates between the states.

Markov chains are characterized by a finite set of states, with a defect always existing in one of these states at any given time. The behavior of a Markov chain can be described in terms of probabilities. The probability of a defect being in any of the *n* states at time *t* is represented by the state vector $\vec{P}(t)$. This model requires that the sum of the probabilities P_i for each state *i* equals one:

$$\sum_{i} P_i(t) = 1 \tag{3.1}$$

Following conventional Markov process theory the transition probabilities from state i to state j can be written as:

$$P\{X_j(t+h) = 1 \mid X_i(t) = 1; i \neq j\} = k_{ij}h + O(h),$$
(3.2)

with *h* being an infinitesimally small time step, $\lim_{h\to 0} \frac{O(h)}{h} = 0$, and k_{ij} as the transition rate. The probability that the defect stays in state *i* can be written as:

$$P\{X_i(t+h) = 1 \mid X_i(t) = 1\} = 1 - h \sum_{\substack{j \\ i \neq j}} k_{ij} + O(h)$$
(3.3)

The transition rates from state *i* to *j* are denoted as k_{ij} and can be expressed as the $n \times n$ transition matrix *K*. An example of a Markov chain in equilibrium is shown in Figure 3.1. The temporal evolution of $\vec{P}(t)$ can be described by the so-called master equation [78]:

$$\frac{dP_i(t)}{dt} = \sum_{i \neq j} (P_j(t)k_{ji} - P_i(t)k_{ij})$$
(3.4)

This equation contains the individual transition rates *k* between the states, which are described by the physical defect model. The diagonal elements in matrix *K* are assigned to represent the dwelling times, which refers to the duration during which no transitions

occur,

$$k_{ii} = 1 - \sum_{i \neq j} k_{ij},\tag{3.5}$$

the master equation can be expressed in the form of a vector:

$$\frac{d\vec{P}}{dt} = K\vec{P},\tag{3.6}$$

where *K* represents the matrix of connections. The transition rate matrix *K* for the Markov chain depicted in Figure 3.1 is as follows:

$$K = \begin{pmatrix} K_{11} & K_{12} & K_{13} \\ K_{21} & K_{22} & K_{23} \\ K_{31} & K_{32} & K_{33} \end{pmatrix} = \begin{pmatrix} 0 & 0.3 & 0.7 \\ 0.4 & 0 & 0.6 \\ 0.5 & 0.5 & 0 \end{pmatrix}$$

After this general mathematical framework is established, it is important to link these parameters with the physical properties of defects. The following sections examine the physical models that define the states within the Markov chain and the transition rates between these states.

3.2 SRH-like models for the transition rate

In the early 1950s, William Shockley, William T. Read, and Robert N. Hall developed the Shockley-Read-Hall (SRH) model, also known as trap-assisted generation/recombination [79]. This model provides a theoretical framework for studying how excess carriers and defects in semiconductor materials interact. The SRH model, which was originally developed for recombination centers situated within the bulk of a semiconductor where both the defect and carrier reservoirs are positioned at the same location, has been expanded to include trapping within oxides.

3.2.1 Recombination through defects - SRH model

According to the SRH, defects can exhibit both charged and neutral states. Acceptorlike defects, for example, capture an electron during charge trapping and subsequently acquire a negatively charged state. On the contrary, donor-like defects release an electron, resulting in a positively charged state. Typically, the SRH model applies to bulk defects within the semiconductor substrate material and does not encompass oxide defects.

The SRH model defines four separate processes, as demonstrated in Figure 3.2, to calculate the carrier transition rates between defects and a reservoir, such as a channel and gate in a MOS transistor.

The SRH model outlines four possible scenarios concerning trap levels:

• An electron in the conduction band may get trapped within an intragap state.

CHAPTER 3. PHYSICAL MODELING OF CHARGE TRANSFER REACTIONS INVOLVING DEFECTS



Figure 3.2. The band diagram schematically represents the capture and emission mechanisms for acceptor-like traps (depicted in blue) and donor-like traps (shown in red). As electron traps transition between neutral and negatively charged states, hole traps fluctuate between neutral and positively charged states. The diagram portrays the initial states of the defects before the occurrence of the specified transition. Figure adapted from [50]

- An electron can be released from a trap level and enter the conduction band.
- A hole located in the valence band may get captured by a trap, similar to an occupied trap releasing an electron into the valence band.
- A trapped hole can also be released into the valence band, equivalent to an electron being captured from the valence band.

It is assumed that a defect is in one of its two states, 1 for neutral and 2 for positive. The expectation values for the defect to be in either state are f_1 and f_2 , with $f_1 + f_2 = 1$. We consider a system consisting of a defect plus an electron that can be moved back and forth between the defect and the reservoir. When the electron is at the defect site, its energy is E_1 , when it is moved to the reservoir, the energy changes to E_2 .

The capture rate is obtained by integrating the differential transitional rates over the interacting band:

$$dk_{12}(E) = c_p(E)f_p(E)g_p(E)dE$$
 (3.7)

$$k_{12} = \int_{-\infty}^{E_v} c_p(E) f_p(E) g_p(E) dE$$
(3.8)

where f_p is the hole occupancy probability given by the Fermi-Dirac distribution, g_p is the density of states and c_p is the capture coefficient.

The emission rate is given by:

$$dk_{21}(E) = e_p(E)f_n(E)g_p(E)dE$$
 (3.9)

$$k_{21} = \int_{-\infty}^{E_v} e_p(E) f_n(E) g_p(E) dE$$
(3.10)

where $f_n(E) = 1 - f_p(E)$ is the electron occupancy probability and e_p the emission coefficient.

Assuming thermal equilibrium, the probability of the defect being neutral, $p_1 = f(E_1)$, and capturing a carrier at a specific energy must be balanced with the probability

of it being charged, p_2 , and emitting a carrier at that energy:

$$p_1c_p(E)f_p(E) = p_2e_p(E)f_n(E)$$
 (3.11)

Assuming that the hole capture from the valence band occurs without a barrier, the probability of charge capture is given by:

$$c_p \approx v_{\rm th} \sigma$$
 (3.12)

where $v_{th} = \sqrt{\frac{8k_BT}{\pi m^*}}$ is the thermal velocity of the carriers in the reservoir, k_B is the Boltzmann constant, T is the lattice temperature, m^* is the effective mass of the carrier, and σ is the capture cross section, that is, the effective area within which a charge has to pass to be captured by a defect [80].

The emission probability e_p can be related to the capture probability c_p and the energy of the captured carrier E_1 :

$$e_p(E) = c_p(E) \frac{p_1}{1 - p_1} \frac{1 - f_n(E)}{f_n(E)} = c_p(E) e^{\left(-\beta(E_1 - E_f)\right)} e^{\left(\beta(E - E_f)\right)}$$
(3.13)

where $\beta = (k_B T)^{-1}$ and E_f is the Fermi energy. Assuming Boltzmann statistics, the probability *p* can be expressed as:

$$p = N_{\rm v} e^{\left(\beta \left(E_{\rm v} - E_{\rm f}\right)\right)} \tag{3.14}$$

where N_v is the valence band weight.

By approximating all carriers to be concentrated at the band edges, which is known as band edge approximation [66], the capture and emission rates can be expressed as:

$$k_{12} = p v_{\rm th} \sigma \tag{3.15}$$

$$k_{21} = N_v v_{\rm th} \sigma e^{(\beta (E_v - E_1))}$$
(3.16)

Capture and emission rates can be extended to oxide defects by including a Wentzel-Kramers-Brillouin tunneling coefficient λ in σ to account for elastic tunneling between the defect and the carrier reservoirs. However, this model cannot accurately describe the bias-dependence or temperature-dependence of defects observed during BTI characterization experiments, and modified equations must be used to model defects located energetically above or below the band gap.

3.2.2 Kirton Uren model

Due to its limited ability to explain the temperature dependence observed in measurements related to oxide defects, the SRH model does not provide an accurate framework to describe the experimental measurements related to charge trapping processes. This suggests the presence of a non-radiative multi-phonon (NMP) process. In response,



Figure 3.3. The configuration coordinate diagram depicts the Kirton-Uren model, where a defect interacts with the reservoir through electron exchange. The position of the parabola is determined by the energy of the carrier. The thermal activation involved in the capture process is represented by the capture barrier, denoted as ΔE_B . In the diagram, the empty defect with an electron in the conduction band is depicted in black, while the captured electron at the defect site is shown in grey. The dashed line represents the energy zero of the system, corresponding to the empty defect with an electron at the Fermi level. This diagram is based on [80, 50].

Kirton and Uren [80] developed an improved approach that considers structural relaxation of defects. They recognized that multi-phonon processes were involved in the charge trapping reactions to explain the temperature dependence of charge trapping at oxide defects. To take this into consideration, they introduced a phenomenological Boltzmann factor into the effective capture cross section [80]:

$$\sigma = \sigma_0 \lambda e^{(-\beta \Delta E_{\rm B})}.\tag{3.17}$$

In this equation, $\Delta E_{\rm B}$ represents the energetic capture barrier between the two equilibrium positions of the defects charge states, as shown in Figure 3.3. By introducing this additional term, the Kirton-Uren model is capable of modeling the temperature dependence of 1/f noise and the average charge transition times. However, it should be noted that the Kirton-Uren model introduces a correlation between the capture rate and the emission rate, which is not observed during BTI characterization experiments. Additionally, the barrier is independent of gate bias and, therefore, fails to describe the bias dependence of charge trapping accurately.

3.3 Non-Radiative Multi-phonon Model

The NMP model [81, 82, 76] has been developed to extend the limitations of the aforementioned models in describing bias-dependent phonon-assisted charge transfer from and to oxide defects. Kirton Uren model is inadequate for describing the bias dependence, as it does not consider the shift of the trap level of the defect due to the



Figure 3.4. Potential energy surfaces (PESs) with a parabolic shape for the charged (depicted in grey) and neutral state (depicted in black) as employed in the NMP defect model, accompanied by the model's parameters. This representation is recreated based on [50].

gate bias, which is observed in measurements.

3.3.1 The 2-State Non-Radiative Multi-Phonon Model

The process of charge transition in a physical system involving electrons and nuclei can be described using the NMP model. The model divides the transition into an electronic part and a vibrational part, since electrons move much faster than the nuclei, using the Born-Oppenheimer approximation [83]. To accurately describe the entire system in the physical defect model surrounding phonon bath needs to be included with all the involved electrons and nuclei. This enables an accurate description of the structural reconfiguration that occurs at the defect site as a result of changing charge states and the electrostatic shift of the trap level due to the oxide field. The process of deformation involves the emission and absorption of phonons, which must be taken into account in the model to ensure an accurate representation of the system.

The transition from one state to another is associated with an energetic barrier, which is determined by a transition state on the potential energy surface (PES), i.e. the highest point along the minimum energy transition path. PES is a complex 3N-dimensional surface for a system with N atoms. Accurate PES of the charge states of a defect structure can be computed by employing DFT, but to obtain a usable model, the multidimensional PES is reduced to one dimension along a reaction coordinate, described by the lowest energy path between the states. Finally, in this approach, PES is substituted with a Taylor expansion centered around the minimum, and this expansion is limited to the quadratic term. This method allows for the depiction of defect states as harmonic oscillators. Consequently, a configuration coordinate diagram is created, featuring parabolic potential energy curves, as illustrated in Figure 3.4.

The Born-Oppenheimer approximation, Frank-Condon principle, and Fermi's golden rule allow the transition rate from one state to another to be expressed as $k_{ij} = A_{ij}f$ [84, 85, 86]. Here, A is the electronic matrix element between the initial and final electronic states, and f is the line shape function. However, since the electronic matrix element cannot be calculated for the systems of interest, it is commonly approximated using a WKB tunneling factor and a prefactor. The line shape function can be calculated,

but is often simplified by neglecting nuclear tunneling [87] below the highest point in the minimum energy path. This leads to transitions occurring at the intersection point of the PES, and the line shape function is written as a Dirac function at the energy of intersection in the classical limit [88]. The energy barrier between the two states can be determined from the intersection of the two PESs. In the semiclassical regime, the transition can be modeled using a Boltzmann factor based on this barrier energy, provided tunneling below the barrier can be ignored.

To model the system, it is necessary to parameterize the two parabolas. Standard parameterization involves the ratio of curvatures $R = \sqrt{c1/c2}$, the relaxation energy $S\hbar\omega$, the Huang-Rhys factor *S*, and the ground state energies E_1 and E_2 . Note that this parameterization eliminates the reaction coordinate between the states, which is necessary for nuclear tunneling calculations, but not required in the semi-classical approximation. Using these parameters and $E_{21} = E_2 - E_1$, it is possible to calculate the energy barrier between state 1 and state 2.

$$\mathcal{E}_{12} = \frac{S\hbar\omega}{(R^2 - 1)^2} \left(1 - R\sqrt{1 + \frac{S\hbar\omega + (R^2 - 1)E_{21}}{S\hbar\omega}} \right)^2$$
(3.18)

If R = 1, there is a singularity, and the energy can be calculated using:

$$\mathcal{E}_{12} = \frac{(S\hbar\omega + E_{21})^2}{4S\hbar\omega} \tag{3.19}$$

The backward barrier can be calculated using:

$$\mathcal{E}_{21} = \mathcal{E}_{12} - E_{21} \tag{3.20}$$

For the two-state NMP defect model, the capture and emission rates can be calculated using the following equations, with the valence band and band edge approximation:

$$k_{12} = p v_{\rm th} \sigma_0 \lambda e^{(-\beta \mathcal{E}_{12})} \tag{3.21}$$

$$k_{21} = N_V v_{\rm th} \sigma_0 \lambda e^{(\beta (E_V - E_1))} e^{(-\beta \mathcal{E}_{21})}$$
(3.22)

where *p* is the hole concentration in the channel, v_{th} is the thermal velocity, σ_0 is the capture cross-section, λ is the Debye length, β is the reciprocal of the thermal energy, E_V is the energy at the valence band edge, and N_V is the density of states at the valence band edge.

3.3.2 The 4-State Non-Radiative Multi-Phonon Model

Single defects can exhibit more than one characteristic dwelling times within the same charge state. Consequently, a single state needs to be splitted into two, as demon-



Figure 3.5. The four-state NMP model characterizes the capture and emission times of individual defects in response to changing biases and temperatures. Broadly, the model considers two stable states, 1 (neutral) and 2 (charged), as well as two meta-stable states, 1' and 2'. It is crucial to consider these states when trying to accurately represent the emission times and other related characteristics that are dependent on bias [76].

strated by single-defect measurements like RTN and time dependent defect spectroscopy (TDDS). In RTN measurements, this phenomenon is observed as anomalous RTN (aRTN), where inactive periods of the defect alternate with active periods of RTN signal [25]. Similarly, TDDS measurements have shown that individual defects may vanish for several measurements and then reappear. These empirical observations, combined with insights from DFT calculations for potential defect candidates, have led to the development of the four-state defect model [60, 89].

The four-state defect model described by a Markov chain, seen in Figure 3.5, considers defects to be both meta-stable and stable for both charged (depicted in red) and neutral charge states (depicted in grey). The transitions between the neutral and charged states, i.e., $1 \leftrightarrow 2'$ and $1' \leftrightarrow 2$, are modeled as NMP transitions. On the other hand, the transitions between states of the same charge are modeled by purely thermal reactions with energetic barriers of constant height. The resulting energy profiles along the multiple reaction paths of such a defect are schematically illustrated in Figure 3.6.

Although the 2-state NMP theory is efficient for describing BTI on large-area devices where large ensembles of defects are at the same time capturing or emitting charge, certain phenomena cannot be explained with only two states. An example of this is the observation of anomalous RTN (aRTN), in which the RTN signal is interrupted for longer periods [25]. This behavior can only be explained by introducing an additional third state [3]. A defect captures and emits a high frequency charge transitioning between states 1 and 2. After a certain time, a transition between state 2 and 2' occurs, and the defect stays for a comparably long period in state 2'. By utilizing the intricate potential energy surfaces depicted in Figure 3.6, it becomes possible to represent the two NMP and the two thermal transition rates [76]. These rates facilitate the determination of the total capture and emission time required to transition from one stable state to another.



Configuration coordinate

Figure 3.6. The four-state NMP model characterizes the capture and emission times of individual defects in response to changing biases and temperatures. Broadly, the model takes into account two stable states, 1 (neutral) and 2 (charged), as well as two meta-stable states, 1' and 2'. These states are essential for accurately depicting the bias-dependent emission times and other associated attributes [76].

Applying Markov theory, the probability p_i of a defect being in state i can be calculated, which makes it feasible to conduct reliability simulations.

According to this model, a defect in a device can switch between charged and neutral states through two different routes. This helps to explain complex behaviors in defect capture and release times, as seen in RTN and TDDS measurements. This approach distinguishes between two different type of traps, namely fixed traps and switching traps [90]. Fixed traps have capture times that change with bias, but their release time might not depend on bias. This is because they use a specific pathway $(1 \rightarrow 2' \rightarrow 2)$ for both capture and release. The release time here mainly depends on a constant barrier between two states (2 and 2'). Switching traps, however, capture charges using the same pathway as fixed traps but release them through a different route $(2 \rightarrow 1' \rightarrow 1)$, resulting in bias-dependent rates for both capture and release. This approach has been used to understand the charge trapping behavior of various defects in different types of devices, including traditional silicon-based ones [76], those with high- κ gate stacks, and even 2D devices [91].



CHAPTER 3. PHYSICAL MODELING OF CHARGE TRANSFER REACTIONS INVOLVING DEFECTS

30

Chapter 4

Reliability Simulations

Contemporary TCAD simulation tools, such as Minimos-NT [92, 93] and Sentaurus Device [94], incorporate a multi-state non-radiative multiphonon (NMP) model [76] to account for the influence of charge trapping on device performance and lifetime. However, these models often demand a comprehensive data set for precise modeling, rather than relying on just a few data points. Additionally, an excessively complex modeling approach might distract from the fundamental physics of charge trapping, which can often be adequately represented by a simplified 2-state NMP model [95, 96].*Parts of this chapter are based on the authors' previous publications* [*KTJ3*, *KTJ4*] *and the PhD thesis of Michl* [51] *and Rzepa* [66].

4.1 Compact Physics Framework (Comphy)

In line with this perspective, the Comphy (compact-physics) framework [66, 97, KTJ3, 95], which was jointly developed by imec and the Institute for Microelectronics at TU Wien, is used as the main simulation tool of this thesis. Comphy is a streamlined Python package aimed at simulating a range of charge-trapping related reliability concerns based on the 2-state NMP model within a concise 1D reliability simulator. It has been shown that Comphy can physically explain the average BTI degradation observed in large area MOSFETs, requiring a smaller set of defect parameters. This framework is particularly suitable for calculating BTI in SiO₂ and SiON devices which are studied in this thesis as a large number of defects can be efficiently handled, and physical defect parameters can be extracted by using established material parameters. The following section outlines the main physical models that provide the electrostatic quantities necessary to compute the charge transfer kinetics and transient ΔV_{th} .

4.2 Electrostatics

The depiction of electrostatics is crucial for every physical degradation model. This section will initially revisit the fundamentals of semiconductor modeling. Following that, these principles will be applied to resolve the electrostatics in a one-dimensional (1D) MOS device.

4.2.1 Effective Density of States

The effective density of states in the conduction and valence bands, denoted as N_C and N_V respectively, can be expressed as follows [98]:

$$N_{\rm C} = 2 \left(\frac{2\pi m_{n,\rm eff} k_{\rm B} T}{h^2}\right)^{3/2} M_{\rm C},$$
(4.1)

$$N_{\rm V} = 2 \left(\frac{2\pi m_{p,\rm eff} k_{\rm B} T}{h^2}\right)^{3/2},$$
(4.2)

where M_C indicates the number of equivalent conduction band minima, $m_{p,eff}$ is the effective mass of the valence band of silicon and $m_{n,eff}$ is the effective mass of the conduction band of silicon [99].

4.2.2 Carrier Concentrations at Thermal Equilibrium

The intrinsic carrier concentration in a semiconductor n_i , is determined using the formula:

$$n_i = \sqrt{N_{\rm C} N_{\rm V}} e^{-\frac{E_{\rm G}}{2k_{\rm B}T}},$$
 (4.3)

the electron concentration *n*, is calculated from the $N_{\rm C}(E)$ across energy levels and their corresponding occupation probability f(E), as follows:

$$n = \int_{E_{\rm C}}^{\infty} N_{\rm C}(E) f(E) dE, \qquad (4.4)$$

where $E_{\rm C}$ is the conduction band minimum energy.

The occupation probability of states is given by the Fermi-Dirac distribution function $f_F(E)$, expressed as:

$$f_F(E) = \frac{1}{1 + e^{\frac{E - E_F}{k_B T}}},$$
(4.5)

The thermal equilibrium electron concentration, n_0 , with respect to the effective density of states in the conduction band, N_C , is represented by:

$$n_0 = \frac{N_{\rm C}}{2} \sqrt{\frac{\pi}{k_{\rm B}T}} \int_{E_{\rm C}}^{\infty} \sqrt{E - E_{\rm C}} \left(\frac{1}{1 + e^{\frac{E - E_{\rm F}}{k_{\rm B}T}}}\right) dE,\tag{4.6}$$

Since this integral cannot be evaluated analytically, approximations are generally used, with the Boltzmann and Joyce-Dixon [100] being the most common ones.

4.2.3 Fermi Level at Thermal Equilibrium

The thermal equilibrium Fermi level (E_{F_0}) in the channel of a MOS device that's been doped can be determined in the following manner : Suppose all dopants in the

semiconductor are ionized; in this case, charge neutrality exists if

$$n_0 + N_{\rm A} = p_0 + N_{\rm D},$$
 (4.7)

where n_0 and p_0 are the intrinsic concentrations, N_A is the acceptor concentration and N_D is the donor concentration. For non-degenerate semiconductors in thermal equilibrium, where $n_{i^2} = n_0 p_0$, for n-type semiconductors where $N_D > N_A$, we have [98]:

$$n_0 = \frac{(N_{\rm D} - N_{\rm A})}{2} + \sqrt{\left(\frac{(N_{\rm D} - N_{\rm A})}{2}\right)^2 + n_{\rm i^2}},\tag{4.8}$$

$$p_0 = \frac{n_{\rm i}^2}{n_0},\tag{4.9}$$

$$N_{\rm D} - N_{\rm A} = n_{\rm i} \exp\left[\frac{(E_{\rm F_0} - E_{\rm i})}{(k_{\rm B}T)}\right],$$
 (4.10)

and similarly for p-type semiconductors, where $N_A > N_D$:

$$n_0 = \frac{n_{\rm i}^2}{p_0},\tag{4.11}$$

$$p_0 = \frac{(N_{\rm A} - N_{\rm D})}{2} + \sqrt{\left(\frac{(N_{\rm A} - N_{\rm D})}{2}\right)^2 + n_{\rm i}^2},\tag{4.12}$$

$$N_{\rm A} - N_{\rm D} = n_{\rm i} \exp\left[\frac{(E_{F_0} - E_{\rm i})}{(k_{\rm B}T)}\right].$$
 (4.13)

Hence, the intrinsic Fermi level E_i , can be used to compute E_{F_0} as:

$$E_{\rm i} = \frac{(E_{\rm C} + E_{\rm V})}{2} + \frac{k_{\rm B}T}{2} \ln\left(\frac{N_{\rm V}}{N_{\rm C}}\right). \tag{4.14}$$

4.2.4 Electrostatics of a 1D MOS Structure

To determine the effective trap level of a defect relative to the carrier reservoirs in the channel, it is necessary to know the relationship between the applied gate bias and the surface potential at the interface between the channel and the oxide. Assuming that a uniform doping concentration and charge neutrality exist, deep in the bulk semiconductor (far from the interface), one can use the approximation of the surface charge Q_S as a function of the surface potential ϕ_S given by [98]:

$$Q_{\rm S} = \pm \frac{\sqrt{2}k_{\rm B}T}{qL_{\rm D}} \sqrt{(e^{-\frac{\phi_{\rm S}}{k_{\rm B}T}} + (k_{\rm B}T)\phi_{\rm S} - 1) + \frac{n_0}{p_0}(e^{\frac{\phi_{\rm S}}{k_{\rm B}T}} - (k_{\rm B}T)\phi_{\rm S} - 1)},\tag{4.15}$$

where the positive sign refers to $\phi_{\rm S} > 0$, while the negative sign is used for $\phi_{\rm S} < 0$. The abbreviation $L_{\rm D} = \sqrt{k_{\rm B}T\epsilon_0\epsilon_{\rm r,chan}/p_0q^2}$ is the Debye length for holes, where $\epsilon_{\rm r,chan}$ is the relative permittivity of the channel. The channel concentrations for electrons n and holes p are calculated using the Joyce Dixon approximation [100]. The temperature dependence of the Si bandgap is computed using the model proposed by Bludau [101]. The effective masses for the valence band and the conduction band are taken from the models proposed by Lang et al. [102] and Green [99], respectively.

One can derive an equation for surface potential ϕ_S by using electrostatic considerations [97]. The equation for an ideal, defect-free, MOS device can be written as:

$$\frac{Q_{\rm S}(\phi_{\rm S})}{C_{\rm ox}} + \phi_{\rm S} - V_{\rm G} + \Delta E_{\rm W}/q = 0, \tag{4.16}$$

Here, ΔE_W represents the work function difference defined as:

$$\Delta E_{\rm W} = \Delta E_{\rm W,gate} - E_{\rm W,chan} \tag{4.17}$$

In this approach, while ΔE_W is an input parameter, it simultaneously depends on $E_{W,chan}$, which is clearly determined by the doping concentrations, also considered as input quantities. Consequently, the intrinsic work function difference $\Delta E_{W,0}$ is given by:

$$\Delta E_{\rm W,0} = \Delta E_{\rm W} - E_{\rm E,0} \tag{4.18}$$

to derive ϕ_S , the following equation is solved through an iterative approach:

$$\frac{Q_{\rm S}(\phi_{\rm S})}{C_{\rm ox}} + \phi_{\rm S} - V_{\rm G} + \Delta E_{\rm W,0}/q + E_{\rm F,0}/q = 0, \tag{4.19}$$

The capacitance of the oxide is given by:

$$C_{\rm ox} = \frac{\epsilon_0 \epsilon_{r,\rm ox} W L}{d_{\rm ox}},\tag{4.20}$$

where $\epsilon_{r,ox}$ denotes the relative permittivity of the oxide, *W* and *L* are the channel width and length, and d_{ox} is the thickness of the oxide. Equation (4.16) can be solved by using an iterative method like a Newton solver method, to obtain ϕ_S .

The trapped charges in the oxide can also be considered in the equation, which leads to the extension of Equation (4.16) as:

$$\frac{Q_{\rm S}(\phi_{\rm S})}{C_{\rm ox}} + \phi_{\rm S} - V_{\rm G} + \Delta E_{\rm W,0}/q + E_{\rm F,0}/q + V_{\rm traps} = 0, \tag{4.21}$$

In this context, V_{traps} signifies the voltage shift due to oxide charge presence. Determining the surface potential ϕ_S is vital for surface potential-focused compact models, as it facilitates the calculation of the potential curve throughout the defect-free insulator. Using this data, the change in trap levels can be computed, and charge capture and emission rates can be assessed through the 2-state NMP theory. For a collection of defects, an average captured charge can be determined at each time interval, contingent on defect occupancy and density. The aggregate of all charge states results in a threshold voltage shift, producing a transient ΔV_{th} feature that can be contrasted with experimental findings. Evaluating the surface potential ϕ_{S} is a crucial component in the precise modeling of MOSFET behavior.

4.3 Occupation Probabilities

The previous section was dedicated to discussing the computation of device electrostatics. Each defect has a probability of being in either a charged or neutral state, as dictated by the transition rates derived from 2-state NMP theory. These rates enable the calculation of each trap's occupancy by solving master equations. These equations consider the interaction of each defect with various charge reservoirs r, including the valence and conduction bands of the channel and the gate in a MOSFET. The probabilities of a defect being in the initial or final state are represented as $p_{i,j}$, and the corresponding differential equations are as follows:

$$\frac{dp_{i(t)}}{dt} = -p_i \sum_{r=1}^{N_{\rm R}} k_{ij,r} + p_j \sum_{r=1}^{N_{\rm R}} k_{ji,r}, \qquad (4.22)$$

$$\frac{dp_{j(t)}}{dt} = p_i \sum_{r=1}^{N_{\rm R}} k_{ij,r} - p_j \sum_{r=1}^{N_{\rm R}} k_{ji,r}.$$
(4.23)

The equations must satisfy the condition $p_i + p_j = 1$. In this context, $k_{ij,r}$ represents the electron emission rate, while $k_{ji,r}$ signifies the electron capture rate of the defect as it interacts with carrier reservoir r. For a defect, the total rates $k_{ij,r}$ and $k_{ji,r}$ are the cumulative sum of the rates across all N_R carrier reservoirs. In the Comphy framework, these carrier reservoirs include the valence band of the channel, the channel's conduction band, and the metal gate.

The solution to the differential equation system is expressed as:

$$p_j(t) = \frac{k_{ij}}{k_{ij} + k_{ji}} + \left(p_j(0) - \frac{k_{ij}}{k_{ij} + k_{ji}}\right) e^{-t(k_{ij} + k_{ji})},$$
(4.24)

$$p_i(t) = 1 - p_j(t).$$
 (4.25)

These equations can be employed to compute the perturbation of the electrostatic potential across the gate stack the impact of the defects on the potential, as demonstrated by the charge sheet approximation (CSA) [103, 104]:

$$V_{\rm traps} = -\frac{q}{C_{\rm ox}N_{\rm T}} \sum_{n=1}^{N_{\rm T}} p_{\rm j}(t) (1 - \frac{x_{{\rm T},n}}{d_{\rm ox}})$$
(4.26)

Subsequently, the device's surface potential is determined through Equation 4.19, which facilitates the computation of potential shift relative to the initial time step, thus determining the threshold voltage shift. The total threshold voltage shift is dependent on the probability of a charge transition, which is affected by factors like defect parameters, the applied gate voltage, and temperature, as illustrated in Figure 4.1.

4.4 Distribution of Defect Parameters



Figure 4.1. Comphy employs an effective 2-state NMP model to describe the charge transition of a pre-existing oxide defect. This transition from state *i* to state *j* occurs by overcoming the energy barrier ϵ_{ij} . The model's harmonic potential energy surfaces are characterized by the trap level E_{T} , the relaxation energy E_{R} , and the curvature ratio *R*. For the computation of the tunneling factor, the spatial coordinate x_{T} is employed. Figure taken from [KTJ4]



Figure 4.2. A defect can be characterized by the following parameters: relaxation energy (E_R) , trap level (E_T) , spatial position in the oxide (x_T) , curvature ratio (R), and configuration coordinate offset (ΔQ) . It is important to note that *R* and ΔQ are associated with the energy levels, while x_T is assumed to have a uniform distribution. Additionally, the energy levels follow a normal distribution within the defect bands. Recreated from [51]

Each defect is described by a set of model parameters, including the relaxation

energy E_R , curvature ratio R, trap level E_T , spatial position inside the oxide x_T , and the configuration coordinate offset ΔQ (if a quantum mechanical 2-state NMP transition rate is used). In devices with large surface areas, a group of defects is typically assumed to be spread across a defect band. The trap levels of the defects are assumed to follow a normal distribution, while the relaxation energies and spatial positions are assumed to follow normal and uniform distributions, respectively [66]. Since the curvature of the potential energy curves (PECs) is correlated with the relaxation energy, R need not be treated as a stochastically distributed unit. Furthermore, a uniform spatial defect concentration N_T across parts of the dielectric is assumed. The relevant parameters are illustrated in Figure 4.2. Defect sampling can be carried out through Monte-Carlo sampling or a more efficient approach involving the computation of the mean degradation of the defect band by sampling the parameters on a grid and introducing a weighting scheme.

4.4.1 Effective Single Defect Decomposition

The utilization of Gaussian trap bands has been successful, but it requires larger relaxation energies than those obtained from DFT calculations of appropriate oxide defect candidates, which is a disadvantage. This can lead to unreasonably long charge transition times, as exemplified by the shallow SiO₂ trap band of the 28 nm technology described in [66]. In Figure 4.3, the majority of all sampled defects exhibited transition times exceeding 10⁷ s, which is not experimentally accessible. The smaller time constants are associated with the low-energy end of the relaxation energy distribution. To address this problem, an alternative method called effective single defect decomposition (ESiD) has been developed, which calibrates the model to experimental data without assuming any defect parameter distributions, as outlined in [KTJ3]. The ESiD algorithm allows for an efficient extraction of trap parameters from measurement data, which can then be compared to DFT simulations.

When using the ESiD method, the total shift in threshold voltage (ΔV_{th}) is expressed as the sum of the threshold voltage shifts caused by each individual defect (δV_{th}):

$$\Delta V_{\rm th} = \sum_{E_{\rm T}, E_{\rm R}, x_{\rm T}, \Delta Q} N(t; E_{\rm T}, E_{\rm R}, x_{\rm T}, \Delta Q) \delta V_{\rm th}(E_{\rm T}, E_{\rm R}, x_{\rm T}, \Delta Q)$$
(4.27)

Here, $N(E_T, E_R, x_T, \Delta Q)$ represents the weight of each defect. Instead of assuming Gaussian distributions for E_T and E_R , the ESiD algorithm employs a uniform parameter grid sampled for E_T , E_R , x_T , and ΔQ . These parameters form a grid that is considered in the response matrix (δV), where the response δV_{th} is computed at each point. Using the response matrix and defining the observation vector $(\Delta V)_j = \Delta V_{\text{th}}(t_j)$, a non-negative linear least square (NNLS) algorithm can be employed to compute the weights $(N)_i = N(E_{\text{Ti}}, E_{\text{Ri}}, x_{\text{Ti}}, \Delta Q_i)$ such that:

$$(\hat{N}) = \underset{\hat{N} \ge 0}{\operatorname{argmin}} ||(\delta V) \cdot \hat{N} - \Delta V||^2$$
(4.28)

Since this is a mathematically ill-posed problem that can lead to solutions with



Figure 4.3. The use of Gaussian trap bands in the 28 nm technology, as discussed in [66], indicates that only a fraction of the distribution contributes to active defects. The greater part of the defect parameters extracted from the trap band results in charge transition times exceeding 10⁷ seconds (as seen on the left), rendering them experimentally unattainable. These prolonged charge transition periods correlate with relaxation energies ER that are significantly larger than those derived from DFT calculations (as seen on the right). Therefore, they can be considered as artifacts of the Gaussian distributions. The figure is a reproduction from [KTJ3].

physically unrealistic high defect densities, a Tikhonov regularization [105] is added to ensure that the least-square solution results in smoother defect densities:

$$(\hat{N}) = \underset{\hat{N} \ge 0}{\operatorname{argmin}} ||(\delta V) \cdot \hat{N} - \Delta V||^2 + \gamma^2 \hat{N}^2$$
(4.29)

Here, γ is the regularization parameter. In practical applications, the parameter γ needs to be adjusted according to the specific problem at hand. However, as demonstrated in Figure 4.4, a reasonable value can be estimated by plotting the total defect density against the error norm to the experimental data for different γ values, resulting in an L-shaped curve. This curve provides valuable insights: for very small values of γ , the approximation error is remarkably low but at the expense of a significantly high defect concentration. Conversely, if γ is too large, the problem becomes overregularized, leading to a substantial increase in approximation error as γ is further increased. Based on the L-criterion [105], the optimal value for γ is located at the "corner" of this L-shaped curve, striking a balance between accurate representation of the experimental data and maintaining a reasonably small defect concentration.

4.5 Issues Related to Charge Sheet Approximation

Even though Comphy provides a convenient way to evaluate device reliability, the usage of a 1D geometry inherently limits its accuracy. This is due to the CSA, see Equation 4.9, used to model the impact of oxide charges on device electrostatics [104].



Figure 4.4. The influence of the regularization parameter γ is significant. When γ is small, the solution necessitates an excessively high defect density, which is not physically realistic. On the other hand, excessively large values of γ result in rapidly increasing approximation errors. The ideal value for γ resides between these two unfavorable regions. Originally published in [KTJ3]

CSA calculations take into consideration a uniform way of conduction, while in reality a percolative way of conduction takes place [106]. These inaccuracies become more pronounced in the subthreshold regime, where the conductivity is heavily affected by random discrete dopants (RDD) and random trapped charges. Many recent studies have demonstrated that the CSA significantly underestimates the impact of a single defect on device performance [106, 107, KTJ4], which can potentially result in a slight overestimation of trap densities. Even for ideal planar devices, this approximation does not accurately describe the distribution of threshold voltage shifts caused by individual trap charges. For more complex device geometries, such as FinFETs or gateall-around (GAA) FETs, the modeling of transmission coefficients and the strongly inhomogeneous electric fields and carrier concentrations lead to further inaccuracies in the 1D approximation [95].

Even though CSA has proven to work pretty effectively in numerous cases, a thorough statistical analysis of single defects and their impact on device performance is needed for accurate extraction of the parameters that influence the impact of defects, like geometry and dopant concentration. Kaczer et al. [48], have proposed the following expression for the impact of a single defect in the ΔV_{th} , mentioned as η :

$$\eta \approx \frac{t_{\rm inv} \times N_A^a}{A},\tag{4.30}$$

where t_{inv} is the oxide thickness corresponding to capacitance in inversion, N_A the channel doping and A the area of the device channel. The exponent a has been observed around 0.5.

However, in references in the literature where statistical analysis is performed, it has also been observed that the area dependence used in the CSA is not valid under

all operating conditions [26, 108]. The nature of percolative conduction appears to be the main reason for these observations [109, 110, 111]. This is especially pronounced in devices with smaller gate areas, where charge trapping at oxide defects can significantly reduce the source-drain current, even a single charged defect can have this effect due to its influence on the percolation path in the channel. In such devices, the lower capacitance means that each defect exerts a greater influence on the surface potential and, consequently, the shift in threshold voltage. Another issue with the uniform way of conduction expected by the CSA is its inability to explain the distribution of traps observed in experiments. The next chapter will delve into the details of statistical methods that to explain the impact of traps on device behavior which were also used for the work of this thesis.

Chapter 5

Single Defect Statistics

5.1 Introduction

As discussed in the previous chapter, the influence of individual defects varies across different technologies, with factors such as random dopant fluctuations playing a significant role. Consequently, approximations like the CSA, while effective in estimating the impact in many cases, can introduce inaccuracies.

In the realm of nanoscale devices, where modern devices are scaled down, only a few defects are present in each device, and each defect significantly affects the device's operation. To better comprehend degradation mechanisms like BTI and RTN in these scaled devices, it is beneficial to analyze the impact of a small group of individual charged traps. Each device can be characterized by the active number of defects in the measurement window, denoted as N_T , and the average impact of a single defect in Δv_{th} , denoted as η . Only the occupied charged defects are assumed to affect the channel current. The occupation of each defect, determined by its voltage and temperature-dependent capture and emission times, also contributes differently to the channel current.

Due to their discrete nature, these small groups of defects can be examined, and statistical models can be developed to estimate their impact. This chapter will present the distributions primarily used to describe the quantity and influence of individual defects. Utilizing these distributions, equations that describe the overall impact of defects in a device are established, allowing the direct extraction of the means N_T and η from the measured ΔV_{th} traces, eliminating the need for separate examination of individual defects. *Parts of this chapter are based on the work of Kaczer et al.* [48, 112] the *authors' previous publications* [KT]5].

5.2 Statistical Distributions of Defects

A set of MSM measurements obtained from devices with the same geometry $(W \times L = 135 \text{ nm} \times 350 \text{ nm})$ is displayed in Figure 5.1(left). The discrete steps (charge transition events) are extracted with the Canny edge detection algorithm [113], which

is integrated into the data analysis framework used within the thesis. The resulting extracted step heights for different devices are collected and statistically analyzed.

These collected amplitudes of step heights, denoted as Δv_{th} , are used to construct a cumulative distribution function (CDF) for each particular geometry. The convention of a complementary CDF (CCDF) normalized to the N_{T} is employed in this work, as shown in Figure 5.1 (middle). This approach enables the direct extraction of N_{T} from the CCDF distribution, allowing for the comparison of multiple geometries within a single plot.

It is worth noting that the distributions obtained from BTI measurements have been found to approximately follow an exponential behavior in the literature [106, 24, 32]. This exponential distribution has also been repeatedly reported for RTN amplitudes and confirmed by various theoretical studies, with the responsible transition times exhibiting the same behavior [32, 114, 115]. These similarities have reinforced the link between RTN and BTI, providing a common framework to study both phenomena.

The exponential distribution of step heights can be generally attributed to the nonuniform conductive paths between source and drain, which are affected by the random dopants of the substrate and other variability sources [106].



Figure 5.1. The left part of the figure illustrates typical measure-stress-measure (MSM) sequences obtained from 40 devices of equal dimensions ($W \times L = 135 \text{ nm} \times 350 \text{ nm}$). Using the Canny algorithm, integrated into the data analysis framework [113, 116] used in this work, discrete steps representing charge transition events are extracted from the recovery traces, leading to the creation of a corresponding distribution. The middle panel demonstrates that the CCDF adheres to an exponential distribution while, for comparison, a uniform distribution obtained from the charge sheet approximation (CSA) is also included. Finally, the right panel exhibits a quantile-quantile (Q-Q) plot, comparing the distribution of the number of steps per device to a Poisson distribution. This plot validates that the number of steps per device can be accurately described by a Poisson distribution. Figure taken from [KTJ5]

The respective exponential probability distribution function (PDF) can be described by [112, 48]

$$f(\Delta v_{\rm th}) = \frac{1}{\eta} \exp\left(-\frac{\Delta v_{\rm th}}{\eta}\right),\tag{5.1}$$

where η is the mean threshold voltage shift caused by a charge transition event of a

defect. From the PDF, the corresponding CDF can be calculated

$$F(\Delta v_{\rm th}) = \int f(\Delta v_{\rm th}) d\Delta v_{\rm th} = 1 - \exp\left(-\frac{\Delta v_{\rm th}}{\eta}\right).$$
(5.2)

By normalizing the complementary CDF to the average number of defects per device, the following relation can be obtained

$$(1 - \text{CDF}) \times \frac{N_{\text{defects}}}{N_{\text{devices}}} = N_{\text{T}} \exp\left(-\frac{\Delta v_{\text{th}}}{\eta}\right),$$
 (5.3)

where N_{defects} is the total number of active defects observed in the measuring time window, N_{devices} is the number of tested devices, and N_{T} is the average number of active defects per device. The impact of an individual defect on the threshold voltage of a device (Δv_{th} , denoted with a lowercase "v" and referred to as step height) depends on the position of the defect in the oxide.

The number of defects per device generally follows a Poisson distribution [112]. To verify this assumption, the number of defects per device are extracted for all the tested traces and the quantiles of the extracted distribution are evaluated, see Figure 5.1 (right). As can be seen, the results confirm that the observed number of defects per device is Poisson distributed.

Note that the higher number of electrically active defects on pMOS devices can be observed directly by the maximum value of the normalized CCDFs. The vertical dashed lines show the maximum limit the CSA would give for this device type. However, as shown, the distribution based on the CSA significantly underestimates the experimentally observed average impact of the defects η . This underestimation is casually related to the neglection of the impact of percolative conduction in weak inversion regimes, which is associated with the dopants of the substrate [24]. As a result, the CSA can lead to overly pessimistic results for defect densities when extracted from ΔV_{th} .

5.3 Total ΔV_{th} **Distribution**

If the lateral locations of *n* successively trapped charges are assumed to be uncorrelated, the overall threshold voltage shift will be [112, 48]

$$\Delta V_{\rm th} = \sum_{i=1}^{n} \Delta v_{\rm th,i}.$$
(5.4)

The distribution of ΔV_{th} can be expressed as a convolution of individual exponential distributions [Equation 5.1], with the PDF and the CDF respectively described by

$$f_{\eta,n}(\Delta V_{\rm th}) = \frac{n}{n!} \frac{\Delta V_{\rm th}^{n-1}}{\eta_n} \exp\left(-\frac{\Delta V_{\rm th}}{\eta}\right),\tag{5.5}$$

and

$$F_{\eta,n}(\Delta V_{\rm th}) = 1 - \frac{n}{n!} \Gamma\left(n, \frac{\Delta V_{\rm th}}{\eta}\right).$$
(5.6)

Here, Γ is the incomplete gamma function.

The population of stressed devices typically contains a varying number *n* of visible oxide defects in each device, with this number following a Poisson distribution [24, 112]. Assuming that the small fraction of the device population exhibiting RTN around V_{th0} can be ignored to avoid complex statistics [117, 118], the total ΔV_{th} distribution can be determined by summing distributions F_n , weighted by the Poisson probability:

$$p_{N_{\rm T},n} = \frac{e^{-N_{\rm T}} (N_{\rm T})^n}{n!}.$$
(5.7)

In Equation 5.7, N_T is the mean number of defects in the FET gate oxide and is related to the oxide trap (surface) density N_{ot} as $N_T = WLN_{ot}$ (i.e., N_T is not an integer). This line of reasoning then results in the total ΔV_{th} CDF given by [112, 48]:

$$H_{\eta,N_{\rm T}}(\Delta V_{\rm th}) = \sum_{n=0}^{\infty} p_{N_{\rm T},n} F_{\eta,N}(\Delta v_{\rm th}) = \sum_{n=0}^{\infty} \frac{e^{-N}(N)^n}{n!} \left[1 - \frac{n}{n!} \Gamma\left(n, \frac{\Delta V_{\rm th}}{\eta}\right) \right].$$
 (5.8)

The corresponding PDF is

$$h_{\eta,N_{\rm T}}(\Delta V_{\rm th}) = e^{-N} \left[\delta(\Delta V_{\rm th}) + \frac{N}{\eta} \exp\left(-\frac{\Delta V_{\rm th}}{\eta}\right) {}_{0}F_{1}\left(2;\frac{N}{\eta}\Delta V_{\rm th}\right) \right], \qquad (5.9)$$

The hypergeometric function ${}_{0}F_{1}(2;x)$ can be expressed in terms of the modified Bessel function I_{1} as ${}_{0}F_{1}(2;x) = x^{-1/2}I_{1}(2\sqrt{x})$. The Dirac delta function $\delta(\Delta V \text{th})$ accounts for the fraction of devices with zero voltage shift [114], which decreases with an increase in N_{T} .

5.3.1 Derived Parameters

The following equations provide a practical way to describe the distribution of total ΔV_{th} , which can be statistically analyzed through its mean and variance. The mean is given by the following equation:

$$\langle \Delta V_{\rm th}(t) \rangle = \eta N_{\rm T}(t) \tag{5.10}$$

where N_T and η are proportional and inversely proportional to the FET gate area A, respectively. The variance of the distribution is given by:

$$\sigma_{\Delta V_{\rm th}}^2(t) = 2\eta^2 N_{\rm T}(t)$$

The relative deviation, $\sigma_{\Delta V_{\text{th}}}/\langle V_{\text{th}}\rangle = (2/N_{\text{T}})^{1/2}$, increases with decreasing gate area and decreases with increasing N_T . The skewness of the distribution is given by:

$$6\eta^3 N_{\rm T}(t) \tag{5.11}$$

and the kurtosis by:

$$12\eta^4 [N_{\rm T}^2(t) + 2N_{\rm T}(t)] \tag{5.12}$$

Expressing N_T and η in terms of more "circuit designer-friendly" parameters $\langle V_{\text{th}} \rangle$ and $\sigma_{\Delta V_{\text{th}}}^2$ gives:

$$N_T(t) = 2\langle \Delta V_{\rm th}(t) \rangle^2 / \sigma_{\Delta V_{\rm th}}^2$$
(5.13)

and

$$\eta = \sigma_{\Delta V_{\rm th}}^2 / [2 \langle \Delta V_{\rm th}(t) \rangle]$$
(5.14)

Hence, the first two moments of a measured total BTI V_{th} distribution are sufficient to extract both N_T and η . The time-dependent variance $\sigma^2_{\Delta V_{\text{th}}}$ can be expressed as a function of η and $\langle V_{\text{th}} \rangle$:

$$\sigma_{\Delta V_{\rm th}}^2 = 2\eta \langle \Delta V_{\rm th}(t) \rangle \tag{5.15}$$

This equation removes the complexity of degradation kinetics and allows degradation to be solely expressed in terms of the average degradation, as seen in Figure 5.2. The technology-dependent parameter η can be obtained through single-emission measurements or measurements of total ΔV_{th} distributions or by considering links with time-zero variance.



Figure 5.2. Experimental data from the largest device in the measurement set show that discrete steps are not observable in larger devices. Therefore, the statistical averages of the defects can be determined using DCM on the ΔV_{th} traces.


Chapter 6

Measurement Methods

This chapter focuses on exploring different measurement techniques utilized to determine the reliability of MOSFET devices, with a particular emphasis on the extraction of device parameters and the evaluation of parameter degradation over time. Initially, an overview of methods for extracting important device parameters, such as the threshold voltage (V_{th}), from electrical characterization is presented. Subsequently, commonly used measurement schemes designed to evaluate device parameter degradation, especially the drift of threshold voltage (ΔV_{th}), are discussed. In addition, the custom-built measurement tools utilized in this research are examined, outlining their benefits and limitations.

6.1 Time-zero Characterization

Time zero properties of MOSFETs generally refer to their initial characteristics and performance metrics when they are manufactured or at the point of initial use. These properties are crucial because they serve as a baseline for understanding how the device will behave and how it could degrade over time under various operating conditions.

6.1.1 Transfer Characteristics

In MOSFET devices, several critical parameters are affected by device operation. The main parameter used in this thesis for device evaluation is the threshold voltage (V_{th}) . It is the minimum bias condition needed for a conductive channel to form between the source and the drain. Another essential parameter is the subthreshold slope (SS), which plays a crucial role in achieving accurate switching between the ON and OFF states within a narrow gate bias range. Additionally, the transconductance (g_m) , which is primarily associated with carrier mobility and correlated with interface defect scattering, is another vital parameter.

The transfer characteristics of a MOSFET device can be used to estimate the timezero properties of a device under test as they describe the relationship between the drain current I_D and the gate bias V_G measured at a fixed drain bias V_D . This relationship can



Figure 6.1. $I_D(V_G)$ measurements for a set of devices, with the channel width W = 135 nm and the length L = 350 nm. Using the constant current method, the threshold voltage V_{th} is extracted at $I_D = I_{ref} \frac{W}{L}$. The reference current I_{ref} is assumed to be 100 nA in this case. The V_{th} values range between 0.8 and 1 V for the identical devices of this technology.

be used to determine the current state of device parameters at the time of measurement. Typically, it is recommended to perform a fast sweep of the gate bias of the Device Under Test (DUT) and measure I_D at each bias point as quickly as possible. This approach aims to minimize the duration of an applied oxide stress field during the sweep, thereby reducing the impact of defects with shorter capture and emission time constants compared to the sweep duration. As a result, fast sweeps help maintain the device's pristine state. Since the theoretical definition of V_{th} , which denotes the exact equilibrium of majority and minority carriers in the conducting channel of a MOSFET [98], cannot be experimentally determined, alternative definitions of V_{th} are needed for its experimental extraction [119, 120]. One common method for determining V_{th} involves applying a constant source-drain current criterion, which is depicted in Figure 6.1. In its simplest form the V_{th} is identified as the gate voltage where:

$$I_{\rm D} = I_{\rm ref} \frac{W}{L} \tag{6.1}$$

 I_{ref} is typically chosen to result in a drain current within the sub-threshold region of the transfer characteristics. Normally it depends on the studied technology and it varies between 1 nA and 200 nA. With this method, V_{th} can be continuously measured by using a feedback loop to maintain a constant I_{D} while tracking V_{G} . The constant current method proves to be beneficial in comparing MOSFETs with varying channel geometries. Due to its ease of application and low extraction variability, this method is suitable for extracting ΔV_{th} over time for wafer-scale device variations, as stated in [121].

Other methods for determining V_{th} from the I_{D} - V_{G} characteristics of a MOSFET device use the transconductance g_{m} . One such approach is the linear extrapolation method, which uses the gate voltage for the maximum transconductance $g_{\text{m,max}}$ to

extrapolate to zero drain current. By subtracting $V_{D/2}$ from the resulting intersection gate bias, V_{th} can be obtained as shown in [122]:

$$V_{\text{th,g}_{m,\text{lin}}} = \frac{I_{\text{D}|g_{m,\text{max}}}}{g_{m,\text{max}}} - V_{\text{G}|g_{m,\text{max}}} - \frac{V_{\text{D}}}{2}$$
(6.2)

Another approach, known as the second derivative method, provides a more accurate comparison with the theoretical value of V_{th} . This method identifies the threshold voltage by locating the gate bias at the minimum point of the second derivative of the logarithmic drain current, as referenced in [123].

6.1.2 Capacitance - Voltage (CV) Measurements for Characterizing MOSFET Devices

Capacitance-voltage (CV) measurements are crucial in obtaining key insights into MOSFET devices, including their inherent properties and defect features. Two primary methods are employed for CV measurements: the ramp method, which involves measuring the displacement current while sweeping the gate voltage, and the impedance method, where a small AC signal is superimposed on the DC gate bias and the AC current is measured. In this section, the emphasis will be on the impedance method, which offers more flexibility in measurement parameters and allows for the observation of small signal conductance.

The impedance method is extensively utilized to gather vital information about the transistor gate stack, such as oxide thickness, doping densities, poly-Si gate-depletion, and permittivity. By altering measurement parameters and comparing the resulting changes in the CV curves' shape, one can extract defect properties. This method involves applying a DC voltage at the gate contact, overlaid with a small-amplitude sinusoidal AC signal, while grounding the bulk contact of a MOSCAP (or all terminals—source, drain, and bulk—for a MOSFET).

The core principle behind a CV measurement, the results of which os illustrated in Figure 6.2, consists of applying a small sinusoidal AC voltage (typically 50 mV at 100 kHz) combined with a discrete varying DC bias to the gate, while grounding the bulk. Simultaneously, the gate voltage and bulk current are measured. Alternatively, the roles of the gate and bulk may be switched, with the voltage applied to the bulk and the gate current measured. These measurements can be performed on either MOSCAP or MOSFET structures. MOSCAP structures offer benefits, such as being effectively onedimensional, which allows for more straightforward and accurate models. However, MOSFETs permit faster gate bias sweeps due to the availability of minority carriers from the source and drain regions. The main distinction between the two structures concerns the supply of minority carriers for inversion.

CV measurements generally involve stepping the gate bias from accumulation to inversion or vice versa to obtain the small signal capacitance and conductance curves (C-V and G-V). The AC signal frequency, ranging from kHz to GHz, is the most critical parameter in this measurement. Typically limited to 1 MHz due to increased



Figure 6.2. The example results from a 100 kHz C-V measurement on nMOS devices display distinct accumulation and inversion branches at the peak values of negative and positive biases, respectively. The capacitance observed during the depletion phase is significantly lower. Additionally, the inversion branch tends to exhibit a steeper slope compared to the accumulation branch. V_{th} can also be estimated by the minimum value of the plot.

measurement effort, the frequency determines how defects contribute to the measured CV curve, as defects unable to follow the signal will not affect the measured capacitance. In MOSCAP structures, the sweep rate compared to the minority response further impacts the measurement outcomes.

CV measurements can complement I_D measurements to study defect-related parameters. Initially, a reference CV curve is measured, and then the device is stressed under well-defined conditions. A second CV curve is recorded, and the variations between the pre- and post-stress CV curves enable the extraction of various defect properties. The difference between subsequent measurements can be connected to changes in the device's defect populations. This technique serves as a potent tool for reliability characterization, offering valuable information about the traps' energetic position. Several options are available for determining defects' contribution to a measurement, including the high-low frequency method, which compares low-frequency measurements to high-frequency measurements, and the single-frequency measurement method, comparing the measured curve to a simulation or calculation. Additionally, measurements can be performed in a MSM-like manner, relating the difference between subsequent measurements to changes in the device's defect populations.

In conclusion, CV measurements can be used alongside $I_D(V_G)$ measurements to offer a valuable method of characterizing MOSFET devices and their defects. Further investigation of this technique may result in the development of enhanced extraction methods and a better understanding of MOSFET device performance and reliability.

6.2 Time-Dependent Characterization

In previous chapters it was discussed how charged defects located between the gate and the conducting channel of a MOSFET can influence the electrostatics at the channel/insulator interface from the sub-threshold region up to inversion. The Coulomb interaction between the trapped charge and the channel carriers can lower the carrier density in the vicinity of the defect, which in turn affects the drain-source conductivity of the device. The impact of a defect on the channel current depends on the resulting perturbation of the current percolation path. This effect is used in various characterization methods for defect detection. The methods discussed in this chapter are based on changes in channel conductivity, which rely on recording the drain-source current at a constant bias and temperature. The circuits used in practice incorporate additional components to switch between multiple amplification ranges, by using different resistors in the feedback loop of an operational amplifier (OPAMP) circuit. Furthermore, the SNR is improved by limiting the frequency range or for circuit stability in the case of the constant-current scheme. In addition, the gate current may also be measured to characterize single interface traps and states [KTJ6].

6.2.1 Measure-Stress-Measure (MSM) Methods

To evaluate the reliability of a technology, it is common practice to subject devices to stress conditions that exceed the nominal operating range, in order to accelerate degradation mechanisms and allow for observations within a reasonable time frame. This can include exposing the device to elevated gate and/or drain biases, temperature, or even radiation. The measure-stress-measure (MSM) method is a well-known technique for this purpose, consisting of three phases: first the measurement of a fresh device, second the application of defined stress conditions, and third the subsequent characterization of the stressed device. The results of the original and post-stress measurements are then contrasted to evaluate the effect of the stress. This process may be repeated under increasingly harsh stress conditions.

MSM on Multiple Devices

For defect-spectroscopy in multiple devices of a same batch for statistical purposes the following MSM sequence is typically used. First, an $I_D(V_G)$ sweep is measured within a narrow gate bias range before the first stress cycle is applied. After that a stress and recovery sequence of a certain duration is applied. Then to the next device, the same sequence with an $I_D(V_G)$ measurement followed by a stress-recovery sequence is applied again. The same sequence is applied for all the devices of the same batch as can be seen in Figure 6.3. A typical set of selected measured traces of nMOS devices can be seen in Figure 6.4.



Figure 6.3. MSM sequence applied in devices of the same technology for statistical purposes. The scheme involves an initial $I_D(V_G)$ measurement followed by alternating stress (red) and recovery (blue) phases with predefined applied gate voltages for a specific duration. During the recovery phase, the drain current I_D is monitored and correlated to ΔV_{th} using the initial $I_D(V_G)$.

eMSM Method

Extended MSM (eMSM) measurements are commonly used to characterize the effects of BTI in large devices. In the eMSM scheme for measuring BTI effects, the device's drain and gate terminals are kept at ground during the stress phase, while a gate bias equal to or greater than the threshold voltage is applied. During the relaxation or recovery phase, the gate bias is set to a lower readout value while a small drain bias is applied to ensure a small channel current. The temporal behavior of the channel current, mapped to a threshold voltage shift during relaxation, is derived using an initial $I_D(V_G)$ measurement recorded before stress. The recovery behavior is monitored after the stress phase, and additional stress phases are performed with increasing stress times. The corresponding recovery phases are chosen long enough to eliminate most of the recoverable part of the degradation, and sampling is usually performed at logarithmic time instances. An illustration of eMSM can be seen in Figure 6.5. A set of selected results for both pMOS and nMOS devices of a SiON technolgy where eMSM has been applied is shown in Figure 6.4.

6.2.2 TDDS

The time dependent defect spectroscopy (TDDS) is another application that makes use of stress-recovery measurements, which aims to characterize single defects. Charging and discharging of such defects can be observed within the relaxation traces on small gate area devices. This can be used in conjunction with BTI measurements to characterize their charge trapping kinetics at various biases and temperatures to explain their behavior physically.



Figure 6.4. The presented recovery traces highlight the threshold voltage drift following PBTI stress in scaled SiON nMOS transistors. These devices exhibit recovery in discrete steps. Within these traces, electron emissions are identifiable as negative steps (in blue), and hole emissions as positive steps (in red), affecting the V_{th} behavior. Originally published in [KTJ4]

For TDDS, the measurement conditions are similar to those of BTI measurements on large devices. The main difference is that each stress condition is repeatedly measured to gather ample statistical data on the capture probabilities and emission times of individual defects. The characterization windows for TDDS are set by the time it takes for charge capture during stress and the charge emission duration during the relaxation phase of the defects. The measurement window in the stress phase of TDDS is determined by several factors: the range of stress times (minimum to maximum) applied, the highest gate bias that avoids causing oxide breakdown, and the lowest gate bias where the defect being studied nearly reaches full occupancy. During the recovery phase, the measurement window is defined by the initial delay after stress when the first measurement is possible, the longest relaxation time used, the highest gate bias where the defect's occupancy remains near zero, and the minimum voltage at which individual recovery steps can be measured

Applying TDDS to extract the charge transition times over wide ranges for stress and recovery bias can be time-consuming. However, TDDS measurements can be performed in conjunction with RTN measurements, which allows the capture of both close to the intersection bias while also extracting the capture and emission times of a defect over wide ranges of gate biases. TDDS data are commonly processed using step detection algorithms, similar to RTN data, but defect parameter extraction is comparatively simple as the emission probabilities for the defects increase drastically after applying the recovery conditions.

6.2.3 **RTN Measurements**

In MOS transistors, random telegraph noise (RTN) is frequently seen in conjunction with 1/f noise, and is also referred to as pink or flicker noise. The origin of this noise can



Figure 6.5. The eMSM-based measurement strategy is utilized to evaluate the change in the threshold voltage. Once the initial $I_D(V_G)$ is measured, stress and recovery voltages are applied in alternating fashion, with increasing stress and recovery durations (t_s and t_r). During the relaxation phase, the source current is recorded and subsequently converted into an equivalent ΔV_{th} using the initially measured $I_D(V_G)$ characteristics. Originally published in [KTJ5]

be attributed to individual defects that dynamically charge and discharge during device operation. As the gate area of MOSFETs shrinks, discrete steps in the channel current become noticeable, connecting 1/f noise to single defects that alter the resistance of the inversion channel.

The experimental evaluation of RTN consists of the following procedure. A constant voltage is applied to the gate of the DUT, while the drain-source current is measured over a specified duration using an equidistant sampling scheme. By initially determining the $I_D(V_G)$ curve, the current signal can be converted to a corresponding V_{th} signal. Active charge traps can capture and release charges, leading to distinct steps in the recorded current and the associated ΔV_{th} curve. For large area devices, where only the combined effect of numerous individual defects can be examined, the primary interest lies in the noise amplitude or power. In the case of small area devices, distinct steps are observable in the drain current over time at multiple gate voltages. This is shown on a few selected traces in Figure 6.7 (left).

In devices with a large gate area, a signal with 1/f noise is typically acquired, whereas in smaller area devices, distinct steps can be seen. The spectral characteristics of each individual defect's contribution can be depicted by a Lorentzian power spectral density (PSD), which exhibits a plateau beneath a specific frequency and declines proportionally to f^{-2} beyond that. When numerous defects have border frequencies loguniformly distributed across the frequency axis, this results in a 1/f PSD, which is observable in large area devices.

Defects that can be characterized by RTN are situated energetically near the Fermi level of the channel or the gate under the applied measurement conditions. These defects undergo stochastic changes in their charge state within a reasonable measurement time frame. By altering the gate bias, the region of the band diagram being scanned can be



Figure 6.6. The outcomes commonly observed when implementing the eMSM sequence on SiON devices with dimensions W × L = 10 × 10 μ m² exhibit a continuous behavior in terms of the measured drain current (I_D). This arises from the simultaneous emission of previously trapped charges by a multitude of defects across the large device area. The individual influence of a single defect on the device's performance is too minute to be discerned through measurements. Originally published in [KTJ3]

adjusted, facilitating the examination of RTN behavior under various circumstances.

To determine how the charge transition times for individual defects are influenced by bias, RTN measurements are performed at different gate voltages and temperatures. This allows us to understand how parameters like average charge capture and emission times, as well as average step heights, depend on the gate voltage V_G and temperature. By adjusting the sampling frequency, it is possible to investigate defects with both high and low capture and emission rates.

6.3 Measurement Setup

As mentioned in the previous section, measurements based on channel conductivity are vital for evaluating and characterizing defects in MOSFET devices. The two prevalent techniques for extracting channel conductance are the constant gate voltage scheme and the constant drain current method. The constant gate voltage scheme uses a transimpedance amplifier (TIA) configuration to measure the source-drain current through the channel and converts it to a voltage proportional to the current. This method is straightforward and highly stable but requires converting the recorded drain current change back to a threshold voltage shift using an initially recorded $I_D(V_G)$ characteristic. Any changes in the subthreshold-slope (SS) and trans-conductance (g_m) during the experiment may result in inaccurate ΔV_{th} measurements, although these seem to be minor [124].

In contrast, the constant drain current method implements a feedback loop from the



Figure 6.7. Four representative RTN signals exhibit varying mean charge capture and emission times, as well as step heights. From these RTN signals, the power spectral density (depicted in the same color as the corresponding RTN signal) can be derived, adhering to a Lorentzian distribution. The corner frequency of the Lorentzian corresponds to the mean capture/emission time τ of the RTN signal. The superposition of uniformly distributed Lorentzian signals results in a 1/*f* behavior, which is widely recognized in large area devices.

amplifier output to the transistor gate, ensuring constant conductance in the channel at a steady drain bias. This technique accounts for changes in trans-conductance and the $I_D(V_G)$ shape, removing extra error from post-processing with an initially recorded characteristic. Researchers should weigh factors like simplicity, stability, and the likelihood of threshold voltage shift errors when choosing a method.

Both constant-voltage and constant-current schemes can benefit from additional components. Relays can be added for switching between various amplification ranges, while other passive components can enhance SNR or ensure circuit stability. A combination of custom-built voltage sources, sampling circuits, commercial source units, or digital storage oscilloscopes can be used for control and measurement devices in these setups. Nonetheless, achieving the highest SNR and managing the timing of experiments continue to be significant challenges in the development of measurement tools in general.

6.3.1 Custom-Made Low Noise Measurement Unit for MOSFET Reliability Measurements

The custom-made low noise measurement solution used in this work, known as the defect probing instrument (DPI) [43], has been developed to tackle the specific needs of single defect spectroscopy in MOSFET devices, including TDDS and RTN measurements. The DPI is specially designed to offer comprehensive control over experimental parameters, such as output voltages and switching behavior, diverse data acquisition methods for current monitoring, and supplementary control outputs, all while maintaining minimal noise levels.

Initially optimized for single-defect spectroscopy in silicon and wide band-gap MOS transistors, the DPI has since been adapted to characterize large-area transistors using MSM schemes, CV characteristics, automated analysis of array structures, and



Figure 6.8. A constant gate voltage measurement setup schematic for methods focused on channel conductivity is displayed. The drain and gate voltages of the device are controlled by DACs. The gate voltage is directly applied to the DUT, while the drain current is connected to the positive terminal of an OPAMP operating as a TIA. The negative input terminal of the OPAMP is linked to the DUT's drain connection. The TIA's voltage output, corresponding to V_{out} , is captured by an analog-to-digital converter (ADC). Recreated from [50].



Figure 6.9. A diagram of a constant source current measurement configuration for channel conductivity-focused methods is presented. The drain bias of the DUT is directly controlled, while the gate bias is managed by the OPAMP. The OPAMP circuit adjusts the MOSFET gate bias such that the source current matches the reference current determined by the DAC bias. Recreated from [50].

devices based on 2D materials like MoS₂. Featuring a modular design concept, the DPI ensures maximum flexibility for continuously evolving requirements.

While general-purpose instruments (GPIs) are suitable for established measurement sequences, custom-made solutions like the DPI present a more versatile and adaptable option. DPI surpasses GPIs by compensating for missing features and overcoming limitations typically encountered in commercial setups. As shown in [43] the DPI's superior performance is evident when compared to GPIs supplemented with digital storage oscilloscopes, programmable pulse units, lock-in amplifiers, and other devices.

The primary configuration of the DPI comprises a three-channel pattern generator unit (PGU) to deliver programmable bias signals, a device connector unit (DCU) to convert device currents into corresponding voltages, and up to two data acquisition units (DAU) to capture voltage signals from the DCU. All units are synchronized using a backpanel bus system and can be individually configured via USB connection. This design approach enables ongoing independent enhancements of each unit, improving SNR and measurement resolution. Moreover, features such as fast ID(VG) or fast CV methods can be incorporated into the system with minimal effort.

• Pattern Generator Unit (PGU)

The PGU maintains strict electrical isolation between digital and analog components, ensuring minimal noise in analog signal paths. An ARM processor manages output signals and facilitates USB communication with the measurement host. The processor also connects to an internal bus system for exchanging control information and maintaining synchronized timing behavior. Utilizing various compensation techniques, the PGU achieves smooth transitions between distinct bias levels without voltage overshoots. The compensated switching transient allows for a maximum output frequency of 1MHz, with an output bias range configurable up to ± 48 V, expanding applicability to wide band-gap technologies like SiC and GaN.

• Data Acquisition Unit (DAU)

The DAU captures analog input voltages at a high sampling rate of up to 1MHz with a resolution of approximately $75 \,\mu$ V. Before the analog input signal is converted into a digital word, it undergoes filtering and pre-amplification. The filter and pre-amplifier stages are calibrated to match the input bandwidth of the analog-to-digital converter stage for optimal performance. The DAU also features an analog voltage level shifter, enabling an extended measurement range without sacrificing accuracy.

• Current Converter Unit (CCU)

The CCU centers around a TIA and aims to achieve the highest current measurement resolution for defect spectroscopy. It employs FET-input OPAMPs to minimize leakage currents. The CCU has seven single channels, each defined by a different gain, selected using mechanical relays. This design choice avoids the leakage currents often associated with integrated switches. Additional hardware components remove the source bias from the TIA output voltage signal, and a filter stage limits the signal bandwidth to the maximum sampling frequency of the DAU.

In conclusion, the custom-made low noise DPI measurement solution offers significant advantages for MOSFET reliability assessments, addressing the specific requirements of single defect spectroscopy and providing a flexible, modular design that can adapt to constantly evolving measurement needs.

Chapter 7 Edge Detection Algorithms

In the previous chapter, it was noted that one of the greatest challenges in conducting electrical characterization measurements is maintaining controlled bias conditions, precise timing, and minimal noise on the typically encountered nano-scale currents. An additional challenge in electrical characterization is the evaluation of the data set to derive parameters related to defects, such as charge transition times and the amplitude of step heights, which are essential for accurate simulations, lifetime estimations, and predictions. This chapter discusses two of the algorithms included in the data analysis framework used for this work: the Canny algorithm for BTI and RTN traces and the Otsu algorithm specifically for the evaluation of RTN data.

7.1 Canny Edge Detector

The Canny edge detection algorithm was originally developed by John F. Canny in 1986 to detect edges in 2-dimensional images [125]. However, this algorithm can also be used to detect steps in RTN signals, which is a 1-dimensional problem. The way the Canny edge detection algorithm works can be seen in Figure 7.1. The input of the algorithm is the RTN signal (marked as ΔV_{th}) with discrete steps, which can be seen in the upper subfigure. The detected steps (η) at a specific time can be seen in the bottom subfigure.

In general, multiple steps are required to ensure the success of the Canny process. The first step involves enhancing the intensity of the RTN signal (referred to as *S* for the coherence of the literature) by convolving the time signal with a discretized first derivative of a Gaussian filter *G*, resulting in a convoluted signal R[n], seen in the middle subfigure. This step is important for mitigation of the effects of noise and drift in the signal. The equation for the convoluted signal can be expressed as follows:

$$R[n] = (S * G)[n] = \sum_{k=-W}^{W} S[k]G_k[n-k],$$
(7.1)

where *W* is the width of the truncation and $G_k[t]$ is a function of the width of the Gaussian filter sigma.

The second step involves simplifying the convoluted signal R[n] by applying a non-maximum suppression, resulting in a simplified signal $R_m[n]$. This step is essential because it helps to identify the boundaries between adjacent regions of the signal with different properties, which is crucial to measure the signal accurately. The equation for the simplified signal can be expressed as:

$$R_{\rm m}[n] = \begin{cases} R[n], & |R[n]| > R[n+1] \land |R[n]| > R[n-1] \\ 0, & \text{else} \end{cases}$$
(7.2)

The final step involves thresholding the simplified signal $R_m[n]$ using a threshold value R_{th} , resulting in a thresholded signal $R_t[n]$. This step is important as it helps to suppress responses originating from noise. The equation for the thresholded signal can be expressed as

$$R_{t}[n] = \begin{cases} R_{m}[n], & |R_{m}[n]| > R_{th} \\ 0, & \text{else} \end{cases}$$
(7.3)

By analyzing the sign of the thresholded signal $R_t[n]$, it is possible to detect upward or downward steps in the signal, and the amplitude of each step can be extracted from the original signal. The Canny algorithm is characterized by its high precision and low error rate, which makes it an excellent tool for edge detection in RTN and BTI measurements of MOSFET devices.

However, the Canny algorithm has some drawbacks, such as the width of the Gaussian filter sigma and the threshold value R_{th} , user-defined parameters that must be adjusted to detect different types of active defects. Therefore, detection quality depends on the choice of these parameters.

In summary, the Canny edge detection algorithm is a powerful tool for edge detection in RTN and BTI measurements of MOSFET devices. It involves several steps, including enhancing the intensity of the signal, simplifying the signal, and thresholding the signal. By analyzing the sign of the thresholded signal, it is possible to detect upward or downward steps in the signal, and the amplitude of each step can be extracted from the original. However, when managing a large set of RTN signals, a fully automated step detection algorithm is necessary. One example is Otsu's method, which will be discussed in the following section.

7.2 Otsu's Algorithm

The second algorithm for edge detection used in this work is Otsu's algorithm [127]. Otsu's algorithm is a widely used method for image thresholding that has been used in various fields such as computer vision, medical image analysis, and remote sensing. As mentioned earlier, it can also be applied to one-dimensional signals, such as RTN measurements. This algorithm, named after its inventor Nobuyuki Otsu in 1979, seeks to find the optimal threshold that separates the signal into two classes: one



Figure 7.1. The Canny method for identifying edges is showcased using a representative measurement trace. The input signal, ΔV_{th} , is convolved with the first derivative of a Gaussian pulse with a specified variance, *G*. This results in a signal *R*, which features peaks corresponding to the initial signal steps. All peaks exceeding a predetermined threshold are identified and utilized to indicate the locations of the steps in the initial trace. The peak heights can be acquired from either the original signal or the peak heights. Recreated from [126].

corresponding to the background noise and the other corresponding to the signal of interest. The two classes are separated by a threshold value *t*. Values greater than or equal to *t* are assigned to the foreground, and those with intensity values less than *t* are assigned to the background. The algorithm aims to find the optimal threshold value *t* that minimizes the intra-class variance and maximizes the inter-class variance. In this case, background noise is the constant value of the measured signal, and the events of interest are the RTN spikes.

The algorithm works by calculating the between-class variance for all possible threshold values and then selecting the threshold value that maximizes this variance. The between-class variance is defined as the product of the probability of the two classes multiplied by the squared difference of their means. Mathematically, it can be expressed as

$$\sigma_b^2(t) = \omega_1(t)\omega_2(t)[\mu_1(t) - \mu_2(t)]^2, \tag{7.4}$$

where $\omega_1(t)$ and $\omega_2(t)$ are the probabilities of the two classes for a given threshold value *t*, and $\mu_1(t)$ and $\mu_2(t)$ are the means of the two classes.

To apply Otsu's algorithm to RTN signals, the first step is to calculate the histogram of the signal, which provides the frequency distribution of the signal values. The histogram is then normalized to obtain the probability distribution function, and the cumulative distribution function is calculated. Following this, the cumulative means and the total mean are computed, which are used to calculate the between-class variance for each possible threshold value.



Figure 7.2. For an RTN signal (left), a normalized histogram (right) with *L* bins is computed. For each bin, the inter-class variance σ_b^2 can be determined, and the bin with the maximum σ_b^2 provides the best threshold value. The locations where the signal transitions from above to below the threshold value can be extracted, and the corresponding step heights and time constants are obtained.

The threshold value that maximizes the between-class variance is selected as the optimal threshold value, and the signal is then thresholded at this value to obtain a binary output image. The thresholded signal can be further analyzed to extract the positions and amplitudes of the discrete steps in the signal, which can provide valuable insights into the underlying physical mechanisms.

Compared to other edge detection algorithms, Otsu's algorithm has the advantage of being fully automated and not requiring any user-defined parameters. However, it is sensitive to noise in the signal, which can affect the accuracy of the thresholding. Therefore, it is important to preprocess the signal to remove any noise before applying the algorithm.

To enhance the accuracy of automated thresholding, it is often beneficial to apply a denoising filter beforehand to suppress measurement noise [128], although this may result in the loss of very short time constants. Nonetheless, this approach ensures a highly robust thresholding process capable of detecting small step heights. In the present study, the denoising algorithm proposed by Chambolle [128] is employed.

However, Otsu's method only applies to measurement data containing a single active defect a single active defect. When multiple active defects are present, such as in cases of multi-level RTN, it is customary to extract only the most prominent defect signal to avoid erroneous detection. Furthermore, additional sanity checks, including the Kolmogorov-Smirnov test, are employed to validate the detection results by assessing whether capture and emission times follow an exponential distribution. For a more comprehensive analysis of multi-level RTN, advanced techniques like factorial hidden Markov model analysis are necessary [129].

Chapter 8

Measurements and Results

This section will present the results obtained by employing the electrical measurements, data analysis techniques, and Comphy reliability simulations discussed in the previous chapters. Section 8.1 will delve into the statistical analysis of individual defect measurements, adopting the defect-centric approach and focusing specifically on Si/SiO₂ MOSFETs under BTI conditions. *The measurement results of this technology have been published in [KTJ7, KTC1, KTJ5]*. In Section 8.2, BTI's characterization will be showcased through single defect measurements on Si/SiON MOSFETs. These measurements and the extracted parameters will be used to calibrate the Comphy framework to simulate MSM sequences. *The measurement results and Comphy simulations based on this technology have been published in [KTC2, KTJ4]*. Finally, Section 8.3 will discuss the results of RTN measurements conducted on high-k MOSFETs integrated within smart arrays.

8.1 Impact of Defects on SiO₂ Transistors

The impact of single defects in nanoscale devices is considered a significant challenge in modern technologies [47]. When compared to larger devices with dimensions in the micrometer range, nanoscale devices exhibit a considerable level of variability in their characteristics [48]. Although the number of defects affecting the performance has decreased over the past few decades, the impact of a single defect on the drain-source current and the associated threshold voltage shift ΔV_{th} has increased [24, 46]. Because of the reduced capacitance in nanoscaled devices, on average, each defect has a more significant effect on the surface potential [106]. In devices with small gate areas, the trapping of charges at oxide defects can lead to a significant reduction in the source-drain current, even for single charged defects. The current is reduced due to the defect's effect on the percolation path, as depicted in Figure 8.1. Due to the increased impact of charge trapping, emission events appear as discrete steps in the MSM measurements [130, 40, 131, 106], and statistical analyses based on their characteristics can be conducted.

The average threshold shift induced by a single defect (η) and the average number of active defects per device (N_T) are the main parameters of interest when studying the variability of devices. The dependence of these parameters on the device geometry is a decisive factor for optimizing device dimensions against the detrimental effects of charge trapping. Two models have been proposed for planar FETs: $\eta \propto 1/(WL)$ [112, 114] and $\eta \propto 1/(W\sqrt{L})$ [26]. For the technology studied here, results show that the defects' impact follows $\eta \propto 1/(W\sqrt{L})$ when measured close to V_{th} and $\eta \propto 1/(WL)$ when measured close to v_{th} and $\eta \propto 1/(WL)$ when measured close operating conditions. Additionally, the DCM [112, 132] has been used to evaluate large-area devices of the same batch, and the results obtained from single-defect extraction agree with the DCM investigations. This provides a fully consistent picture and methodology to evaluate the effect of charge trapping in both large-area and scaled technology nodes.



Figure 8.1. Schematic of the carrier densities in the channel of a scaled device and the resulting current percolation path. (a) The randomly distributed dopants lead to locally decreased carrier densities. The neutral defect does not influence the current path. (b) When the defect is charged, the restrictions on the percolation path increase, leading to a decreased drain current and a higher threshold voltage [50]. Originally published in [KTJ7]

8.1.1 Devices and Measurements

Electrical measurements under NBTI and PBTI stresses were conducted on pMOS and nMOS devices with a wide range of gate areas, fabricated in a commercial technology using SiO_2 as gate insulator. A MSM scheme, see section 6.2.1, was employed to study the temporal recovery of device degradation, which was monitored using DPI [43].

The MSM measurements for each device under test consisted of three phases: an initial $I_D(V_G)$ measurement within a narrow gate bias window, a stress phase with a gate field of 7 MV/cm and a temperature of $T = 100 \,^{\circ}$ C for a stress time of 100 s, and a recovery phase with a recovery field corresponding to V_{th} and time of 100 s. The drain current was recorded during the recovery phase using a constant current criterion. The resulting recovery curves were mapped to equivalent ΔV_{th} traces using initially measured $I_D(V_G)$ characteristics, assuming negligible changes in SS. The V_{th} values from the constant current criterion were confirmed using the method of the second derivative of the drain current.

The stress conditions were optimized to capture a sufficient number of defects for statistical significance without needing excessively long measurement sequences or



Figure 8.2. Recovery traces obtained from scaled pMOS and nMOS devices showing the drift of the threshold voltage after NBTI/PBTI stress over time, measured on a scaled SiO₂ technology. As can be seen, the recovery proceeds in positive discrete steps for pMOS and negative for nMOS. Increased variability is observed between nominally identical devices. Originally published in [KTJ7].



Figure 8.3. Recovery traces obtained from large pMOS and nMOS devices showing the drift of the threshold voltage after NBTI/PBTI stress. The traces recover similarly in a continuous manner as the increased number of defects with reduced impact on ΔV_{th} lead to a smaller variability. Originally published in [KTJ7].

special efforts to disentangle the signals from simultaneously occurring charge emission events at different defects. These parameters were tested on the smallest geometry, where three different batches were available to test devices with the same parameters without previous stress affecting the results. The operating conditions of the test devices were around 3 MV/cm, making the stress bias strong enough to charge a large ensemble of defects while avoiding extensive damage to the devices.

Figure 8.2 shows ΔV_{th} traces as a function of recovery time after gate stress conditions for pMOS and nMOS devices with SiO₂ as the gate insulator. These traces were obtained by converting measured drain currents to an equivalent ΔV_{th} . Typically, pMOS devices with SiO₂ insulators are more affected by BTI than nMOS devices, as evidenced by the difference in the total ΔV_{th} in Figures 8.2 and 8.3. This difference can be explained by the relative position of trap bands in the insulator with respect to the silicon band edges [133].



Figure 8.4. The CCDFs of the extracted discrete steps are shown. The pMOS data seem to follow a unimodal exponential function, while nMOS data can be described by a bimodal exponential function. The maximum limit of the CSA is shown as vertical dashed lines. Originally published in [KTJ7].

8.1.2 Measurement Analysis

The two primary methods used for the measurement analysis are the step height extraction of single defects and the DCM.

Single-defect Analysis

The Canny algorithm, see Section 7.1, embedded in the used data analysis framework [113, 116], is used to extract the discrete transitions of ΔV_{th} shown in Figure 8.2. To avoid measuring artifacts due to the noise of the measuring system [43], the range of extracted step heights is considered inside the noise limits of each measured time interval.

To describe the contribution of electron and hole traps on ΔV_{th} , a CCDF is created based on the collected amplitude of step heights (Δv_{th}) for each device geometry normalized to the active number of defects per device, see Section 5.2. Previous studies have observed that pMOS devices can be described by a unimodal exponential distribution [24], while both unimodal and bimodal exponential behavior has been observed for nMOS devices [45, KTC2]. The CCDFs obtained for the smallest geometry of the pMOS and nMOS set can be seen in Figure 8.4, where a unimodal exponential CCDF is observed for the pMOS data and a bimodal one for the nMOS data.

Defect Centric Model (DCM)

The number of defects per device is typically modeled by a Poisson distribution [112]. The DCM [112], see Section 5.3.1, combines the Poisson distribution of the number of defects per device with the exponentially distributed step heights to calculate the distributions of the total ΔV_{th} . The first two moments of ΔV_{th} can be used to extract the two statistical parameters of interest: the average $\langle \Delta V_{\text{th}}(t) \rangle$ at a given time and its variance $\sigma^2(t)$. $\sigma^{2}(t) = 2\langle \Delta V_{\rm th}(t) \rangle \eta,$ $\langle \Delta V_{\rm th}(t) \rangle = N_{\rm T}(t)\eta,$

In this work, DCM is employed to evaluate the data from both nanoscale and large-area devices. The DCM is particularly useful for evaluating device geometries where single charge transitions are obscured by measurement noise, such as largearea transistors. Therefore, the combination of both single-defect analysis and DCM provides a versatile framework for evaluating time-dependent variability in transistor technologies.

8.1.3 Negative BTI in pMOS Devices

In the following, the examination focuses on how the impact of charge traps on the threshold voltage (ΔV_{th}) is influenced by the recovery bias, as opposed to previous studies, which focused primarily on charge trapping near the sub-threshold regime. It is important to note that when utilizing this information in a circuit, understanding BTI's effect over a wide bias range is required for efficient and robust design.

Operation Regime Close to V_{th}

Specifically, the operation regime close to the threshold voltage (V_{th}) is studied, by examining the collected amplitudes of charge transitions following NBTI stress on pMOS devices with scaled geometries. These amplitudes are represented in the form of CCDFs in Figure 8.5, where the symbols represent experimentally extracted step heights and the dashed lines represent unimodal exponential functions used to describe the data. It was found that the unimodal exponential distributions provided a smaller error for the tested data set when compared to other distributions in the literature, such as lognormal or bimodal exponential distributions [134, 135, 136, 137], despite having fewer available parameters. However, it should be noted that the range of each measurement unit may influence the optimal distribution for explaining experimentally created distributions. In this particular data, a limit of 0.3 mV is used for the amplitudes, to avoid kinks at small step heights or missing steps in that range due to measurement noise.

The mean of the unimodal exponential distributions, i.e. the average threshold shift induced by a single emission event (η), determines the slope of the function. Devices of the same area are expected to have the same mean in terms of the CSA. However, in the studied geometry, it is observed that devices with the same area but different W/L lead to different η . The same slope is observed in two device sets with the same $W \times \sqrt{L}$.

To further investigate this observation, η was plotted against $W \times L$ and $W \times \sqrt{L}$ in Figure 8.6. The obtained root mean square error (RMSE) of a linear regression is around three times smaller in the $W \times \sqrt{L}$ case. The RMSE for the first case is RMSE = 0.15 mV while in the case of a $W \times \sqrt{L}$ dependence it is only RMSE = 0.046 mV. This asymmetry



Figure 8.5. The CCDFs of step heights of hole traps extracted after NBTI stress on pMOS devices. The dashed lines depict the unimodal exponential distribution used to describe the results. A recovery bias close to V_{th} was applied during the recovery phase. Originally published in [KTJ7].

can be explained by the different impact of variations of W and L on percolative conduction [26, 108, 138]. When devices are measured close to the threshold voltage, multiple non-uniform conduction paths form due to the impact of random dopants in the substrate [110, 111]. The local potential associated with each discrete dopant acts as a barrier for the current, which consequently flows in a few narrow channels connecting the source to the drain. In this perspective, a weaker dependence on the device length can be observed, as in a narrower channel, there is a higher probability for a trap to effectively suppress a percolative conduction path [110]. Generally, any increase of W and L increases the number of charges in the inversion layer. This can lead to reduced values of η [106]. However, when considering percolative conduction, W and L will have a different impact. Increasing the area while keeping L constant, by only increasing W, will lead to a higher number of current channels between source and drain compared to a case where W is kept constant but L is increased [111]. Due to that, a single defect could have a higher impact in the latter case of a narrower channel.

The extrapolated line of the $W \times \sqrt{L}$ is extended towards larger device areas in Figure 8.7, in order to test the validity of the results for large-area devices, where the measurements are evaluated using DCM. It can be seen that the DCM values, which are depicted with the half-full symbols, are nicely captured by the $W \times \sqrt{L}$ trend. In some geometries, the markers for the experimental data and data from the DCM model overlap as their values are too close for a difference to be observed on the logarithmic scale.

Next, the number of active traps N_T are plotted against both geometrical trends in Figure 8.8. It should be noted that a comparison of the N_T values is meaningful here, as data sets for the same NBTI bias conditions are compared. Again, for the comparison of N_T , the $W \times \sqrt{L}$ dimensional scaling captures all the data more accurately. Regarding the number of defects, this trend can be explained by the generally higher number of low-impact active defects which can be detected due to the increased percolative conduction in wider devices [110, 111]. Thus, scaling the width can increase the number



Figure 8.6. (Top) The η values for the unimodal fit used in Figure 8.5 are plotted against the area. The devices with the same *W* scale nicely, but the devices with different *W* deviate from this trend, showing higher step heights. (Bottom) The η values for the unimodal fit are plotted against $W \times \sqrt{L}$. The fitted line seems to accurately capture all the data obtained from different geometries. Originally published in [KTJ7].



Figure 8.7. The η values are plotted against $W \times \sqrt{L}$ trend. The fitted line nicely captures both experimental and DCM values. For some geometries, the markers used for the experimental data and data from the DCM model overlap as their values are too close to see a difference on a logarithmic scale. Originally published in [KTJ7].



Figure 8.8. The average number of defects per device is plotted against both dimensional trends. The marker and color of the symbols follow the same convention as previous figures. Originally published in [KTJ7].



Figure 8.9. The $I_D(V_G)$ characteristics of the smallest devices of the set. The grey dashed lines indicate the measurement conditions where the recovery traces were recorded. Originally published in [KTJ7].

of defects more substantially but also decrease the impact of every single defect.

On-state Operation Regime

In order to study the impact of the percolative conduction on the average step height, the device set was measured at higher biases, corresponding to operating conditions, where a more uniform way of conduction is expected due to the increased carrier density in the inversion layer. The two different measurement conditions used in measurement cases can be seen in Figure 8.9, together with the set of initial $I_D(V_G)$ curves for the smallest device geometry. The measurements were conducted as close to the limit as possible, where the discrete nature of recovery persists. As can be seen in Figure 8.10, the plotted η seem to follow the area trend, as the η values of narrow devices are close to the values of the wider devices of the same area. The error for the area trend is around three times lower compared to what has been observed for the measurements around V_{th} . The calculated values from the DCM are also captured nicely by the same trend.

The η values are expected to decrease when measured at higher recovery biases



Figure 8.10. The η values for the on-state measurements are plotted against $W \times \sqrt{L}$. The fitted line accurately captures all the data obtained from different geometries and the DCM values for the two large geometries. Originally published in [KTJ7].



Figure 8.11. Traces of the smallest device of the set measured close to Vth (red lines) and at high operating conditions. At high operating conditions, the recovery is no longer observed in a discrete manner due to the increased charge in the inversion layer. Originally published in [KTJ7].

due to the increased screening of trapped charges from the inversion layer [106]. This can be observed clearly in Figure 8.11, where a high recovery bias, five times larger than V_{th} , is used. In this operation regime, the traces are recovering in a continuous form and discrete steps can not be distinguished anymore, as η values become smaller than the measurement noise, similar to what has been observed in large-area devices. In Figure 8.11, the convention of $\Delta I_D/I_D$ is used, to avoid comparisons of ΔV_{th} for such large differences in recovery biases, as the drain current becomes too large to allow for accurate extraction of ΔV_{th} .

The average number of active defects for measurements close to V_{th} and close to the on-state can be seen in Figure 8.12 marked with red and blue color, respectively. Interestingly, more active defects are observed in the measurement window for the higher recovery bias. The only deviation from this overall trend is observed for the largest geometry shown in Figure 8.12. A possible explanation of this deviation is that some events can be missed in the extraction process as the amplitude decreases when the area and the recovery bias increase.

For the case of a higher recovery bias, a smaller portion of the activated defects is expected to recover due to the smaller active energy region (AER). Furthermore, the measurement window's range and the readout bias's impact on the charge transition times can affect the results. Emission times of defects are expected to have a slight dependence on the recovery bias [40, 41]. If emission times increase with higher recovery biases, events with emission times smaller than the selected measurement range for $V_{\rm th}$ can become accessible for the recovery conditions in the on-state. Additionally, the bias dependence of each trap differs, thus, it is plausible that more events with fast emission times fit in the measurement window in comparison to the slow emission times, which are no longer accessible.



Figure 8.12. The average number of active defects for the two measurement cases plotted for all the geometries. Higher values are observed when devices are measured at the on-state. Originally published in [KTJ7].

8.1.4 Positive BTI in nMOS devices

The measurements of nMOS devices have been performed on four scaled geometries where the step extraction was possible and two large ones that have been analyzed with the DCM. The CCDFs for the scaled nMOS device can be seen in Figure 8.13. For all nMOS device geometries studied, the recovery condition was chosen close to V_{th} . In addition to the exponential unimodal distributions used in the pMOS case, bimodal distributions were used to describe the results since they appear to show more minor errors when used for nMOS devices. The solid lines depict the bimodal exponential functions.

It is worth noting that the bimodal exponential distributions have been mainly observed in the past for devices with high- κ gate stacks, where each branch of the distribution has been assigned to charge transfer interactions of defects within each of the two layers, e.g. SiO₂ and HfO₂, with the channel [45]. Recently, the bimodal exponential distributions have been reported for devices with SiON insulators [KTJ4].

In order to develop an understanding of the geometrical dependencies of the two exponential branches, the average step heights were collected and plotted against $W \times L$



Figure 8.13. The CCDFs of step heights of electron traps extracted after PBTI stress on nMOS devices. The bimodal exponential distributions (solid lines) describe the results more accurately than the unimodal exponential distributions (dashed lines). Originally published in [KTC1].

and $W \times \sqrt{L}$, see Figure 8.14. To compare the errors of the two distributions, the rootmean-square error (RMSE) was computed and normalized (nRMSE) to the range of the values for the two cases.

The $W \times \sqrt{L}$ behavior describes the data more accurately for the unimodal case than pure area scaling. For the $W \times L$ dependence, nRMSE_{Unimodal} = 0.16 while for the case of a $W \times \sqrt{L}$ dependence is nRMSE_{Unimodal} = 0.02. In the case of the nMOS devices, a clear overall trend of the η values describing the bimodal exponential distribution cannot be observed. While the slope of the second branch scales for both cases, the first branch does not follow the trend and instead seems to be uncorrelated to the device dimensions.

The $W \times \sqrt{L}$ trend is presented in Figure 8.15 together with DCM data. The η values of step detection are taken from the second branch of bimodal distribution as they appear to be closer to DCM values. The DCM values of large-area devices deviate from the trend observed for the experimentally extracted step heights from single defect measurements. The reasons for these deviations observed for PBTI on nMOS devices are currently unclear and need to be analyzed together with the origin of the bimodal distribution in more detailed future investigations.



Figure 8.14. (Top) The average step heights induced by single defects (η) for the data used in Figure 8.13 are plotted against the area. The red line shows the scaling of the values following a unimodal exponential distribution, and the gray dashed line is used as a comparison to the CSA. The RMSE values were normalized to compare the two different η ranges. (Bottom) The average step heights induced by a single defect (η) for the two distributions used in Figure 8.13 are plotted against $W \times \sqrt{L}$. A smaller error is observed in the latter case for the unimodal distribution. Originally published in [KTJ7].



Figure 8.15. The average step heights induced by single defects (η) is used together with calculated values from DCM for the large devices that are plotted against $W \times \sqrt{L}$. The step detection values are taken from the second branch of bimodal distribution as they are closer to DCM values than the values extracted from unimodal distribution. Originally published in [KTJ7].

8.1.5 Body Bias Dependence

In previous sections, it was examined how lateral dimensions and recovery gate conditions impact the statistical quantities related to defects such as η and $N_{\rm T}$. Next, research focus is extended on how body bias (BB) can also play a pivotal role in shaping the effect of individual charged defects on device behavior.

Time-zero Characteristics



Figure 8.16. Probability distribution of the initial V_{th0} values across a sample set of nanoscaled devices under varying V_B conditions, conforming to normal distributions. This change is largely attributed to modifications occurring in the depletion layer thickness X_{dep} as a response to alterations in BB settings.

In the analysis of the $I_D(V_G)$ measurements, the extracted V_{th0} values manifested a normal distribution, see Figure 8.16. The BB impact is known to modulate the channel depletion layer thickness (X_{dep}) and, as a result, the average threshold voltage, but it seems to also influence the variability of V_{th0} [139]. With the introduction of a forward body bias (FBB), the spread of V_{th0} in the standard deviation affiliated with the distribution was reduced, as depicted in Figure 8.17 for pMOS and Figure 8.18 for nMOS devices



Figure 8.17. Influence of the body bias on V_{th0} variability for pMOS devices. Under FBB, a reduced standard deviation of the normal distribution is observed. The smaller geometry exhibits higher initial variance and a more pronounced slope.



Figure 8.18. Influence of the body bias on V_{th0} variability for nMOS devices. The standard deviation is a few times higher in nMOS devices compared to their pMOS counterparts.



Figure 8.19. Exponential distributions of individual charge ΔV_{th} step heights for a consistent device sample set under varied V_{B} conditions. RBB shows a heightened variance (elevated η), whereas the FBB exhibits a diminished variance (lower η).

accordingly. This effect was more pronounced in pMOS devices with dimensions of $135\,\mathrm{nm}\times350\,\mathrm{nm}.$

Reducing X_{dep} through adjustments in BB has similar positive effects as reducing the channel doping level, notably decreasing the initial variability. This observation suggests that fine-tuning of BB could emerge as a valuable strategy to control variability in device functionalities, similar to how adjustments in doping levels are used [139].

Negative BTI for pMOS devices

The CCDFs for pMOS geometries with increasing width are displayed in Figure 8.19 together with their unimodal fits depicted with dashed lines. The extracted η values are plotted against V_B values in Figure 8.20, where a linear fit approximation is used for this bias range. A rise in V_B results in greater η values. Furthermore, the smaller geometry exhibits higher η values at $V_B = 0$ and a more pronounced slope, revealing the clear influence of substrate dopants on η values together with area dependence. TDDS relaxation traces captured for two different BB conditions on a device with a dominant oxide trap can be seen in Figure 8.21. Notably, the BB seemed to impact the step heights



Figure 8.20. The average step heights induced by a single defects (η) against the V_B for pMOS devices. Increased V_B leads to higher η values. The smaller geometry exhibits higher η values at $V_B = 0$ and a more pronounced slope.



Figure 8.21. TDDS on two different BB settings on a single device. The studied step height is heavily influenced. The larger RBB leads to an increase in the step height, an observation consistent with what has been observed in the CCDFs of multiple devices.

in a similar way as in Figure 8.20, where the increase of $V_{\rm B}$ leads to higher η values. This phenomenon proved consistent, repeating in numerous devices showcasing substantial gate oxide defects.

The trends observed in the experiments have a strong connection to the adjustments in X_{dep} . RBB can expand X_{dep} , allowing the dopants in the substrate to have a more pronounced effect on the current's flow paths. Essentially, utilizing BB to manage X_{dep} emerges as a promising strategy to reduce the complications arising from RDD, enhancing bot time-zero and time-dependent variability.

Positive BTI for nMOS devices

In Figure 8.22, the CCDFs for the nMOS set of measurements are displayed. Applying a bimodal exponential function typically provides a more accurate representation of the experimental data, characterized by a reduced statistical error. Despite this, and considering the inconsistent scaling highlighted in [KTJ7], a single exponential distribution is emplyed to derive the η values. The outcomes of this process are illustrated in



Figure 8.22. The CCDFs of step heights of hole traps extracted after NBTI stress on pMOS devices. The dashed lines depict the unimodal exponential function used to explain the results.



Figure 8.23. The average step heights induced by a single defect (η) for the data used in Figure 8 are plotted against the *V*_B. Increased -*V*_B leads to higher η values. The slope is less steep for the nMOS case.

Figure 8.23, where it can be observed that the extracted η values follow a similar scaling trend as those of pMOS, albeit with a slightly diminished slope.

Shifting focus to Figure 8.24, the TDDS results are presented, delineating the outcomes under the extreme settings of both FBB and RBB. Analyzing a selected defect, it was anticipated and confirmed that higher step heights are attained with RBB. Interestingly, the discrepancy between the two clusters was less pronounced than initially hypothesized, especially when compared to their pMOS counterparts. This subtler distinction may be attributed to the relatively lower concentration of dopants present in the nMOS devices, which influences the observed behaviors.

8.1.6 Conclusions

The statistical distributions of the step heights of threshold voltage drift describe the contribution of single defects. These step heights are observed by applying the MSM scheme under PBTI conditions on nMOS devices and NBTI on pMOS devices with dimensions small enough to observe discrete steps when charge transitions occur. Based



Figure 8.24. TDDS for two different BB conditions on a selected device. The step heights at larger RBB are increased, an observation consistent with what has been observed in the CCDFs. BB impact is smaller in nMOS devices than in pMOS devices.

on the detected steps, the CCDFs of the step heights are created and η values by utilizing exponential functions are extracted. The findings demonstrate that a bimodal exponential distribution can explain the results extracted from nMOS data and a unimodal exponential distribution the results from pMOS data. For the measurements close to V_{th} , it is shown that the scaling of statistical parameters is described better by a $W \times \sqrt{L}$ trend rather than the area of the devices $W \times L$ which is typically used in literature. However, if a higher recovery bias is applied η decreases, especially on devices affected more by the percolative conduction, and the $W \times L$ trends capture nicely the values for all the tested geometries. At the same time, more defects contribute to the observed degradation for higher recovery bias conditions. Nevertheless, the dimensional scaling trends for the step heights for a bimodal distribution for nMOS data can still not be fully resolved and need to be addressed in future studies.

Finally, in the results of the influence of individual charged gate oxide defects on nanoscaled pMOSFETs, the role of body bias emerged as a critical factor. Distinct changes were identified in both time-zero and time-dependent variabilities. The underpinnings of these observations lie in the alteration of unscreened dopant atoms in the channel depletion zone, which are modulated by the body bias. Using an RBB raises both X_{dep} and V_{th0} , making it useful for applications that require low leakage. On the other hand, an FBB decreases X_{dep} and V_{th0} , which can be beneficial for high-performance applications, albeit with a trade-off of higher junction leakage. These findings are of high relevance to the reliability of nanoscaled MOSFETs as they constitute a flexible electrical technique to improve the time-zero and the time-dependent variability in nanoscaled MOSFETs without involving any technological tuning.

8.2 Single-defect analysis of BTI employing SiON Transistors

A single defect analysis has also been performed in a commercial SiON technology in addition to SiO₂, which was examined in the previous section. Furthermore, it is observed that the electron and the hole trap contributions for each device kind, indicating that trapping at both trap types is of equal importance for precise descriptions of the devices. Previous research on SiON technologies has suggested that the step heights of single defects are best described by unimodal exponential distributions. However, the investigation has uncovered evidence that suggests bimodal exponential distributions are a more accurate representation. The importance of this discovery lies in accurately assessing the tail of the distribution, which encompasses defects that can have a significant impact on the ΔV_{th} and cause device and circuit failure. To better understand the effect of these defects, their statistical distributions are analyzed and compared to values obtained using the commonly used CSA. The results indicate that the CSA significantly underestimates the true impact of the defects for the studied SiON technology. Finally, the obtained distributions are used to evaluate their impact on the variability of measure-stress-measure simulations, utilizing Comphy.

8.2.1 Devices and Measurements

For this technology, both nMOS and pMOS devices with different geometries have been investigated under both PBTI and NBTI conditions. The measurement process is the same MSM sequence as the one mentioned in the previous section. The only difference is that in this study the device set was measured with stress and recovery times of 1 ks.

8.2.2 Positive BTI in nMOS Transistors

Initially, the temporal variability and reliability of nMOS devices are examined by applying a positive stress bias. When a positive bias is applied on the gate of a MOS structure, the valence and conduction band edges bend towards lower energies. Figure 8.25 (left) presents a schematic representation of the band diagram during PBTI stress.

For a defect to impact ΔV_{th} , it must be located in the AER where charge trapping occurs. In the case of PBTI, electron traps above the Fermi level of the channel carrier reservoir and hole traps below the Fermi level at the poly-gate can acquire and release charges under specific conditions. These conditions require that the applied stress lasts longer than the time it takes to capture the charge and that the recovery time exceeds the emission time of the respective defect. Consequently, the energetic range covered by the AER, governing the charge exchange between the channel and electron traps (represented by blue triangle) and between the poly-gate and hole traps (depicted as red triangles), is determined by the bias utilized for the MSM measurement.



Figure 8.25. The figure presents the band diagram of nMOS devices under PBTI conditions. It illustrates the presence of electron traps (depicted in blue) and hole traps (depicted in red), along with their respective AERs for charge trapping. The blue region represents the AER for charge transitions between the defects and the channel. In contrast, the red region represents the AER for charge transitions between the defects can change their charge state during the energetic boundaries within which the defects can change their charge state during the experiments, thereby contributing to the drift of the measurement signal. It should be noted that the height of the AERs varies with the applied gate bias. On the right side, the distribution of step heights, measured from SiON nMOS transistors after undergoing PBTI stress, is shown for three sets of devices with different geometries. The majority of observed traps are electron traps (depicted on the left), although a certain number of hole traps (depicted on the right) can also be observed. The electron traps show a bimodal exponential behavior. Originally published in [KTC2].

Figure 8.25 (right) displays the corresponding CCDFs for the PBTI scenario for three sets of devices with different geometries. The largest device (denoted as (T) in the figure legend) possesses a thicker oxide compared to the other two devices. Consequently, a higher gate bias is employed to maintain a consistent oxide field. Electron trapping, involving the interaction between the silicon channel and defects, is represented by blue shades, while red shades represent hole traps, which denote interactions between the poly-gate and defects. The CCDFs following PBTI stress exhibit a bimodal exponential behavior for electron traps and a unimodal exponential characteristic for hole traps. To fit the experimental data and extract device parameters η and $N_{\rm T}$, the normalized CCDF (Equation 5.3) is employed. The unimodal and bimodal fits are represented by dashed and solid lines, respectively. The obtained parameters for the PBTI case are listed in Table II.

Based on the gathered parameters and the CCDF plots, it can be seen that the impact on ΔV_{th} during PBTI stress conditions is primarily governed by electron traps. However, a small number of active hole traps have also been identified. Out of a total of 990 recorded traps, 910 defects (92%) have been characterized as electron-related, whereas hole trapping originates from interactions between defects and the gate, as indicated by recent research [46]. An advantageous aspect of investigating single defects is the ability to separate the contributions of electron traps and hole traps. In contrast, when examining large-area devices, only the average response of numerous defects can be **Table 8.1.** Extracted parameters of CCDF fitting on experimental data for the PBTI study case. Superscripts e and h refer to electron and hole trapping accordingly. The bimodal fit is only used for electron traps since hole traps do not exhibit two branches.

W(nm)	L(nm)	$\eta^{\rm e}(mV)$	N ^e	$\eta^{\rm h}(mV)$	N^{h}						
400	180	1.8	4	0.35	2.4						
440	360	0.83	0.83 7 0.37		0.5						
1000	700	0.56	10.4	0.6	0.31						
bimodal											
W(nm)	\mathbf{I} (mass)	$(\ldots, \mathbf{T}\mathbf{Z})$	3.7	(
••(1111)	L(IIIII)	$\eta_1(mV)$	N_1	$\eta_2(mV)$	N_2						
400	180	$\eta_1(mv)$ 0.52	N ₁ 6.64	$\frac{\eta_2(mV)}{2.86}$	$\frac{N_2}{1.4}$						
400 440	180 360	$\eta_1(mv)$ 0.52 0.42	N ₁ 6.64 11.79	$\eta_2(mV)$ 2.86 1.86	$\frac{N_2}{1.4}$ 0.89						

Table 8.2. Extracted parameters of CCDF fitting on experimental data for the nMOS/NBTI study case. Superscripts e,h refer to electron and hole trapping, respectively.

W(nm)	L(nm)	$\eta^{e}_{1}(mV)$	N^{e}_{1}	$\eta^{e_2}(mV)$	N^{e}_{2}	$\eta^{\rm h}{}_1(mV)$	N^{h}_{1}	$\eta^{\rm h}{}_2(mV)$	N^{h}_{1}
400	180	0.33	6.73	2.49	0.21	0.25	4.22	1.04	2.26
440	360	0.37	6.9	1.39	1.45	0.37	4.15	2.22	0.3
1000	700	0.16	18.54	0.62	1.08	0.18	26.33	0.72	1.02

measured.

The absolute contribution of a particular carrier type, such as electrons, to the overall threshold voltage shift ΔV_{th} , can be calculated using the equation [46]:

$$r_{e} = \frac{\sum_{i=1}^{N_{e}} |d_{e}|}{\sum_{i=1}^{N_{e}} |d_{e}| + \sum_{i=1}^{N_{h}} |d_{h}|},$$
(8.1)

Here, $N_{\rm e}$ and $N_{\rm h}$ represent the numbers of electron and hole defects, respectively, while $|d_{\rm |}$ and $|d_{\rm h}|$ correspond to the step heights of individual hole or electron emission events. It can be observed that hole trapping leads to a reduction in the total $\Delta V_{\rm th}$ by approximately 5.9 % under the specific bias and temperature conditions employed for the PBTI case.

8.2.3 Negative BTI in nMOS Transistors

In addition to the previous section, the investigated device types were subjected to negative bias stress conditions. The application of a negative gate voltage leads to a reverse band bending compared to the PBTI case, hence the conduction and valence band edges are bent towards higher energies. The corresponding band diagram for the NBTI case is displayed on the left side of Figure 8.26. The AERs for hole trapping are
indicated in red, while those for electron trapping are shown in blue. During NBTI stress, hole traps below the Fermi level can capture charges from the channel, while carriers from the poly-gate can charge the electron traps. Consequently, a more significant number of hole traps and fewer electron traps are anticipated to contribute to the threshold voltage shift compared to the PBTI case.



Figure 8.26. (Left) The band diagram of nMOS devices for the NBTI conditions. In contrast to the PBTI case, more hole traps can contribute to total ΔV_{th} now as the channel's AER covers the band-gap's lower half. The AERs for charge transitions between the defects and the channel and for charge transitions between the defects and the gate are marked in blue and red color, respectively. (Right) Distribution of step heights of defects extracted after NBTI stress. Bimodal exponential behavior can be observed for both types of traps. Originally published in [KTC2].

The increased prevalence of hole trapping is clearly evident in the extracted CCDFs for both electron and hole traps, presented on the right side of Figure 8.26. Each type of trap exhibits a two-branch distribution, and considering an unimodal model would underestimate the tail of these distributions. A notable distinction from the PBTI case is the presence of bimodal exponential distributions for both types of traps in this scenario. The parameters obtained from the bimodal fitting are compiled in Table 8.2. Unlike studies utilizing unimodal models, the extracted parameters from the bimodal model do not appear to be associated with the device area, as observed in previous works [24].

8.2.4 Negative BTI of pMOS Transistors

The previous section demonstrated that both electron and hole traps contribute to the ΔV_{th} shift, and that the step height distribution requires a bimodal description. To complement these results, pMOS devices under NBTI conditions are analyzed using the same device geometries as in the nMOS case. Similar to the nMOS device, the largest devices have a thicker oxide compared to the other geometries, necessitating a higher stress gate bias to maintain an equivalent oxide field.

The CCDFs of the pMOS device are presented in Figure 8.27. In contrast to the nMOS device, only hole traps are observed in the pMOS device, with electron traps having no contribution to the threshold voltage shift. This observation can be attributed to the higher effective number of defects typically found in pMOS devices, as previously

Table 8.3. Extracted parameters of CCDFs for experimental data of pMOS devices under NBTI stress conditions.

unimodal			
W(nm)	L(nm)	$\eta^{\rm h}(mV)$	N^{h}
220	180	1.32	54.63
400	180	0.72	56.39
800	180	0.44	120.66
440	360	0.37	121.8
1000	500	0.4	42.77

reported [140]. The step heights exhibit a unimodal distribution across all geometries, and the extracted parameters can be found in Table 8.3.



Figure 8.27. Distribution of step heights of hole traps extracted after NBTI stress on pMOS devices. The dashed lines depict the unimodal exponential function used to explain the results. originally published in [KTJ4].

8.2.5 Comparing NBTI and PBTI degradation

The step heights of electron traps extracted from nMOS devices exhibit a bimodal exponential distribution, which is distinct from what has been reported in the literature for SiON technologies, where only unimodal exponential distributions have been observed [140, 45]. bimodal distributions of single defects have been observed in technologies utilizing high- κ gate stacks [45]. In high- κ devices, each branch of the distribution is associated with charge-transfer interactions between defects residing in one of the insulating layers and the channel.

Regarding PBTI, the behavior of nMOS devices is predominantly influenced by electron traps (approximately 92% of all traps are electron traps), while for NBTI, hole trapping becomes significantly more important. The ratio between electron and hole traps is approximately 52% to 48%. Therefore, it is necessary to consider both electron and hole traps in the analysis. On pMOS devices, the ΔV_{th} originates solely from hole traps. It is worth noting that a large number of traps with small average step height are more likely to be located farther from the channel [46], and thus their measurement



Figure 8.28. Extracted impact of a single defect on the $\Delta V_{\text{th}}(\eta)$ and the number of traps per device (N_T) considering unimodal exponential distribution. Clearly visible is the underestimation of the CSA of the impact of the defects, on the device behavior. The maximum impact of CSA is considered by considering the traps directly at the interface. Originally published in [KTC2].

requires optimized low-noise tools [43]. Otherwise, charge trapping might be entirely masked by the measurement noise.

Comparison to CSA

The extracted parameters for PBTI in nMOS devices, including the average step height and the number of traps per device, considering both unimodal and bimodal CCDFs, are presented in Figure 8.28 and Figure 8.29, respectively, along with the maximum limit of the CSA. The extracted parameters from measurements performed on pMOS devices are displayed in Figure 8.30. It appears that the η values are proportional to the gate area under the assumption of a unimodal distribution of step heights.

To estimate the impact of a single defect on ΔV_{th} , the CSA (4.26), is often employed in device simulators. In this work, the extracted values of the threshold voltage shift obtained from CCDFs are compared with the maximum impact predicted by the CSA, which occurs when $x_{\text{T}} = 0$, i.e., directly at the interface. It is evident from the dashed lines in Figure 8.28 and Figure 8.29 that, for both distributions, the CSA significantly underestimates the extracted average impact of the defects, η . As a result, defect densities derived from ΔV_{th} will yield overly pessimistic results. The findings presented here are of particular importance for improving charge-trapping models and enhancing the accuracy of simulations.

8.2.6 Simulation Results

The BTI simulator Comphy [97] is used for verification of the distributions, which were extracted in the previous sections. Comphy uses an one-dimensional gate-stack for which the surface potential is computed at a given gate voltage using several input quantities such as doping concentrations or the work-function difference for the ideal device. During the execution of a BTI simulation, defects are sampled in the oxide.



Figure 8.29. The extracted values for the two modes of the bimodal exponential distribution are shown. As can be seen from Figure 8.25, the distribution follows more a bimodal exponential behavior than a unimodal one. Again, the CSA significantly underestimates the impact of the defects, which leads to a too pessimistic estimation for defect density from ΔV_{th} . Originally published in [KTC2].



Figure 8.30. The extracted values for the unimodal exponential distribution of the hole traps of pMOS devices. The η values seem to be proportional to the gate area. The max limit of CSA approximation is around three times smaller than values extracted from the CCDFs. Originally published in [KTC2].

Capture and emission rates of these pre-existing defects can be computed using a twostate nonradiative-multiphonon model [76]. For the computation of the rates, the barrier heights ϵ_{ij} between the defect and the band edges needs to be computed.

For this, the trap level E_T , the relaxation energy E_R and the ratio of the curvatures of the potential energy surfaces R is needed. Additionally, it is necessary to know the spatial position x_T of the defect for computing the Wentzel-Kramers-Brillouin (WKB) factor. By sampling E_T , E_R , x_T for a fixed R-value, it is possible to compute the transition rates:

$$k_{ij} = n v_{\rm th} \vartheta_{\rm WKB} \sigma \exp\left(-\frac{\epsilon_{ij}}{k_{\rm B}T}\right)$$
(8.2)

with *n* being the carrier concentration, the thermal velocity v_{th} , the WKB factor ϑ_{WKB} and the capture cross-section σ [76]. The defect bands extracted in [KTJ3] are used for



Figure 8.31. Extracted defect bands from eMSM measurements on the SiON devices. nMOS bands are magnified to be accurately visible in the figure. Recreated from [KTJ3].



Figure 8.32. Samples are obtained from the extracted defect distributions within the band diagram. The search regions used in the ESiD algorithm are represented by shaded areas. The highest concentration of both electron (blue) and hole (red) traps is found within the IL. Originally published [KTJ3].

the simulations, which have been extracted for large-area devices using a non-negative least square approach for finding optimal defect distributions semi-automatically. The extracted defect band can be seen in Figure 8.31 and the positions of the defects in the device in Figure 8.32. Defects are randomly drawn and an η -value for every contributing defect is sampled from these defect bands using the CCDFs extracted in sections 8.2.2 and 8.2.4. Using this setup, it is possible to simulate the measured MSM-traces from large area devices presented in [KTJ3], as shown in the top part of Figure 8.33 for nMOS devices and Figure 8.34 for the pMOS ones. As can be seen, the step heights extracted from the CCDFs allow a precise simulation of the measurement data. The lower figures show recovery traces for the measurement with the highest oxide field for two selected regions (marked grey) with more detail. The solid lines show the average of all recovery



Figure 8.33. In the top figure MSM measurements (dots) on large-area nMOS devices are shown, as presented in [KTJ3], at T = 100 °C under three different stress conditions. The lines are simulated with the BTI simulator Comphy averaged for 100 sets of sampled step heights using the CCDFs extracted in section 8.2.2. The regions marked in grey are shown in more detail in the two bottom figures. Every light line presents a BTI simulation with a sampled set of step heights, the solid line is the average of all performed simulations. Originally published in [KTJ4].

traces with different sampled step heights.

8.2.7 Conclusions

The distribution of step heights is crucial in characterizing the impact of single defects in nanoscale devices. In this study, MSM measurements were conducted for both NBTI and PBTI stress conditions. Specifically, the utilized devices were of dimensions small enough to enable the detection of individual steps during charge transitions. By analyzing these detected steps, the distributions of step heights are created and demonstrated that bimodal distributions are necessary for an accurate description, particularly in the case of PBTI/nMOS. Notably, the extracted distributions yielded higher average step heights compared to the commonly used CSA employed in simulations. This discrepancy indicates that the contribution of single defects is underestimated when relying solely on the CSA.

Subsequently, CCDFs were incorporated into Comphy and successfully accounted for the measurements obtained from large-area devices. This approach highlights the significance of considering realistic step height distributions in device simulations to better capture the influence of single defects on device performance.



Figure 8.34. In the top figure MSM measurements (dots) on large-area pMOS devices are shown, as presented in [KTJ3], at T = 100 °C under three different stress conditions. The lines are simulated with the BTI simulator Comphy averaged for 100 sets of sampled step heights using the CCDFs extracted in Section 8.2.4. The regions marked in grey are shown in more detail in the two bottom figures. Every light line presents a BTI simulation with a sampled set of step heights, the solid line is the average of all performed simulations.

8.3 Statistical Characterization of RTN in high-*κ* devices using Smart Arrays

In addition to the previously discussed BTI measurements, RTN measurements provide valuable insights into the trapping kinetics of individual defects. For pMOS devices unlike BTI, which freezes out at cryogenic temperatures, RTN continues to be a persistent reliability concern even at these low temperatures, regardless of the channel materials and device geometries employed [126]. In the upcoming section, RTN measurements conducted on smart arrays [141, 142] comprising high-k devices will be explored.

8.3.1 Devices and Measurements

In this section, the RTN measurements were conducted using a commercially available bulk CMOS technology with a metal gate contact and a high- κ layer of HfO₂. The technology utilizes a thin interface layer created through rapid thermal oxidation on the substrate. The high- κ layer has an EOT of 1.41 nm. The measurements involved devices with a width of 100 nm and different lengths of 70 nm, 100 nm, 135 nm, 170 nm, and 200 nm. The measurement array consisted of 2500 transistors, including both nMOS and pMOS devices. The first 50 gate lines allowed the selection of 500 devices with the smallest geometry using ten drain lines. The subsequent drain lines corresponded to devices with progressively larger geometries.

To collect statistical data on the distribution of step heights, a standardized characterization sequence was followed for each tested device. The sequence began with an initial $I_D(V_G)$ characterization, which ensured proper device operation, determined the gate voltages for RTN measurements, and established a correlation between recorded drain-source currents and equivalent threshold voltage shifts. Figure 6.2 illustrates the initial $I_D(V_G)$ measurements of the biggest geometry and the bias range used to capture the RTN signals.

After the initial $I_D(V_G)$ curve, the sequence continued with RTN traces taken at five different constant biases near V_{th} , where the RTN measurement window is available [50]. At each gate voltage, RTN traces were recorded with a sampling time T_s of 100 µs and a duration t_r of 1 s. Each device was measured for these RTN traces. The measurements were conducted at four different temperatures: 4 K, 77 K, 225 K, and 300 K.

8.4 Extraction of Step Heights

For each recorded trace, the steps were identified using Otsu's method (see Section 7.2), allowing the extraction of distributions for τ_c , τ_e , and η . Examples of these distributions can be observed in Figures 8.36 and 8.37. By utilizing these extracted distributions, the mean values of τ_c , τ_e , and η were calculated for each V_G within the measured range.



Figure 8.35. Initial $I_D(V_G)$ measurements of the biggest geometry in the bias range to capture the RTN signals. Originally published in [51, 143].

Compared to the Canny edge detector, Otsu's method offers the advantage of fully automated thresholding. It is often advantageous to apply a filter before automated thresholding to suppress measurement noise, although this may result in the loss of very fast charge transition events. However, it significantly enhances the robustness of thresholding and enables the detection of small step heights. In this study, the denoising algorithm proposed by Chambolle was employed for this purpose [128].

It is important to note that Otsu's method can only be applied to measurement data containing a single active defect. In cases where multiple active defects are detected, typically only the most prominent defect signal is extracted (referred to as multi-level RTN). To prevent incorrect detection, additional sanity checks, such as the χ^2 test and Kolmogorov-Smirnov test, are implemented to assess whether the capture and emission times follow an exponential distribution. For a comprehensive analysis of multi-level RTN, advanced techniques like factorial hidden Markov model analysis are required [129].

Figure 8.36 presents a non ideal measurement scenario for an pMOS device where even though it seems to have an appropriate RTN behavior, the two used tests lead us to reject the trace for conducting statistics. The ideal scenario of a used trace can be seen in Figure 8.37.



Figure 8.36. After identifying the distinctive levels of an arbitrary RTN signal (shown at the top), various parameters can be derived, such as step magnitudes (shown at bottom left) and the times for charge capture and emission (shown at bottom right and middle). However, in this particular signal, the distributions of t_e and t_c fail to meet the criteria set by the χ^2 and Kolmogorov-Smirnov tests. This exclusion ensures the avoidance of erroneous detections, thus making this particular trace unsuitable for statistical analysis.



Figure 8.37. After identifying the distinct steps present in an arbitrary RTN signal (depicted at the top), numerous parameters can be derived, such as step magnitudes (shown at bottom left) and the times for charge capture and emission (shown at bottom right and middle). In this particular trace, the distributions of t_e and t_c successfully pass both the χ^2 and Kolmogorov-Smirnov tests, ensuring accurate detection. As a result, this trace is suitable for utilization in the statistical analysis.

8.5 Distributions of Step Heights

The CCDFs of the steps extracted for the RTN traces of pMOS devices can be seen in the following Figures for each different temperature condition. In Figures 8.38 and 8.39 the CCDFs of T=4 K and T=77 K are shown. At these temperatures, the results cannot be described correctly by the commonly used exponential or lognormal distributions [144]. The step heights seem to distribute randomly, and from the lack of clear slopes, an area scaling is not possible to observe.



Figure 8.38. The CCDFs of step heights of hole traps extracted after RTN measurements on pMOS devices at T= 4 K. The dashed lines depict the unimodal exponential distribution that has been used to describe the results. At this temperature, the exponential distribution seems unable to describe the results, also the mean values do not scale with the area but present a random behavior.



Figure 8.39. The CCDFs of step heights of hole traps extracted after RTN measurements on pMOS devices at T = 7 K. The dashed lines depict the unimodal exponential distribution that has been used to describe the results. Similar to the T = 7 K case, the exponential distribution seems unable to describe the results and also the mean values seem to not scale with the area but present a random behavior, too.

Contrary to that, in Figures 8.40 and 8.41 where CCDFs of T=225K and T=300K are shown, exponential distributions seem to capture the results nicely. One possible

explanation is that the number of defects for the larger geometries is smaller in lower temperatures, see Figure 8.42 and thus accurate distributions cannot be observed.



Figure 8.40. The CCDFs of step heights of hole traps extracted after RTN measurements on pMOS devices at T= 225 K. The dashed lines depict the unimodal exponential distribution that has been used to describe the results. At this temperature, the exponential distribution captures nicely the results. Also, a scaling with the area starts to exist.



Figure 8.41. The CCDFs of step heights of hole traps extracted after RTN measurements on pMOS devices at T = 300 K. The dashed lines depict the unimodal exponential distribution that has been used to describe the results. At this temperature similar to the T = 225 K case, the exponential distribution captures nicely, and the area scaling is clear.

8.6 Parameter Extraction

Extracted η values of the CCDFs of the previous section can be seen in Figures 8.43 and 8.44 plotted against the area. The markers of red shade represent the experimental η values for each geometry. The red line is the extrapolated area trend, and the grey dashed line represents the maximum value expected from CSA for a single defect. The results seem to follow area scaling in a consistent way with RMSE = 0.11 and RMSE = 0.9 for



Figure 8.42. The number of devices exhibiting RTN versus temperature for all the different areas. Even though the results are mixed, more devices exhibiting RTN can be seen in higher temperatures.

temperatures T = 225 K and T = 300 K accordingly. The experimental values are higher than the maximum limit of CSA, consistent with what has been observed in the other studied geometries.

The η values are also plotted against the $W \times \sqrt{L}$ trend, which described the results of the first studied technology better than the area trend, in Figures 8.43 and 8.44 accordingly. Interestingly, the results follow the $W \times \sqrt{L}$ trend with the same RMSE values as in the area case. This observation can attributed to the small range of *L* in the studied geometries. To better analyze the two trends, a broader set of geometries is needed, and ideally, devices with the same area and different W/L ratios as in the SiO₂ case of the previous section.



Figure 8.43. The extracted η values for the T= 225 K case presented in Figure 8.40 are plotted against the area. The results follow the area trend except for a small outlier at L=170 nm. The experimental values are higher than the maximum limit of CSA, similar to the other studies presented in the thesis.



Figure 8.44. The extracted η values for the T= 225 K case presented in Figure 8.41 are plotted against the area. The results follow the area trend with a smaller RMSE than in the T= 225 K case. The experimental values are higher than the maximum limit of CSA, although their deviation is smaller than the T= 225 K case.



Figure 8.45. The extracted η values for the T= 225 K case presented in Figure 8.40 are plotted against the $W \times \sqrt{L}$ trend. The results follow the trend with the same RMSE as observed in the area trend.



Figure 8.46. The extracted η values for the T= 300 K case presented in Figure 8.41 are plotted against the $W \times \sqrt{L}$ trend. The results follow the trend with the same RMSE as observed in the area trend.

Chapter 9

Summary and Outlook

9.1 Summary

The MOSFET has been critical in many technological advancements over the past decades. Ensuring their reliable function under diverse operating conditions, such as various biases and temperatures, is crucial for the consistent performance of the circuits they comprise. In this thesis, reliability issues in MOSFET devices are explored, focusing specifically on bias temperature instability (BTI) and random telegraph noise (RTN), effects arising from individual defects. These defects originate from the intrinsic device structure, from manufacturing procedures, or develop during device usage and create localized electric field fluctuations, capture charges, and form interface states, negatively impacting the device's performance.

In the studies included in this thesis, the impact of single defects on the device performance is studied by leveraging MSM measurements focusing on threshold voltage shift ΔV_{th} , one of the most critical parameters for the device reliability. The main focus is the statistical analysis of the discrete amplitudes of single defects, which can be extracted from ΔV_{th} when nanoscale devices are employed. By creating statistical distributions based on these amplitudes, essential metrics for the impact of defects can be extracted, such as the average number of active defects per device N_{T} and the average threshold shift induced by a single defect η . The investigation of devices of different technologies and geometries under various bias conditions led to insights into how parameters, such as gate width and length, affect the impact of single defects. In addition to the experimental distributions, defect-centric model (DCM) is used to validate these approaches.

More specifically, from the analysis of the SiO₂ technology on pMOS devices, it is observed that for measurements close to threshold voltage V_{th} , the scaling of η is described better by a $W \times \sqrt{L}$ trend rather than the area of the devices $W \times L$, which is typically used in literature. However, if a higher recovery bias is applied η decreases, especially on devices affected more by the percolative conduction, and the $W \times L$ trends capture nicely the values for all the tested geometries. At the same time, more defects contribute to the observed degradation for higher recovery bias conditions. From the measurements on the same technology with varying body bias, modulations of both time-zero variability and time-dependent variability are found. These observations were explained by the modulation of the number of unscreened dopant atoms within the channel depletion region induced by the body bias. It was shown that using reverse body bias (RBB) can increase both depletion region thickness X_{dep} and $|V_{th}|$, making it advantageous for applications that require low leakage. On the other hand, forward body bias (FBB) reduces X_{dep} and $|V_{th}|$, which can be beneficial for high-performance applications, albeit with a trade-off of higher junction leakage.

The statistical distributions obtained for SiON technology revealed that bimodal distributions are necessary for an accurate description of the amplitudes of step heights, particularly in the case of PBTI/nMOS. Nevertheless, the dimensional scaling trends for the step heights for a bimodal distribution for nMOS data still need to be fully resolved and addressed in future studies. Subsequently, the observed distributions were incorporated into reliability simulator Comphy and successfully accounted for the measurements obtained from large-area devices. This approach highlights the importance of considering realistic step height distributions in device simulations to better capture the influence of single defects on device performance.

The knowledge gained from the studies presented in this thesis can further be applied to different technologies and device structures to determine the impact of single defects on their performance. The integration of experimental measurements, statistical modeling, and compact physics simulations lead to a comprehensive understanding of the defect nature in MOSFETs and pave the way for improved device performance and reliability.

9.2 Outlook

The findings presented in this thesis contribute to the understanding of the link between device geometry and charge trapping effects, which is essential for making circuits resilient against charge trapping. Building upon the insights gained from the current study, several key areas warrant further exploration.

Future research can involve incorporating gate current measurements in addition to the measurements presented in this thesis. The study of defect charge trapping behavior in this thesis primarily focuses on its impact on the drain-source current. However, these defects also affect the gate current, either through direct charge exchange with the gate or by affecting tunneling currents [145, 146]. Additionally, the new trap-assisted tunneling (TAT) model implemented into Comphy [KTJ6, 147] can be used to simultaneously study BTI and TAT within a single technology. To do this effectively, it is necessary to design test structures with sufficiently large gate areas to detect small leakage currents and smaller structures for detailed BTI analysis.

Furthermore, investigating the impact of the device geometry on η for different technologies and novel device structures can shed more light on the physical mechanisms of single defects. Experimental test structures with strategically placed defects

can be used to connect the results obtained from the methods used in this thesis with the predetermined characteristics of the injected defects. Such an analysis can shed light on the relationship between device geometry and the severity of reliability issues, enabling the development of design guidelines to optimize MOSFET reliability in various configurations.

In addition to the impact of gate geometry and body bias on the impact of single defects, future research can also explore the influence of other device parameters, such as dopant concentration. Identifying how devices with different dopant concentrations can interact under different body bias conditions can give a more precise understanding of how substrate dopants affect the device performance [139].

By systematically varying these parameters, researchers can comprehensively understand their effects on the occurrence and severity of reliability issues. This knowledge can guide device manufacturers to make informed decisions during the fabrication process to enhance the reliability of MOSFETs. Finally, these ideas for future research can lead to a complete statistical model for η where various device parameters and operating conditions that affect device reliability can be included.

Finally, additional studies to address strategies to mitigate the impact of defects would naturally be a next step based on the work of this thesis. Having characterized the impact of defects, which allows the proper incorporation into a statistical modeling and simulation environment, controlling or suppressing their impact based on the affecting parameters is promising for future research.



References

- [1] C. Hu. *Modern Semiconductor Devices for Integrated Circuits*. Prentice Hall, 2010. ISBN: 9780136085256.
- [2] B. Razavi. *Fundamentals of Microelectronics*. John Wiley & Sons, 2013. ISBN: 9781118156322.
- [3] D. K. Schroder. "Negative bias temperature instability: What do we understand?" In: *Microelectronics Reliability* 47.6 (2007), pp. 841–852. DOI: https: //doi.org/10.1016/j.microrel.2006.10.006.
- [4] J. H. Stathis, S. Mahapatra, and T. Grasser. "Controversial Issues in Negative Bias Temperature Instability". In: *Microelectronics Reliability* 81 (2018), pp. 244– 251. DOI: 10.1016/j.microrel.2017.12.035.
- [5] J. E. Lilienfeld. "Method and Apparatus for Controlling Electric Currents". US1745175. Filed 1926. Jan. 28. Jan. 1930.
- [6] W. Shockley and G. Pearson. "Modulation of Conductance of Thin Films of Semi-Conductors by Surface Charges". In: *Physical Review* 74.2 (1948), pp. 232.
 DOI: 10.1103/PhysRev.74.232.
- [7] F. Wanlass and C. Sah. "Nanowatt logic using field-effect metal-oxide semiconductor triodes". In: 1963 IEEE International Solid-State Circuits Conference. Digest of Technical Papers. Vol. VI. 1963, pp. 32–33. DOI: 10.1109/ISSCC.1963. 1157450.
- [8] G. E. Moore. Cramming more components onto integrated circuits, Reprinted from Electronics, volume 38, number 8, April 19, 1965, pp.114 ff. 2006. DOI: 10.1109/N– SSC.2006.4785860.
- [9] G. D. Wilk, R. M. Wallace, and J. M. Anthony. "High-gate dielectrics: Current status and materials properties considerations". In: *Journal of Applied Physics* 89.10 (May 2001), pp. 5243–5275. DOI: 10.1063/1.1361065.
- [10] Robertson, J. "High dielectric constant oxides". In: *Eur. Phys. J. Appl. Phys.* 28.3 (2004), pp. 265–291. DOI: 10.1051/epjap:2004206.

- [11] L. Witters, S. Takeoka, S. Yamaguchi, A. Hikavyy, D. Shamiryan, M. Cho, T. Chiarella, L.-Å. Ragnarsson, R. Loo, C. Kerner, Y. Crabbe, J. Franco, J. Tseng, W. Wang, E. Rohr, T. Schram, O. Richard, H. Bender, S. Biesemans, P. Absil, and T. Hoffmann. "8Å Tinv gate-first dual channel technology achieving low-Vt high performance CMOS". In: 2010 Symposium on VLSI Technology. 2010, pp. 181–182. DOI: 10.1109/VLSIT.2010.5556219.
- [12] E. Cartier, A. Kerber, T. Ando, M. M. Frank, K. Choi, S. Krishnan, B. Linder, K. Zhao, F. Monsieur, J. Stathis, and V. Narayanan. "Fundamental aspects of HfO2-based high-k metal gate stack reliability and implications on tinvscaling". In: 2011 International Electron Devices Meeting. 2011, pp. 18.4.1–18.4.4. DOI: 10.1109/IEDM.2011.6131579.
- [13] M. S. Narula and A. Pandey. "A Comprehensive Review on FinFET, Gate All Around, Tunnel FET: Concept, Performance and Challenges". In: 2022 8th International Conference on Signal Processing and Communication (ICSC). 2022, pp. 554–559. DOI: 10.1109/ICSC56524.2022.10009504.
- [14] L. Qin, C. Li, Y. Wei, G. Hu, J. Chen, Y. Li, C. Du, Z. Xu, X. Wang, and J. He. "Recent Developments in Negative Capacitance Gate-All-Around Field Effect Transistors: A Review". In: *IEEE Access* 11 (2023), pp. 14028–14042. DOI: 10.1109/ACCESS.2023.3243697.
- [15] International Technology Roadmap for Semiconductors. 2020. URL: http://www. itrs2.net/.
- [16] IEEE International Roadmap for Devices and Systems. 2022. URL: https://irds. ieee.org/.
- [17] "Electromigration in submicron interconnect features of integrated circuits". In: *Materials Science and Engineering: R: Reports* 71.5 (2011), pp. 53–86. ISSN: 0927-796X. DOI: https://doi.org/10.1016/j.mser.2010.09.001.
- [18] R. de Orio, H. Ceric, and S. Selberherr. "Physically based models of electromigration: From Black's equation to modern TCAD models". In: *Microelectronics Reliability* 50.6 (2010). 2009 Reliability of Compound Semiconductors (ROCS) Workshop), pp. 775–789. ISSN: 0026-2714. DOI: https://doi.org/10.1016/ j.microrel.2010.01.007. URL: https://www.sciencedirect.com/science/ article/pii/S0026271410000193.
- [19] A. Ghetti. "Gate Oxide Reliability: Physical and Computational Models". In: *Predictive Simulation of Semiconductor Processing: Status and Challenges*. Ed. by J. Dabrowski and E. R. Weber. Berlin, Heidelberg: Springer Berlin Heidelberg, 2004, pp. 201–258. ISBN: 978-3-662-09432-7. DOI: 10.1007/978-3-662-09432-7_6. URL: https://doi.org/10.1007/978-3-662-09432-7_6.
- [20] J. Suehle. "Ultrathin gate oxide reliability: physical models, statistics, and characterization". In: *IEEE Transactions on Electron Devices* 49.6 (2002), pp. 958–971. DOI: 10.1109/TED.2002.1003712.

- [21] M. Jech. "The Physics of Non–Equilibrium Reliability Phenomena". PhD thesis. Institut für Mikroelektronik, 2022. DOI: 10.34726/hss.2020.85163.
- [22] M. Denais, V. Huard, C. Parthasarathy, G. Ribes, F. Perrier, N. Revil, and A. Bravaix. "Interface trap generation and hole trapping under NBTI and PBTI in advanced CMOS technology with a 2-nm gate oxide". In: *IEEE Transactions on Device and Materials Reliability* 4.4 (2004), pp. 715–722. DOI: 10.1109/TDMR. 2004.840856.
- [23] T. Grasser, B. Kaczer, W. Gös, H. Reisinger, T. Aichinger, P. Hehenberger, P.-J. Wagner, J. Franco, M. Toledano-Luque, and M. Nelhiebel. "The Paradigm Shift in Understanding the Bias Temperature Instability: From Reaction-Diffusion to Switching Oxide Traps". In: *IEEE Transactions on Electron Devices* 58.11 (Nov. 2011), pp. 3652–3666. DOI: 10.1109/ted.2011.2164543.
- [24] B. Kaczer, T. Grasser, P. J. Roussel, J. Franco, R. Degraeve, L.-A. Ragnarsson, E. Simoen, G. Groeseneken, and H. Reisinger. "Origin of NBTI variability in deeply scaled pFETs". In: 2010 IEEE International Reliability Physics Symposium. 2010, pp. 26–32. DOI: 10.1109/IRPS.2010.5488856.
- [25] M. J. Uren, M. J. Kirton, and S. Collins. "Anomalous Telegraph Noise in Small-Area Silicon Metal-Oxide-Semiconductor Field-Effect Transistors". In: *Physical Review B* 37 (14 May 1988), pp. 8346–8350. DOI: 10.1103/PhysRevB.37.8346.
- [26] A. Ghetti, C. Monzio Compagnoni, A. S. Spinelli, and A. Visconti. "Comprehensive Analysis of Random Telegraph Noise Instability and Its Scaling in Deca–Nanometer Flash Memories". In: *IEEE Transactions on Electron Devices* 56.8 (2009), pp. 1746–1752. DOI: 10.1109/TED.2009.2024031.
- [27] M. Luo, R. Wang, S. Guo, J. Wang, J. Zou, and R. Huang. "Impacts of Random Telegraph Noise (RTN) on Digital Circuits". In: *IEEE Transactions on Electron Devices* 62.6 (June 2015), pp. 1725–1732. DOI: 10.1109/ted.2014.2368191.
- [28] E. Rosenbaum and L. Register. "Mechanism of stress-induced leakage current in MOS capacitors". In: *IEEE Transactions on Electron Devices* 44.2 (1997), pp. 317–323. DOI: 10.1109/16.557724.
- [29] W. Goes, M. Waltl, Y. Wimmer, G. Rzepa, and T. Grasser. "Advanced Modeling of Charge Trapping: RTN, 1/f Noise, SILC, and BTI". In: 2014 International Conference on Simulation of Semiconductor Processes and Devices (SISPAD). IEEE, Sept. 2014, pp. 77–80. DOI: 10.1109/sispad.2014.6931567.
- [30] S. Abbas and R. Dockerty. "Hot-Carrier Instability in IGFET's". In: *Applied Physics Letters* 27.3 (1975), pp. 147–148. DOI: 10.1063/1.88387.
- [31] T. Ning, P. Cook, R. Dennard, C. Osburn, and S. Schuster. "1 /spl mu/m MOS-FET VLSI technology. IV. Hot-electron design constraints". In: *IEEE Journal of Solid-State Circuits* 14.2 (1979), pp. 268–275. DOI: 10.1109/JSSC.1979.1051173.

- [32] A. Asenov, S. Roy, R. A. Brown, G. Roy, C. Alexander, C. Riddet, C. Millar, B. Cheng, A. Martinez, N. Seoane, D. Reid, M. F. Bukhori, X. Wang, and U. Kovac. "Advanced simulation of statistical variability and reliability in nano CMOS transistors". In: 2008 IEEE International Electron Devices Meeting. 2008, pp. 1–1. DOI: 10.1109/IEDM.2008.4796712.
- [33] K. J. Kuhn, M. D. Giles, D. Becher, P. Kolar, A. Kornfeld, R. Kotlyar, S. T. Ma, A. Maheshwari, and S. Mudanai. "Process Technology Variation". In: *IEEE Transactions on Electron Devices* 58.8 (2011), pp. 2197–2208. DOI: 10.1109/TED. 2011.2121913.
- [34] A. Asenov, A. Brown, J. Davies, S. Kaya, and G. Slavcheva. "Simulation of intrinsic parameter fluctuations in decananometer and nanometer-scale MOSFETs". In: *IEEE Transactions on Electron Devices* 50.9 (2003), pp. 1837–1852. DOI: 10.1109/TED.2003.815862.
- [35] H. Dadgour, K. Endo, V. De, and K. Banerjee. "Modeling and analysis of grain-orientation effects in emerging metal-gate devices and implications for SRAM reliability". In: 2008 IEEE International Electron Devices Meeting. 2008, pp. 1–4. DOI: 10.1109/IEDM.2008.4796792.
- [36] Mizuno, Okamura, and Toriumi. "Experimental Study Of Threshold Voltage Fluctuations Using An 8k MOSFET's Array". In: Symposium 1993 on VLSI Technology. 1993, pp. 41–42. DOI: 10.1109/VLSIT.1993.760235.
- [37] H.-S. Philip Wong, Y. Taur, and D. J. Frank. "Discrete random dopant distribution effects in nanometer-scale MOSFETs". In: *Microelectronics Reliability* 38.9 (1998), pp. 1447–1456. DOI: https://doi.org/10.1016/S0026-2714(98)00053-5.
- [38] T. Mizuno, J. Okumtura, and A. Toriumi. "Experimental study of threshold voltage fluctuation due to statistical variation of channel dopant number in MOSFET's". In: *IEEE Transactions on Electron Devices* 41.11 (1994), pp. 2216– 2221. DOI: 10.1109/16.333844.
- [39] K. J. Kuhn, M. D. Giles, D. Becher, P. Kolar, A. Kornfeld, R. Kotlyar, S. T. Ma, A. Maheshwari, and S. Mudanai. "Process Technology Variation". In: *IEEE Transactions on Electron Devices* 58.8 (2011), pp. 2197–2208. DOI: 10.1109/TED. 2011.2121913.
- [40] T. Grasser, H. Reisinger, P.-J. Wagner, F. Schanovsky, W. Goes, and B. Kaczer.
 "The time dependent defect spectroscopy (TDDS) for the characterization of the bias temperature instability". In: 2010 IEEE International Reliability Physics Symposium. 2010, pp. 16–25. DOI: 10.1109/IRPS.2010.5488859.
- [41] M.Waltl. "Reliability of Miniaturized Transistors from the Perspective of Single-Defects". In: *MDPI Micromachines* 11(8) (2020), pp. 736. DOI: 10.3390/ mi11080736.

- [42] M. Waltl. "Experimental Characterization of Bias Temperature Instabilities in Modern Transistor Technologies". PhD thesis. Institut für Mikroelektronik, 2016. DOI: https://doi.org/10.34726/hss.2016.38201.
- [43] M. Waltl. "Ultra-Low Noise Defect Probing Instrument for Defect Spectroscopy of MOS Transistors". In: *IEEE Transactions on Device and Materials Reliability* 20.2 (2020), pp. 242–250. DOI: 10.1109/TDMR.2020.2988650.
- [44] J. Franco, B. Kaczer, M. Toledano-Luque, P. J. Roussel, J. Mitard, L.-Å. Ragnarsson, L. Witters, T. Chiarella, M. Togo, N. Horiguchi, G. Groeseneken, M. F. Bukhori, T. Grasser, and A. Asenov. "Impact of single charged gate oxide defects on the performance and scaling of nanoscaled FETs". In: 2012 IEEE International Reliability Physics Symposium (IRPS). 2012, pp. 5A.4.1–5A.4.6. DOI: 10.1109/IRPS.2012.6241841.
- [45] M. Toledano-Luque, B. Kaczer, E. Simoen, P. Roussel, A. Veloso, T. Grasser, and G. Groeseneken. "Temperature and Voltage Dependences of the Capture and Emission Times of Individual Traps in High-k Dielectrics". In: *Microelectronics Engineering* 88.7 (2011), pp. 1243–1246. DOI: 10.1016/j.mee.2011.03.097.
- [46] M. Waltl, B. Stampfer, G. Rzepa, B. Kaczer, and T. Grasser. "Separation of electron and hole trapping components of PBTI in SiON nMOS transistors". In: *Microelectronics Reliability* 114 (2020), pp. 113746. DOI: 10.1016/j.microrel. 2020.113746.
- [47] B. Kaczer, J. Franco, P. Weckx, P. J. Roussel, E. Bury, M. Cho, R. Degraeve, D. Linten, G. Groeseneken, H. Kukner, P. Raghavan, F. Catthoor, G. Rzepa, W. Goes, and T. Grasser. "The defect-centric perspective of device and circuit reliability From individual defects to circuits". In: 2015 45th European Solid State Device Research Conference (ESSDERC). 2015, pp. 218–225. DOI: 10.1109/ESSDERC.2015.7324754.
- [48] B. Kaczer, M. Toledano-Luque, J. Franco, and P. Weckx. "Statistical Distribution of Defect Parameters". In: *Bias Temperature Instability for Devices and Circuits*. Ed. by T. Grasser. New York, NY: Springer New York, 2014, pp. 161–176. ISBN: 978-1-4614-7909-3. DOI: 10.1007/978-1-4614-7909-3_7.
- [49] D. Waldhoer, A.-M. B. El-Sayed, Y. Wimmer, M. Waltl, and T. Grasser. "Atomistic Modeling of Oxide Defects". In: *Noise in Nanoscale Semiconductor Devices*. Springer International Publishing, 2020, pp. 609–648. DOI: 10.1007/978-3-030-37500-3_18.
- [50] B. Stampfer. "Advanced electrical characterization of charge trapping in MOS transistors". In: Dissertation, Technische Universität Wien (2020). DOI: https://doi.org/10.34726/hss.2020.86423.
- [51] J. Michl. "Charge Trapping and Variability in CMOS Technologies at Cryogenic Temperatures". PhD thesis. Institut für Mikroelektronik, 2022. DOI: 10.34726/ hss.2022.105545.

- [52] M. L. Reed and J. D. Plummer. "Chemistry of Si-SiO₂ interface trap annealing". In: *Journal of Applied Physics* 63.12 (June 1988), pp. 5776–5793. DOI: 10.1063/1. 340317.
- [53] C. R. Helms and E. H. Poindexter. "The silicon-silicon dioxide system: Its microstructure and imperfections". In: *Reports on Progress in Physics* 57.8 (Aug. 1994), pp. 791. DOI: 10.1088/0034-4885/57/8/002.
- [54] P. J. Caplan, E. H. Poindexter, B. E. Deal, and R. R. Razouk. "ESR Centers, Interface States, and Oxide Fixed Charge in Thermally Oxidized Silicon Wafers". In: *Journal of applied Physics* 50.9 (1979), pp. 5847–5854. DOI: 10.1063/1.326732.
- [55] Y. Nishi. "Study of silicon-silicon dioxide structure by electron spin resonance I". In: *Japanese Journal of Applied Physics* 10.1 (1971), pp. 52. DOI: 10.1143/JJAP. 10.52.
- [56] G. J. Gerardi, E. H. Poindexter, P. J. Caplan, and N. M. Johnson. "Interface Traps and Pb Centers in Oxidized (100) Silicon Wafers". In: *Applied Physics Letters* 49 (1986), pp. 348–350. DOI: 10.1063/1.97611.
- [57] N. H. Thoan, K. Keunen, V. V. Afanas'ev, and A. Stesmans. "Interface State Energy Distribution and Pb Defects at Si(110)/SiO2 Interfaces: Comparison to (111) and (100) Silicon Orientations". In: *Journal of Applied Physics* 109 (2011), pp. 013710. ISSN: 0021-8979. DOI: 10.1063/1.3527909.
- [58] E. H. Poindexter. "MOS Interface States: Overview and Physicochemical Perspective". In: *Semiconductor Science and Technology* 4.12 (Dec. 1989), pp. 961–969. DOI: 10.1088/0268-1242/4/12/001.
- [59] B. Stampfer, A. Grill, and M. Waltl. "Advanced Electrical Characterization of Single Oxide Defects Utilizing Noise Signals". In: *Noise in Nanoscale Semiconductor Devices*. Springer International Publishing, 2020, pp. 229–257. DOI: 10.1007/978-3-030-37500-3_7.
- Y. Wimmer, A.-M. El-Sayed, W. Goes, T. Grasser, and A. L. Shluger. "Role of Hydrogen in Volatile Behaviour of Defects in SiO2-based Electronic Devices". In: *Proc. R. Soc. A.* Vol. 472. 2190. The Royal Society. 2016, pp. 20160009. DOI: 10.1098/rspa.2016.0009.
- [61] T. Aichinger, S. Puchner, M. Nelhiebel, T. Grasser, and H. Hutter. "Impact of hydrogen on recoverable and permanent damage following negative bias temperature stress". In: 2010 IEEE International Reliability Physics Symposium. IEEE. IEEE, 2010, pp. 1063–1068. DOI: 10.1109/irps.2010.5488672.
- [62] V. V. Afanas'ev and A. Stesmans. "Interfacial Defects in SiO₂ Revealed by Photon Stimulated Tunneling of Electrons". In: *Phys. Rev. Lett.* 78 (12 Mar. 1997), pp. 2437–2440. DOI: 10.1103/PhysRevLett.78.2437.

- [63] J. P. Campbell and P. M. Lenahan. "Atomic-Scale Defects Associated with the Negative Bias Temperature Instability". In: *Bias Temperature Instability for Devices and Circuits*. Ed. by T. Grasser. New York, NY: Springer New York, 2014, pp. 177–228. ISBN: 978-1-4614-7909-3. DOI: 10.1007/978-1-4614-7909-3_8.
- [64] W. L. Warren and P. M. Lenahan. "Electron-nuclear double-resonance and electron-spin-resonance study of silicon dangling-bond centers in silicon nitride". In: *Phys. Rev. B* 42 (3 July 1990), pp. 1773–1780. DOI: 10.1103/PhysRevB. 42.1773.
- [65] D. Griscom, E. Friebele, and G. Sigel. "Observation and analysis of the primary 29Si hyperfine structure of the E center in non-crystalline SiO2". In: *Solid State Communications* 15.3 (1974), pp. 479–483. DOI: 10.1016/0038-1098(74)91124-7.
- [66] G. Rzepa. "Efficient Physical Modeling of Bias Temperature Instability". PhD thesis. Institut für Mikroelektronik, 2018. DOI: 10.34726/hss.2018.57326.
- [67] J. Strand, M. Kaviani, D. Gao, A.-M. El-Sayed, V. V. Afanas'ev, and A. L. Shluger. "Intrinsic charge trapping in amorphous oxide films: status and challenges". In: *Journal of Physics: Condensed Matter* 30.23 (May 2018), pp. 233001. DOI: 10.1088/1361-648X/aac005.
- [68] F. Bohra, B. Jiang, and J.-M. Zuo. "Textured crystallization of ultrathin hafnium oxide films on silicon substrate". In: *Applied physics letters* 90.16 (2007), p. 161917.
- [69] J. Strand, O. A. Dicks, M. Kaviani, and A. L. Shluger. "Hole trapping in amorphous HfO2 and Al2O3 as a source of positive charging". In: *Microelectronic Engineering* 178 (2017). Special issue of Insulating Films on Semiconductors (INFOS 2017), pp. 235–239. ISSN: 0167-9317. DOI: https://doi.org/10.1016/j.mee.2017.05.012.URL: https://www.sciencedirect.com/science/article/pii/S0167931717302095.
- [70] M. Kaviani, J. Strand, V. V. Afanas'ev, and A. L. Shluger. "Deep electron and hole polarons and bipolarons in amorphous oxide". In: *Phys. Rev. B* 94 (2 July 2016), pp. 020103. DOI: 10.1103/PhysRevB.94.020103.
- [71] F. Cerbu, O. Madia, D. V. Andreev, S. Fadida, M. Eizenberg, L. Breuil, J. G. Lisoni, J. A. Kittl, J. Strand, A. L. Shluger, V. V. Afanas'ev, M. Houssa, and A. Stesmans. "Intrinsic electron traps in atomic-layer deposited HfO2 insulators". In: *Applied Physics Letters* 108.22 (May 2016), pp. 222901. DOI: 10.1063/1.4952718.
- [72] A. Shluger. "Defects in Oxides in Electronic Devices". In: Handbook of Materials Modeling: Applications: Current and Emerging Materials. Ed. by W. Andreoni and S. Yip. Cham: Springer International Publishing, 2018, pp. 1–22. ISBN: 978-3-319-50257-1. DOI: 10.1007/978-3-319-50257-1_79-1. URL: https: //doi.org/10.1007/978-3-319-50257-1_79-1.

- [73] A. Kerber and E. A. Cartier. "Reliability Challenges for CMOS Technology Qualifications With Hafnium Oxide/Titanium Nitride Gate Stacks". In: *IEEE Transactions on Device and Materials Reliability* 9.2 (2009), pp. 147–162. DOI: 10.1109/TDMR.2009.2016954.
- [74] F. Cerbu, O. Madia, D. V. Andreev, S. Fadida, M. Eizenberg, L. Breuil, J. G. Lisoni, J. A. Kittl, J. Strand, A. L. Shluger, V. V. Afanas'ev, M. Houssa, and A. Stesmans. "Intrinsic electron traps in atomic-layer deposited HfO2 insulators". In: *Applied Physics Letters* 108.22 (2016), pp. 222901. DOI: 10.1063/1.4952718.
- S. R. Bradley, G. Bersuker, and A. L. Shluger. "Modelling of oxygen vacancy aggregates in monoclinic HfO2: can they contribute to conductive filament formation?" In: *Journal of Physics: Condensed Matter* 27.41 (Sept. 2015), p. 415401. DOI: 10.1088/0953-8984/27/41/415401. URL: https://dx.doi.org/10.1088/0953-8984/27/41/415401.
- [76] T. Grasser. "Stochastic Charge Trapping in Oxides: From Random Telegraph Noise to Bias Temperature Instabilities". In: *Microelectronics Reliability* 52.1 (Jan. 2012), pp. 39–70. DOI: 10.1016/j.microrel.2011.09.002.
- [77] D. T. Gillespie. *Markov Processes: An Introduction for Physical Scientists*. Academic Press, 1992. ISBN: 978-0-12-283955-9.
- [78] N. G. Van Kampen. Stochastic Processes in Physics and Chemistry. Vol. 1. Elsevier, 1992.
- [79] W. Shockley and W. Read Jr. "Statistics of the recombinations of holes and electrons". In: *Physical review* 87.5 (1952), pp. 835. DOI: 10.1103/PhysRev.87.835.
- [80] M. Kirton and M. Uren. "Noise in Solid-State Microstructures: A New Perspective on Individual Defects, Interface States and Low-Frequency (1/f) Noise". In: *Adv. Phys.* 38.4 (1989), pp. 367–468. DOI: 10.1080/00018738900101122.
- [81] K. Huang and A. Rhys. "Theory of Light Absorption and Non-Radiative Transitions in F-Centres". In: *Proceedings of the Royal Society of London A: Mathematical, Physical and Engineering Sciences*. Vol. 204. 1078. 1950, pp. 406–423. DOI: 10.1098/rspa.1950.0184.
- [82] C. Henry and D. Lang. "Nonradiative Capture and Recombination by Multiphonon Emission in GaAs and GaP". In: *Physical Review B* 15.2 (1977), pp. 989. DOI: 10.1103/PhysRevB.15.989.
- [83] M. Born and R. Oppenheimer. "Zur Quantentheorie der Molekeln". In: *Annalen der Physik* 389.20 (1927), pp. 457–484. DOI: 10.1002/andp.19273892002.
- [84] J. Franck and E. G. Dymond. "Elementary Processes of Photochemical Reactions". In: *Transactions of the Faraday Society* 21 (February 1926), pp. 536–542. DOI: 10.1039/TF9262100536.

- [85] E. U. Condon. "Nuclear Motions Associated with Electron Transitions in Diatomic Molecules". In: *Physical Review* 32 (6 Dec. 1928), pp. 858–872. DOI: 10.1103/PhysRev.32.858.
- [86] E. Fermi. *Nuclear Physics: A Course Given by Enrico Fermi at the University of Chicago*. University of Chicago Press, 1950. ISBN: 9780226243658.
- [87] K. V. Mikkelsen and M. A. Ratner. "Electron tunneling in Solid-State Electron-Transfer Reactions". In: *Chemical Reviews* 87.1 (1987), pp. 113–153. DOI: 10. 1021/cr00077a007.
- [88] F. Schanovsky, O. Baumgartner, and T. Grasser. "Multi Scale Modeling of Multi Phonon Hole Capture in the Context of NBTI". In: 2011 International Conference on Simulation of Semiconductor Processes and Devices. IEEE. IEEE, Sept. 2011, pp. 15–18. DOI: 10.1109/sispad.2011.6035038.
- [89] W. Goes, Y. Wimmer, A.-M. El-Sayed, G. Rzepa, M. Jech, A. Shluger, and T. Grasser. "Identification of Oxide Defects in Semiconductor Devices: A Systematic Approach Linking DFT to Rate Equations and Experimental Evidence". In: *Microelectronics Reliability* 87 (2018), pp. 286–320. DOI: https: //doi.org/10.1016/j.microrel.2017.12.021.
- [90] T. Grasser, H. Reisinger, W. Goes, T. Aichinger, P. Hehenberger, P.-J. Wagner, M. Nelhiebel, J. Franco, and B. Kaczer. "Switching Oxide Traps as the Missing Link Between Negative Bias Temperature Instability and Random Telegraph Noise". In: *IEEE Int. Electron Devices Meet.* IEEE, Dec. 2009, pp. 1–4. DOI: 10.1109/iedm.2009.5424235.
- [91] B. Stampfer, F. Zhang, Y. Y. Illarionov, T. Knobloch, P. Wu, M. Waltl, A. Grill, J. Appenzeller, and T. Grasser. "Characterization of Single Defects in Ultrascaled MoS2 Field-Effect Transistors". In: ACS Nano 12.6 (June 2018), pp. 5368–5375. ISSN: 1936-0851. DOI: 10.1021/acsnano.8b00268.
- [92] A. Institut für Mikroelektronik Technische Universität Wien. *MINIMOS-NT User's Guide*, URL: http://www.iue.tuwien.ac.at/software/minimos-nt.
- [93] A. 2. M.-N. Global TCAD Solutions Vienna. *MINIMOS-NT User's Guide*, URL: https://www.globaltcad.com/minimos-nt.
- [94] C. U. Synopsys Mountain View. *Sentaurus Device*. 2022. URL: https://www. synopsys.com/silicon/tcad/device-simulation/sentaurus-device.
- [95] D. Waldhoer, C. Schleich, J. Michl, A. Grill, D. Claes, A. Karl, T. Knobloch, G. Rzepa, J. Franco, B. Kaczer, M. Waltl, and T. Grasser. "Comphy v3.0—A compact-physics framework for modeling charge trapping related reliability phenomena in MOS devices". In: *Microelectronics Reliability* 146 (2023), p. 115004. DOI: https://doi.org/10.1016/j.microrel.2023.115004.

- [96] G. Rzepa, J. Franco, A. Subirats, M. Jech, A. Chasin, A. Grill, M. Waltl, T. Knobloch, B. Stampfer, T. Chiarella, N. Horiguchi, L. A. Ragnarsson, D. Linten, B. Kaczer, and T. Grasser. "Efficient Physical Defect Model Applied to PBTI in High-κ Stacks". In: 2017 IEEE International Reliability Physics Symposium (IRPS). IEEE, Apr. 2017. DOI: 10.1109/irps.2017.7936425.
- [97] G. Rzepa, J. Franco, B. O'Sullivan, A. Subirats, M. Simicic, G. Hellings, P. Weckx, M. Jech, T. Knobloch, M. Waltl, P. Roussel, D. Linten, B. Kaczer, and T. Grasser. "Comphy A Compact-physics Framework for Unified Modeling of BTI". In: *Microelectronics Reliability* 85 (2018), pp. 49–65. DOI: https://doi.org/10.1016/j.microrel.2018.04.002.
- [98] S. M. Sze, Y. Li, and K. K. Ng. *Physics of semiconductor devices*. John wiley & sons, 2021. ISBN: 9780471143239.
- [99] M. A. Green. "Intrinsic concentration, effective densities of states, and effective mass in silicon". In: *Journal of Applied Physics* 67.6 (1990), pp. 2944–2954. DOI: 10.1063/1.345414. eprint: https://doi.org/10.1063/1.345414.
- [100] W. B. Joyce and R. W. Dixon. "Analytic approximations for the Fermi energy of an ideal Fermi gas". In: *Applied Physics Letters* 31.5 (1977), pp. 354–356. DOI: 10.1063/1.89697. eprint: https://doi.org/10.1063/1.89697.
- [101] W. Bludau, A. Onton, and W. Heinke. "Temperature dependence of the band gap of silicon". In: *Journal of Applied Physics* 45.4 (1974), pp. 1846–1848. DOI: 10.1063/1.1663501. eprint: https://doi.org/10.1063/1.1663501.
- [102] D. V. Lang. "Deep-level Transient Spectroscopy: A New Method to Characterize Traps in Semiconductors". In: *Journal of Applied Physics* 45.7 (1974), pp. 3023–3032. DOI: 10.1063/1.1663719.
- [103] T. Tewksbury and H.-S. Lee. "Characterization, modeling, and minimization of transient threshold voltage shifts in MOSFETs". In: *IEEE Journal of Solid-State Circuits* 29.3 (1994), pp. 239–252. DOI: 10.1109/4.278345.
- [104] O. Roux dit Buisson, G. Ghibaudo, and J. Brini. "Model for drain current RTS amplitude in small-area MOS transistors". In: *Solid-State Electronics* 35.9 (1992), pp. 1273–1276. ISSN: 0038-1101. DOI: https://doi.org/10.1016/0038-1101(92)90161-5.
- [105] A. N. Tikhonov and V. Y. Arsenin. "Solutions of Ill-Posed Problems". In: (1977), pp. xiii+258. DOI: https://doi.org/10.1007/978-94-015-8480-7.
- [106] A. Asenov, R. Balasubramaniam, A. Brown, and J. Davies. "RTS amplitudes in decananometer MOSFETs: 3-D simulation study". In: *IEEE Transactions on Electron Devices* 50.3 (2003), pp. 839–845. DOI: 10.1109/TED.2003.811418.
- [107] B. Kaczer, P. Roussel, T. Grasser, and G. Groeseneken. "Statistics of Multiple Trapped Charges in the Gate Oxide of Deeply Scaled MOSFET Devices-Application to NBTI". In: *IEEE Electron Device Letters* 31.5 (May 2010), pp. 411– 413. DOI: 10.1109/led.2010.2044014.

- [108] S. M. Amoroso, C. M. Compagnoni, A. Ghetti, L. Gerrer, A. S. Spinelli, A. L. Lacaita, and A. Asenov. "Investigation of the RTN Distribution of Nanoscale MOS Devices From Subthreshold to On-State". In: *IEEE Electron Device Letters* 34.5 (2013), pp. 683–685. DOI: 10.1109/LED.2013.2250477.
- [109] A. Ghetti, C. Monzio Compagnoni, F. Biancardi, A. L. Lacaita, S. Beltrami, L. Chiavarone, A. Spinelli, and A. Visconti. "Scaling trends for random telegraph noise in deca-nanometer Flash memories". In: 2008 IEEE International Electron Devices Meeting. 2008, pp. 1–4. DOI: 10.1109/IEDM.2008.4796827.
- [110] A. Ghetti, M. Bonanomi, C. Monzio Compagnoni, A. Spinelli, A. Lacaita, and A. Visconti. "Physical modeling of single-trap RTS statistical distribution in flash memories". In: 2008 IEEE International Reliability Physics Symposium. 2008, pp. 610–615. DOI: 10.1109/RELPHY.2008.4558954.
- [111] A. Calderoni, P. Fantini, A. Ghetti, and A. Marmiroli. "Modeling the V_{th} fluctuations in nanoscale Floating Gate memories". In: *International Conference on Simulation of Semiconductor Processes and Devices*. 2008, pp. 49–52. DOI: 10.1109/SISPAD.2008.4648234.
- [112] B. Kaczer, T. Grasser, J. Martin-Martinez, E. Simoen, M. Aoulaiche, P. Roussel, and G. Groeseneken. "NBTI from the Perspective of Defect States with Widely Distributed Time Scales". In: 2009 IEEE International Reliability Physics Symposium. 2009, pp. 55–60. DOI: 10.1109/IRPS.2009.5173224.
- [113] J. Canny. "A Computational Approach to Edge Detection". In: *IEEE Transactions on Pattern Analysis and Machine Intelligence* PAMI-8.6 (1986), pp. 679–698.
 DOI: 10.1109/TPAMI.1986.4767851.
- [114] K. Takeuchi, T. Nagumo, S. Yokogawa, K. Imai, and Y. Hayashi. "Single-chargebased modeling of transistor characteristics fluctuations based on statistical measurement of RTN amplitude". In: *Symposium on VLSI Technology*. 2009, pp. 54–55.
- [115] T. Grasser, K. Rott, H. Reisinger, M. Waltl, J. Franco, and B. Kaczer. "A unified perspective of RTN and BTI". In: 2014 IEEE International Reliability Physics Symposium. 2014, pp. 4A.5.1–4A.5.7. DOI: 10.1109/IRPS.2014.6860643.
- [116] B. Stampfer, F. Schanovsky, T. Grasser, and M. Waltl. "Semi-Automated Extraction of the Distribution of Single Defects for nMOS Transistors". In: *MDPI Micromachines* 11(4) (2020), pp. 446. DOI: 10.3390/mi11040446.
- [117] S. E. Rauch. "Review and Reexamination of Reliability Effects Related to NBTI-Induced Statistical Variations". In: *IEEE Transactions on Device and Materials Reliability* 7.4 (2007), pp. 524–530. DOI: 10.1109/TDMR.2007.910437.
- [118] V. Huard, C. Parthasarathy, C. Guerin, T. Valentin, E. Pion, M. Mammasse, N. Planes, and L. Camus. "NBTI degradation: From transistor to SRAM arrays". In: 2008 IEEE International Reliability Physics Symposium. 2008, pp. 289–300. DOI: 10.1109/RELPHY.2008.4558900.

- [119] A. Ortiz-Conde, F. García Sánchez, J. Liou, A. Cerdeira, M. Estrada, and Y. Yue. "A review of recent MOSFET threshold voltage extraction methods". In: *Microelectronics Reliability* 42.4 (2002), pp. 583–596. ISSN: 0026-2714. DOI: https://doi.org/10.1016/S0026-2714(02)00027-6.
- [120] Y.-H. Cheng. "Comparison of MOSFET Threshold Voltage Extraction Methods with Temperature Variation". In: 2019 IEEE 32nd International Conference on Microelectronic Test Structures (ICMTS). 2019, pp. 126–131. DOI: 10.1109/ICMTS. 2019.8730978.
- [121] H. Reisinger, O. Blank, W. Heinrigs, A. Muhlhoff, W. Gustin, and C. Schlunder.
 "Analysis of NBTI Degradation- and Recovery-Behavior Based on Ultra Fast VT-Measurements". In: 2006 IEEE International Reliability Physics Symposium Proceedings. 2006, pp. 448–453. DOI: 10.1109/RELPHY.2006.251260.
- [122] S. Sun and J. Plummer. "Electron mobility in inversion and accumulation layers on thermally oxidized silicon surfaces". In: *IEEE Transactions on Electron Devices* 27.8 (1980), pp. 1497–1508. DOI: 10.1109/T-ED.1980.20063.
- K. Aoyama. "A Method for Extracting the Threshold Voltage of MOSFETs Based on Current Components". In: *Simulation of Semiconductor Devices and Processes*. Ed. by H. Ryssel and P. Pichler. Vienna: Springer Vienna, 1995, pp. 118–121. ISBN: 978-3-7091-6619-2. DOI: 10.1007/978-3-7091-6619-2_28.
- B. Ullmann, K. Puschkarsky, M. Waltl, H. Reisinger, and T. Grasser. "Evaluation of Advanced MOSFET Threshold Voltage Drift Measurement Techniques". In: *IEEE Transactions on Device and Materials Reliability* 19.2 (2019), pp. 358–362. DOI: 10.1109/TDMR.2019.2909993.
- [125] J. Canny. "A Computational Approach to Edge Detection". In: *Readings in Computer Vision*. Elsevier, 1987, pp. 184–203. DOI: 10.1016/b978-0-08-051581-6.50024-6.
- [126] J. Michl, A. Grill, B. Stampfer, D. Waldhoer, C. Schleich, T. Knobloch, E. Ioannidis, H. Enichlmair, R. Minixhofer, B. Kaczer, B. Parvais, B. Govoreanu, I. Radu, T. Grasser, and M. Waltl. "Evidence of Tunneling Driven Random Telegraph Noise in Cryo-CMOS". In: 2021 IEEE International Electron Devices Meeting (IEDM). 2021, pp. 31.3.1–31.3.4. DOI: 10.1109/IEDM19574.2021.9720501.
- [127] N. Otsu. "A Threshold Selection Method from Gray-Level Histograms". In: *IEEE Transactions on Systems, Man, and Cybernetics* 9.1 (1979), pp. 62–66. DOI: 10.1109/TSMC.1979.4310076.
- [128] A. Chambolle. "An Algorithm for Total Variation Minimization and Applications". In: *Journal of Mathematical Imaging and Vision* 20 (2004), pp. 89–97. DOI: https://doi.org/10.1023/B:JMIV.0000011325.36760.1e.

- [129] F. M. Puglisi and P. Pavan. "Factorial Hidden Markov Model analysis of Random Telegraph Noise in Resistive Random Access Memories". In: *ECTI Transactions on Electrical Engineering, Electronics, and Communications* 12 (2014), pp. 24–29. DOI: 10.37936/ecti-eec.2014121.170814.
- [130] T. Grasser, K. Rott, H. Reisinger, M. Waltl, F. Schanovsky, and B. Kaczer. "NBTI in Nanoscale MOSFETs—The Ultimate Modeling Benchmark". In: *IEEE Transactions on Electron Devices* 61.11 (2014), pp. 3586–3593. DOI: 10.1109/TED. 2014.2353578.
- K. Hung, P. Ko, C. Hu, and Y. Cheng. "Random telegraph noise of deepsubmicrometer MOSFETs". In: *IEEE Electron Device Letters* 11.2 (1990), pp. 90– 92. DOI: 10.1109/55.46938.
- [132] M. Toledano-Luque, B. Kaczer, J. Franco, P. Roussel, T. Grasser, and G. Groeseneken. "Defect-centric perspective of time-dependent BTI variability". In: *Microelectronics Reliability* 52.9 (2012), pp. 1883–1890. ISSN: 0026-2714. DOI: https://doi.org/10.1016/j.microrel.2012.06.120.
- [133] J. Franco, B. Kaczer, P. J. Roussel, J. Mitard, S. Sioncke, L. Witters, H. Mertens, T. Grasser, and G. Groeseneken. "Understanding the suppressed charge trapping in relaxed- and strained-Ge/SiO2/HfO2 pMOSFETs and implications for the screening of alternative high-mobility substrate/dielectric CMOS gate stacks". In: *IEEE International Electron Devices Meeting*. 2013, pp. 15.2.1–15.2.4. DOI: 10.1109/IEDM.2013.6724634.
- K. Sonoda, K. Ishikawa, T. Eimori, and O. Tsuchiya. "Discrete Dopant Effects on Statistical Variation of Random Telegraph Signal Magnitude". In: *IEEE Transactions on Electron Devices* 54.8 (2007), pp. 1918–1925. DOI: 10.1109/TED. 2007.900684.
- [135] Z. Zhang, S. Guo, X. Jiang, R. Wang, R. Huang, and J. Zou. "Investigation on the amplitude distribution of random telegraph noise (RTN) in nanoscale MOS devices". In: 2016 IEEE International Nanoelectronics Conference (INEC). 2016, pp. 1–2. DOI: 10.1109/INEC.2016.7589332.
- Z. Zhang, S. Guo, X. Jiang, R. Wang, Z. Zhang, P. Hao, Y. Wang, and R. Huang.
 "New Insights into the Amplitude of Random Telegraph Noise in Nanoscale MOS Devices". In: 2017 IEEE International Reliability Physics Symposium (IRPS). 2017, pp. 3C-3.1-3C–3.5. DOI: 10.1109/IRPS.2017.7936288.
- [137] K. H. Tok, M. Mehedi, J. F. Zhang, Z. Ye, Z. Ji, W. Zhang, and J. Marsland. "Criteria for selecting statistical distribution for the Amplitude of Random Telegraph Noise". In: *IEEE International Conference on Solid-State & Integrated Circuit Technology (ICSICT)*. 2022, pp. 1–4. DOI: 10.1109/ICSICT55466.2022. 9963221.

- [138] Z. Zhang, Z. Zhang, R. Wang, X. Jiang, S. Guo, Y. Wang, X. Wang, B. Cheng, A. Asenov, and R. Huang. "New approach for understanding "random device physics" from channel percolation perspectives: Statistical simulations, key factors and experimental results". In: 2016 IEEE International Electron Devices Meeting (IEDM). 2016, pp. 7.2.1–7.2.4. DOI: 10.1109/IEDM.2016.7838366.
- [139] J. Franco, B. Kaczer, M. Toledano-Luque, P. J. Roussel, G. Groeseneken, B. Schwarz, M. Bina, M. Waltl, P.-J. Wagner, and T. Grasser. "Reduction of the BTI time-dependent variability in nanoscaled MOSFETs by body bias". In: 2013 IEEE International Reliability Physics Symposium (IRPS). 2013, pp. 2D.3.1–2D.3.6. DOI: 10.1109/IRPS.2013.6531958.
- [140] M. Waltl, W. Goes, K. Rott, H. Reisinger, and T. Grasser. "A Single-Trap Study of PBTI in SiON nMOS Transistors: Similarities and Differences to the NBTI/pMOS Case". In: 2014 IEEE International Reliability Physics Symposium. 2014, pp. XT.18.1–XT.18.5. DOI: 10.1109/IRPS.2014.6861195.
- [141] E. Bury, B. Kaczer, K. Chuang, J. Franco, P. Weckx, A. Chasin, M. Simicic, D. Linten, and G. Groeseneken. "Statistical assessment of the full VG/VD degradation space using dedicated device arrays". In: 2017 IEEE International Reliability Physics Symposium (IRPS). 2017, pp. 2D-5.1-2D–5.6. DOI: 10.1109/ IRPS.2017.7936265.
- [142] A. Grill, J. Michl, J. Diaz-Fortuny, A. Beckers, E. Bury, A. Chasin, T. Grasser, M. Waltl, B. Kaczer, and K. De Greve. "A Comprehensive Cryogenic CMOS Variability and Reliability Assessment using Transistor Arrays". In: 2023 7th IEEE Electron Devices Technology Manufacturing Conference (EDTM). 2023, pp. 1– 3. DOI: 10.1109/EDTM55494.2023.10102937.
- [143] A. Grill, J. Michl, J. Diaz-Fortuny, A. Beckers, E. Bury, A. Chasin, T. Grasser, M. Waltl, B. Kaczer, and K. De Greve. "A Comprehensive Cryogenic CMOS Variability and Reliability Assessment using Transistor Arrays". In: 2023 7th IEEE Electron Devices Technology Manufacturing Conference (EDTM). 2023, pp. 1– 3. DOI: 10.1109/EDTM55494.2023.10102937.
- [144] M. Mehedi, K. H. Tok, Z. Ye, J. F. Zhang, Z. Ji, W. Zhang, and J. S. Marsland. "On the Accuracy in Modeling the Statistical Distribution of Random Telegraph Noise Amplitude". In: *IEEE Access* 9 (2021), pp. 43551–43561. DOI: 10.1109/ ACCESS.2021.3065869.
- [145] O. Baumgartner, M. Bina, W. Goes, F. Schanovsky, M. Toledano-Luque, B. Kaczer, H. Kosina, and T. Grasser. "Direct tunneling and gate current fluctuations". In: 2013 International Conference on Simulation of Semiconductor Processes and Devices (SISPAD). 2013, pp. 17–20. DOI: 10.1109/SISPAD.2013.6650563.

- [146] W. Goes, M. Toledano-Luque, O. Baumgartner, M. Bina, F. Schanovsky, B. Kaczer, and T. Grasser. "Understanding correlated drain and gate current fluctuations". In: *Proceedings of the 20th IEEE International Symposium on the Physical and Failure Analysis of Integrated Circuits (IPFA)*. 2013, pp. 51–56. DOI: 10.1109/IPFA.2013.6599125.
- [147] C. Schleich, D. Waldhör, T. Knobloch, W. Zhou, B. Stampfer, J. Michl, M. Waltl, and T. Grasser. "Single- Versus Multi-Step Trap Assisted Tunneling Currents—Part I: Theory". In: *IEEE Transactions on Electron Devices* 69.8 (2022), pp. 4479–4485. DOI: 10.1109/TED.2022.3185966.

List of Publications

Scientific Journals

- [KTJ1] M. Waltl, T. Knobloch, K. Tselios, L. Filipovic, B. Stampfer, Y. Hernandez, D. Waldhör, Y. Illarionov, B. Kaczer, and T. Grasser. "Perspective of 2D integrated electronic circuits: Scientific pipe dream or disruptive technology?" In: *Advanced Materials* (2022), p. 2201082. DOI: 10.1109/TED.2022.3185965.
- [KTJ2] M. Waltl, D. Waldhoer, K. Tselios, B. Stampfer, C. Schleich, G. Rzepa, H. Enichlmair, E. G. Ioannidis, R. Minixhofer, and T. Grasser. "Impact of single-defects on the variability of CMOS inverter circuits". In: *Microelectronics Reliability* 126 (2021). Proceedings of ESREF 2021, 32nd European Symposium on Reliability of Electron Devices, Failure Physics and Analysis, p. 114275. DOI: https://doi.org/10.1016/j.microrel.2021.114275.
- [KTJ3] D. Waldhoer, C. Schleich, J. Michl, B. Stampfer, K. Tselios, E. G. Ioannidis, H. Enichlmair, M. Waltl, and T. Grasser. "Toward Automated Defect Extraction From Bias Temperature Instability Measurements". In: *IEEE Transactions on Electron Devices* 68.8 (2021), pp. 4057–4063. DOI: 10.1109/TED.2021.3091966.
- [KTJ4] K. Tselios, D. Waldhoer, B. Stampfer, J. Michl, E. G. Ioannidis, H. Enichlmair, T. Grasser, and M. Waltl. "On the Distribution of Single Defect Threshold Voltage Shifts in SiON Transistors". In: *IEEE Transactions on Device and Materials Reliability* 21.2 (2021), pp. 199–206. DOI: 10.1109/TDMR.2021.3080983.
- [KTJ5] K. Tselios, J. Michl, T. Knobloch, H. Enichlmair, E. G. Ioannidis, R. Minixhofer, T. Grasser, and M. Waltl. "Evaluation of the impact of defects on threshold voltage drift employing SiO2 pMOS transistors". In: *Microelectronics Reliability* 138 (2022). 33rd European Symposium on Reliability of Electron Devices, Failure Physics and Analysis, p. 114701. DOI: https://doi.org/10.1016/j. microrel.2022.114701.
- [KTJ6] C. Schleich, D. Waldhör, A.-M. El-Sayed, K. Tselios, B. Kaczer, T. Grasser, and M. Waltl. "Single- Versus Multi-Step Trap Assisted Tunneling Currents—Part II: The Role of Polarons". In: *IEEE Transactions on Electron Devices* 69.8 (2022), pp. 4486–4493. DOI: 10.1109/TED.2022.3185965.

[KTJ7] K. Tselios, T. Knobloch, D. Waldhoer, B. Stampfer, E. Ioannidis, H. Enichlmair, R. Minixhofer, T. Grasser, and M. Waltl. "Revealing the Impact of Gate Area Scaling on Charge Trapping employing SiO2 Transistors". In: *IEEE Transactions* on Device and Materials Reliability (2023), pp. 1–1. DOI: 10.1109/TDMR.2023. 3262141.

Conference Proceedings

- [KTC1] K. Tselios, T. Knobloch, J. Michl, D. Waldhoer, C. Schleich, E. Ioannidis, H. Enichlmair, R. Minixhofer, T. Grasser, and M. Waltl. "Impact of Single Defects on NBTI and PBTI Recovery in SiO2 Transistors". In: 2022 IEEE International Integrated Reliability Workshop (IIRW). 2022, pp. 1–5. DOI: 10.1109/IIRW56459. 2022.10032748.
- [KTC2] K. Tselios, B. Stampfer, J. Michl, E. Ioannidis, H. Enichlmair, and M. Waltl. "Distribution of Step Heights of Electron and Hole Traps in SiON nMOS Transistors". In: 2020 IEEE International Integrated Reliability Workshop (IIRW). 2020, pp. 1–6. DOI: 10.1109/IIRW49815.2020.9312871.